

Sheringham Shoal and Dudgeon Offshore Wind Farm Extension Projects

Book of Reference (Revision H) (Clean)

Revision H Deadline 8 July 2023 Document Reference: 4.1 APFP Regulation: 5(2)(d)









Title: Sheringham Shoal and Dudgeon Offshore Wind Farm Extension Projects DCO Application Book of Reference (Revision H) (Clean)		
PINS document re	eference: 4.1.1	
Revision: H	Revision: H	
Document reference: C282-DM-Z-GA-00003		3
Date:	Classification	
July 2023	Final	
Prepared by:		
Burges Salmon LLP/Dalcour Maclaren		
Approved by:		Date:
Sheery Atkins, E	quinor	July 2023



Doc. No. C282-DM-Z-GA-00003

Rev. H

Table of Contents

1	Introduction	8
2	Purpose of Document	
2.1	Structure of Document	
3	Project Description	
4	Categories of New Rights	
5	Book of Reference	



Glossary of Acronyms

BoR	Book of Reference
DCO	Development Consent Order
DEP	Dudgeon Offshore Wind Farm Extension Project
HDD	Horizontal directional drilling
HVAC	High-Voltage Alternating Current
km	Kilometre
MW	Megawatts
NSIP	Nationally Significant Infrastructure Project
PA 2008	Planning Act 2008
SEP	Sheringham Offshore Wind Farm Extension Project
SoS	Secretary of State



Glossary of Terms

APFP Regulations	The Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009
Dudgeon Offshore Wind Farm Extension site	The Dudgeon Offshore Wind Farm Extension consisting of the DEP wind farm site, interlink cable corridors and offshore export cable corridor (up to mean high water springs).
Dudgeon Offshore Wind Farm Extension Project (DEP)	The Dudgeon Offshore Wind Farm Extension onshore and offshore sites including all onshore and offshore infrastructure.
DEP North array area	The wind farm site area of the DEP offshore site located to the north of the existing Dudgeon Offshore Wind Farm
DEP South array area	The wind farm site area of the DEP offshore site located to the south of the existing Dudgeon Offshore Wind Farm
DEP wind farm site	The offshore area of DEP within which wind turbines, infield cables and offshore substation platform/s will be located and the adjacent Offshore Temporary Works Area. This is also the collective term for the DEP North and South array areas.
Grid option	Mechanism by which SEP and DEP will connect to the existing electricity network. This may either be an integrated grid option providing transmission infrastructure which serves both of the wind farms, or a separated grid option, which allows SEP and DEP to transmit electricity entirely separately.
Infield cables	Cables which link the wind turbine generators to the offshore substation platform(s).
Interlink cables	Cables linking two separate project areas. This can be cables linking:
	 DEP South array area and DEP North array area DEP South array area and SEP
	3) DEP North array area and SEP
	1 is relevant if DEP is constructed in isolation or first in a phased development.
	2 and 3 are relevant where both SEP and DEP are built.



Integrated Grid Option	Transmission infrastructure which serves both extension projects.
Jointing bays	Underground structures constructed at regular intervals along the onshore cable route to join sections of cable and facilitate installation of the cables into the buried ducts.
Landfall	The point at the coastline at which the offshore export cables are brought onshore, connecting to the onshore cables at the transition joint bay above mean high water.
Offshore cable corridors	This is the area which will contain the offshore export cables or interlink cables, including the adjacent Offshore Temporary Works Area.
Offshore export cable corridor	This is the area which will contain the offshore export cables between offshore substation platform/s and landfall, including the adjacent Offshore Temporary Works Area.
Offshore export cables	The cables which would bring electricity from the offshore substation platform(s) to the landfall. 220 – 230kV.
Offshore substation platform (OSP)	A fixed structure located within the wind farm site/s, containing electrical equipment to aggregate the power from the wind turbine generators and convert it into a more suitable form for export to shore.
Offshore Temporary Works Area	An Offshore Temporary Works Area within the offshore Order Limits in which vessels are permitted to carry out activities during construction, operation and decommissioning encompassing a 200m buffer around the wind farm sites and a 750m buffer around the offshore cable corridors. No permanent infrastructure would be installed within the Offshore Temporary Works Area.
Onshore cable corridor	The area between the landfall and the onshore substation sites, within which the onshore cable circuits will be installed along with other temporary works for construction.
Onshore export cables	The cables which would bring electricity from the landfall to the onshore substation. 220 – 230kV.
Onshore Substation	Compound containing electrical equipment to enable connection to the National Grid.
Order Land	The land and rights over land for which compulsory powers of acquisition are sought in the DCO as set out in the Book of Reference.



Order Limits	The area subject to the application for development consent, including all permanent and temporary works for SEP and DEP.
Separated Grid Option	Transmission infrastructure which allows each project to transmit electricity entirely separately.
Sheringham Shoal Offshore Wind Farm Extension Project (SEP)	The Sheringham Shoal Offshore Wind Farm Extension onshore and offshore sites including all onshore and offshore infrastructure.
SEP offshore site	Sheringham Shoal Offshore Wind Farm Extension consisting of the SEP wind farm site and offshore export cable corridor (up to mean high water springs).
SEP onshore site	The Sheringham Shoal Wind Farm Extension onshore area consisting of the SEP onshore substation site, onshore cable corridor, construction compounds, temporary working areas and onshore landfall area.
SEP wind farm site	The offshore area of SEP within which wind turbines, infield cables and offshore substation platform/s will be located and the adjacent Offshore Temporary Works Area.
Study area	Area where potential impacts from the project could occur, as defined for each individual Environmental Impact Assessment (EIA) topic.
The Applicant	Equinor New Energy Limited. As the owners of SEP and DEP, Scira Extension Limited (SEL) and Dudgeon Extension Limited (DEL) are the named undertakers that have the benefit of the Development Consent Order. References in this document to obligations on, or commitments by, 'the Applicant' are given on behalf of SEL and DEL as the undertakers of SEP and DEP.
Transition joint bay	Connects offshore and onshore export cables at the landfall. The transition joint bay will be located above mean high water.



1 Introduction

- This Book of Reference (BoR) relates to the powers of compulsory acquisition sought in the draft development consent order (DCO) application by Equinor New Energy Limited (the Applicant), on behalf of Scira Extension Limited and Dudgeon Extension Limited, to the Secretary of State (SoS) under the Planning Act 2008 (the PA 2008) for powers to construct and operate two offshore wind farms, known as Sheringham Shoal Offshore Wind Farm Extension Project (SEP) and Dudgeon Offshore Wind Farm Extension Project (DEP), both located off the coast of Norfolk.
- 2. SEP is the proposed extension to the operational Sheringham Shoal Offshore Wind Farm and will comprise up to 23 wind turbine generators, together with the associated onshore and offshore infrastructure. The offshore cable corridor from SEP to landfall will be approximately 40km in length and the onshore cable corridor will be approximately 60km in length.
- 3. DEP is the proposed extension to the operational Dudgeon Wind Farm and will comprise up to 30 wind turbine generators, together with the associated onshore and offshore infrastructure. The offshore export cable corridor for the project will be approximately 62km in length and the onshore cable corridor will be approximately 60km in length.
- 4. Development consent is required to the extent that development is or forms part of a Nationally Significant Infrastructure Project (NSIP) pursuant to sections 14(1)(a) and 15(3) of the PA 2008. As SEP and DEP will each have an overall capacity greater than 100 megawatts (MW), they are NSIPs for the purposes of the PA 2008. It is for this reason that SEP and DEP fall within the remit of the Secretary of State (SoS).
- 5. In order to develop SEP and DEP a series of land rights and interests in land will be required, on a permanent and temporary basis. In the event it has not been possible to acquire the land rights and interests by agreement, it will be necessary to compulsorily acquire land these for the purposes of developing SEP and DEP and as such, the **draft DCO** (document reference 3.1) includes powers to compulsorily acquire land.
- 6. This BoR should be read alongside the following Application documents:
 - **draft DCO** (document reference 3.1)
 - **Explanatory Memorandum** (document reference 3.2)
 - Land Plans (document reference 2.3, 2.4 and 2.5)
 - **Special Category Land Plan** (document reference 2.5)
 - **Crown Land Plan** (document reference 2.4)
 - Works Plans (Onshore) (document reference 2.6)
 - **Statement of Reasons** (document reference 4.3)



2 **Purpose of Document**

- 7. Section 122 of the PA 2008 allows development consent orders to be granted with rights to compulsorily acquire land included within them. The **draft DCO** (document reference 3.1) seeks powers to acquire land (on a permanent or temporary basis) and to acquire new and existing rights over land, as well as powers to override existing easements and other rights and for the extinguishment of private rights in land which is compulsorily acquired. The purpose of these powers is to enable the construction, operation, maintenance and decommissioning of SEP and DEP.
- 8. The purpose of this BoR is to describe all of the land, and identify all of the interests, subject to compulsory acquisition pursuant to the draft DCO (document reference 3.1). This BoR has been prepared in accordance with the requirements of Regulation 5(2)(d) and Regulation 7 of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 (APFP Regulations).
- 9. The land described in this BoR, which is to be subject to powers of compulsory acquisition and/or temporary possession, is required to deliver the authorised development, as described in detail in Part 1 of Schedule 1 of the **draft DCO** (document reference 3.1) and shown on the **Works Plans** (document references 2.6 and 2.7).
- Every parcel of land that is affected is identified on a plot by plot basis and a unique number has been ascribed to each plot. The plots are shown on the Land Plans (document reference 2.3), the Crown Land Plan (document reference 2.4) and the Special Category Land Plan (document reference 2.5), which are submitted with the Application.
- 11. A number of the plots identified in Part 1 of the BoR will be subject to freehold acquisition pursuant to Article 18 (Compulsory acquisition of land) of the draft DCO (document reference 3.1). These plots are shown coloured pink on the Land Plans (document reference 2.3).
- 12. A number of plots identified in Part 1 of the BoR will be subject to the acquisition of permanent rights (including restrictive covenants) pursuant to Article 20 (Compulsory acquisition of rights) and Schedule 7 (Land in which only new rights etc. may be acquired) of the draft DCO (document reference 3.1). These plots are shown coloured blue on the Land Plans (document reference 2.3).
- 13. The land shown coloured blue and pink on the Land Plans (document reference 2.3) will also be subject to powers of temporary possession for the purpose of carrying out the authorised development (by virtue of Article 26 (Temporary use of land for carrying out the authorised project)).
- 14. Plots that are subject to powers of temporary possession only, such as for the purpose of access to and/or use as a temporary construction compound, are also listed in Schedule 9 (Land of which temporary possession may be taken) of the **draft DCO** (document reference 3.1) and are shown coloured yellow on the Land Plans (document reference 2.3).



15. By virtue of Article 27 (Temporary use of land for maintaining the authorised project) of the **draft DCO** (document reference 3.1), any land within the Order limits which is reasonably required for the purpose of maintaining the authorised development may be entered and/or temporarily possessed.

2.1 Structure of Document

- 16. This BoR is comprised of five Parts, in accordance with Regulation 7(1) of the APFP Regulations, as follows:
 - Part 1 contains the names and addresses of those who own, lease, occupy or have another interest in the land that will be affected by the compulsory acquisition powers contained in the **draft DCO** (document reference 3.1);
 - Part 2 contains the names and addresses of those whose land, while not directly affected by the authorised development, may be entitled to claim compensation for loss resulting from the implementation of the **draft DCO** (document reference 3.1) and use of the authorised development;
 - Part 3 contains the names and addresses of those entitled to enjoy easements or other private rights which it is proposed may be extinguished, suspended or interfered with in connection with the authorised development, pursuant to the **draft DCO** (document reference 3.1);
 - Part 4 identifies plots in which there is a Crown interest that will be affected by the authorised development and the powers contained in the **draft DCO** (document reference 3.1) (please note that these plots will not be subject to powers of compulsory acquisition); and
 - Part 5 identifies plots which constitute "special category land" for the purposes of Section 132 of the PA 2008 that will be affected by the authorised development and the powers contained in the **draft DCO** (document reference 3.1).
- 17. The BoR provides the area in square metres of all land subject to compulsory acquisition powers pursuant to the **draft DCO** (document reference 3.1). Each measurement is rounded up to the nearest whole square metre.

3 Project Description

- 18. A description of the key components of the proposed SEP and DEP, as well as details of how the wind farms will be constructed, operated, maintained and decommissioned is provided in the **Project Description** (document reference 6.1.4).
- SEP and DEP will each have a maximum export capacity greater than 100 megawatts (MW). The SEP and DEP wind farm sites are 15.8 kilometres (km) and 26.5km from the coast for SEP and DEP respectively at their closest point.



- 20. SEP and DEP will be connected to shore by offshore export cables installed to the landfall at Weybourne, on the north Norfolk coast. From there, the onshore export cables travel approximately 60km inland to a new high voltage alternating current (HVAC) onshore substation near to the existing Norwich Main substation. The onshore substation will be constructed to accommodate the connection of both SEP and DEP to the transmission grid.
- 21. The key offshore components comprise:
 - Wind turbines;
 - Offshore substation platform/s (OSP);
 - Foundation structures for wind turbines and OSP/s;
 - Infield cables;
 - Interlink cables; and
 - Export cables from the wind farm sites to the landfall.
- 22. The key onshore components comprise:
 - Landfall and associated transition joint bay/s;
 - Onshore export cables installed underground from the landfall to the onshore substation and associated joint bays and link boxes;
 - Onshore substation and onward 400 kilovolt (kV) connection to the existing Norwich Main substation;
 - Trenchless crossing zones (e.g. Horizontal Directional Drilling (HDD));
 - Construction and operational accesses; and
 - Temporary construction compounds.
- 23. Further details of the key components of offshore and onshore infrastructure can be found in **Chapter 4 Project Description** of the ES (document reference 6.1.4).
- 24. The Applicant is seeking to coordinate the development of SEP and DEP as far as possible. The preferred option is a development scenario with an integrated transmission system, providing transmission infrastructure which serves both of the wind farms, where both projects are built concurrently. However, given the different commercial ownerships of each Project, alternative development scenarios such as a separated grid option (i.e. transmission infrastructure which allows each Project to transmit electricity entirely separately) will allow SEP and DEP to be constructed in a phased approach, if necessary. Therefore, the DCO application seeks to consent a range of development scenarios in the same overall corridors to allow for separate development if required, and to accommodate either sequential or concurrent build of the two projects.
- 25. Reasons for the requirement to retain separate and phased (sequential) development scenarios alongside more coordinated approaches are further described in the **Scenarios Statement** (document reference 9.28).



4 Categories of New Rights

- 26. Article 20 (Compulsory acquisition of rights) and Schedule 7 (Land in which only new rights etc. may be acquired) of the **draft DCO** (**Revision F**) (document reference 3.1) authorise the compulsory acquisition of new rights over land.
- 27. The descriptions of the new rights sought by the Applicant set out in Table 1-1 below shall apply to the column headed "extent, description and situation of the land" in Part 1 of this BoR. For this purpose the letter in column 1 of Table 1-1 is used to indicate the new rights sought over each relevant plot of land. The details are described in column 2 of Table 1-1. The new rights are sought for the purposes of construction, operation, maintenance and decommissioning of SEP and DEP.
- 28. In **Table 1-1** the terms used shall have the meaning given to those terms within the **draft DCO (Revision F)** (document 3.1) unless stated below:
 - "adjoining land" means such other parts of the land within the Order limits as may be required for the authorised development;
 - "cables" means collectively underground electricity cables together with telemetry and fibre optic lines, structures, ducting, earthing measures, thermal protection, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such electricity cables;
 - "Land" means the relevant plot shown on the Land Plans (Revision D) (document reference 2.3) and described in this BoR;

F	Description of rights and restrictive covenants Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development
n	maintenance and decommissioning of the authorised development
A – Landfall Connection Works	 a) adjust, alter, remove, replace, and create tunnels under sea defences including walls and groynes; b) effect access to offshore apparatus and carry out works for the purposes of construction, installation, operation, maintenance and decommissioning of the parts of the authorised development that communicate between the onshore and offshore elements of the authorised development; and c) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove



	ducts and jointing bays, and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences and existing infrastructure.
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	 d) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto); e) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or
	surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised development nor make it materially more difficult or expensive to maintain the authorised development);
	f) prevent anything to be done by way of mole draining or excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.7m for the purposes of arable farming) or are required to be carried out by National Grid in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the Land;
	 g) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development; and
	 h) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development.
B – Landfall and Onshore Connection Works	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to:



 a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables and ducts (including the removal of materials including spoil) in, under, over and/or on the Land, including by way of pull-through within the cable ducts and jointing bays, and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, highways (including public rights of way) and railways, woodlands, solar energy generation facilities, existing infrastructure; b) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, vessels, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables, or use of the cables, cable ducts and jointing works; c) benefit from continuous vertical and lateral support for the authorised development; d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or h		
 the cables, cable ducts and jointing works; c) benefit from continuous vertical and lateral support for the authorised development; d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 		use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables and ducts (including the removal of materials including spoil) in, under, over and/or on the Land, including by way of pull-through within the cable ducts and jointing bays, and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, highways (including public rights of way) and railways, woodlands, solar energy generation facilities, existing infrastructure; enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, vessels, machinery, apparatus and equipment which is ancillary to the
 c) benefit from continuous vertical and lateral support for the authorised development; d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 		purposes of transmitting electricity along the cables, or use of
 authorised development; d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 		
 d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 	c)	
 apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 		
 e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 	d)	apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised
 including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 		
 structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; g) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 	e)	including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development
 hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; h) erect and remove temporary fencing, gates, walls, barriers or 	f)	structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel
		hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land;
	h)	



i) take access to the Land and to use it to effect access to and egress from adjacent land, without or without vehicles and equipment, for the purposes of exercising any right vested in the undertaker, maintaining, replacing or altering for environmental or ecological mitigation or enhancement works, or for the purposes of undertaking surveying or monitoring activities required for the purposes of the authorised development; i) construct and install the cables in, on or under the Land and thereafter use the Land for all necessary purposes for the commissionina. construction. repair, testina and maintenance of the cables; k) place and use plant, machinery, structures and temporary structures within the Land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the cables, cable ducts and jointing works and to erect temporary signage and provide measures for the benefit of public and personnel safety; I) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas; m)effect access to and from the highway; n) make such investigations in or on the Land as required; o) alter, fell, lop or cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the operation of the cables; p) remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the Land or reinstate the any existing drainage scheme on the Land; q) install, alter, re-lay, maintain, protect, adjust, use or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and



any ancillary equipment and apparatus, public and private drains, watercourses, sewers, ponds or culverts and service media (including the pipes, cables or conduits or apparatus of statutory undertakers and providing connections to the authorised development);

- r) remove fences hedges or other barriers within the Land during any period during which construction, maintenance, upgrading, improvement, repair, removal or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);
- s) enter upon the Land to create temporary secure areas and to remove, store and stockpile materials (including excavated material) within the Land;
- t) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair, replacement or decommissioning and to reinstate the Land;
- u) excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;
- v) carry out environmental mitigation, remediation and enhancement works;
- w)install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seedlings and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs, landscaping for the period of aftercare and maintenance of such planting (as set out in the landscape and ecological management plan approved under this Order);
- x) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including the creation of hibernacula, introduction of artificial badger setts and installation of bat boxes, temporary works and the installation of temporary barriers for the protection of fauna;



 y) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the Land; z) install, construct, use and remove temporary welfare facilities during any periods of construction, maintenance, repair, replacement, renewal, upgrade and removal of the cables, cable ducts and jointing works; aa) place temporarily and use plant, machinery and structures on the land in connection with the lighting of the land and the
authorised development;
bb) place equipment on the land, including portakabins and welfare equipment;
cc) create car parking sites, site offices, site areas for temporary security and welfare facilities;
 dd) create fuel storage and bunded facilities for the storage of materials ancillary to the implementation of the authorised development;
ee) lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace transition joint bays;
 ff) to adjust, alter, remove, replace, and create tunnels under sea defences including walls and groynes; gg) effect access to offshore apparatus and carry out works for the purposes of construction, installation, operation, maintenance and decommissioning of the parts of the authorised development that communicate between the onshore and offshore elements of the authorised development; and
hh) install, retain, and connect apparatus to connect onshore transmission apparatus to offshore transmission apparatus.ii) A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
 jj) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto);
kk) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the



undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised development nor make it materially more difficult or expensive to maintain the authorised development);

- II) prevent anything to be done by way of mole draining or excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.7m for the purposes of arable farming) or are required to be carried out by National Grid in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the Land;
- mm) prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development);
- nn) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development;
- oo) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development;
- pp) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas, areas of habitat creation and areas of reinstatement where planting requires time to re-establish, including any ploughing or grazing without the prior written consent of the undertaker;
- qq) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation or areas of reinstatement where planting requires time to re-establish



	 including any ploughing or grazing without the prior written consent of the undertaker; rr) prevent the uprooting, removal, destruction or damaging of any plant, bush, shrub, sapling or tree planted on the Land by the undertaker during the period within which the undertaker is responsible for the aftercare and maintenance of planting (as set out in the landscape and ecological management plan approved under this Order); and ss) control or restrict access and use where such control or restriction is necessary for ecological or environmental mitigation or enhancement works to establish, including preventing access by the public on foot during any period where planting required under this Order is establishing (set out in the landscape and ecological management plan approved under this Order).
C – Permanent Access - Landfall	 Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to: a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials to access adjoining land for the purposes of the Works, the inspection, use, testing, maintenance, renewal, upgrading, replacement and removal of the cables, cable ducts and jointing works and connection into any adjacent cables and associated works and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rights; b) construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway; c) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; and d) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land.



	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	 remainder of the Order land to: e) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto); f) prevent anything to be done by way of mole draining or excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.7m for the purposes of arable farming) or are required to be carried out by National Grid in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the Land; g) prevent the planting or growing within the Land of any trees, shrubs or underwood without the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development); and h) to prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage
	the authorised development.
	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to:
D - Onshore connection works (Cable Corridor)	 a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables and ducts (including the removal of materials including spoil) in, under, over and/or on the Land, including by way of pull-through within the cable ducts and jointing bays, and in doing so, to use or resort to trenchless installation techniques including (but not limited to) directional drilling beneath sea defences, watercourses, highways (including public rights of way) and railways, woodlands, solar energy generation facilities, existing infrastructure;



 b) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, vessels, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables, or use of the cables, cable ducts and jointing works; c) benefit from continuous vertical and lateral support for the authorised development; d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) retain, maintain and use temporary supporting or protective structures and erect temporary supporting or protective structures (including the bridging over or protection of the apparatus of the statutory undertakers) for the purposes of access to adjoining land; g) place and use plant, machinery, structures and temporary signage and provide maceurae for the barefit of public and parsonnal.
g) place and use plant, machinery, structures and temporary
 h) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; i) erect and remove temporary fencing, gates, walls, barriers or other means of enclosure;
 j) lay out and maintain temporary paths and bridleways for public use as temporary diversions for public rights of way which are interfered with during any period in which construction, maintenance, repair or renewal decommissioning is being carried out;



k) take access to the Land and to use it to effect access to and egress from adjacent land, without or without vehicles and equipment, for the purposes of exercising any right vested in the undertaker, maintaining, replacing or altering for environmental or ecological mitigation or enhancement works, or for the purposes of undertaking surveying or monitoring activities required for the purposes of the authorised development; I) construct and install the cables in, on or under the Land and thereafter use the Land for all necessary purposes for the commissionina. construction. repair, testina and maintenance of the cables; m)place and use plant, machinery, structures and temporary structures within the Land for the purposes of the installation, construction, maintenance, repairing, renewing, upgrading, inspecting, removal and replacing of the cables, cable ducts and jointing works and to erect temporary signage and provide measures for the benefit of public and personnel safety; n) erect fencing, gates, walls, barriers or other means of enclosure, and create secure working areas and compounds including trenchless installation technique compounds and working areas; o) effect access to and from the highway; p) make such investigations in or on the Land as required; q) alter, fell, lop or cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be standing on the Land or other land which would if not felled, lopped, cut or removed would obstruct or interfere with the operation of the cables; r) remove and discharge water from the Land, and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, install, use, inspect, maintain, adjust, alter, renew, repair, test or cleanse drainage schemes on the Land or reinstate the any existing drainage scheme on the Land; s) install, alter, re-lay, maintain, protect, adjust, use or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and



any ancillary equipment and apparatus, public and private drains, watercourses, sewers, ponds or culverts and service media (including the pipes, cables or conduits or apparatus of statutory undertakers and providing connections to the authorised development);

- t) remove fences hedges or other barriers within the Land during any period during which construction, maintenance, upgrading, improvement, repair, removal or renewal is being carried out (subject to erection of any temporary stock-proof fencing as is reasonably required and the re-instatement or suitable replacement of the fences or structures following the exercise of the rights);
- u) enter upon the Land to create temporary secure areas and to remove, store and stockpile materials (including excavated material) within the Land;
- v) create boreholes and trial excavation pits for the purposes of intrusively surveying the land and monitoring the use of any trenchless installation technique, to keep in place and monitor the same through construction, maintenance repair, replacement or decommissioning and to reinstate the Land;
- w) excavate materials below ground level, including soils, and to store and re-use or dispose of the same, and in so excavating to undertake any works, including works of protection or removal of archaeological remains as may be required by any written scheme of investigation approved under this Order;
- x) carry out environmental mitigation, remediation and enhancement works;
- y) install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seedlings and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs, landscaping for the period of aftercare and maintenance of such planting (as set out in the landscape and ecological management plan approved under this Order);
- z) install, execute, implement, retain, repair, improve, renew, relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including the creation of hibernacula, introduction of artificial badger setts and installation of bat boxes, temporary works and the installation of temporary barriers for the protection of fauna;



aa) (in an emergency only when the cables are temporarily unusable) to lay down, install, use, maintain and inspect	
 underground cables, telephone signalling and fibre-optic cables and ancillary equipment, associated works and other conducting media together with conduits or pipes for containing the same in and under the Land; bb) install, construct, use and remove temporary welfare facilities during any periods of construction, maintenance, repair, replacement, renewal, upgrade and removal of the cables, cable ducts and jointing works; cc) place temporarily and use plant, machinery and structures on the land in connection with the lighting of the land and the authorised development; dd) place equipment on the land, including portakabins and welfare equipment; ee) create car parking sites, site offices, site areas for temporary security and welfare facilities; and ff) create fuel storage and bunded facilities for the storage of materials ancillary to the implementation of the authorised development. 	r r ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ;
A restrictive covenant over the Land for the benefit of the remainder of the Order land to:	
 gg) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto); hh) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised development nor make it materially more difficult or expensive to maintain the authorised development); ii) prevent anything to be done by way of mole draining or excavation of any kind in the Land nor any activities which 	r e r e t
excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.7m for the purposes of arable farming);	s f



	jj) prevent the planting or growing within the Land of any trees,
	 shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development); kk) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development; ll) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development; mm) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas, areas of habitat creation and areas of reinstatement where planting requires time to reestablish, including any ploughing or grazing without the prior written consent of the undertaker; nn) prevent the uprooting, removal, destruction or damaging of any plant, bush, shrub, sapling or tree planted on the Land by the undertaker during the period within which the undertaker is reasonable for the aftercare and maintenance of such planting (as set out in the landscape and ecological mitigation or enhancement works to establish, including preventing access by the public on foot during any period where planting required under this Order is establish, including preventing management plan approved under this Order is establish, including preventing access by the public on foot during any period where planting required under this Order is establishing (set out in the landscape and ecological management plan
	approved under this Order).
E - Onshore substation connection works	 Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to: a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables and ducts (including the removal)

	of materials including spoil) in, under, over and/or on the Land,
	including by way of pull-through within the cable ducts and
	jointing bays;
	b) enter, be on, and break open and break up the surface of the
	Land and remain with or without plant, vehicles, machinery,
	apparatus and equipment which is ancillary to the purposes
	of transmitting electricity along the cables or use of the
	cables, cable ducts and jointing works;
	c) benefit from continuous vertical and lateral support for the
	authorised development;
	d) pass and re-pass with or without vehicles, plant, machinery,
	apparatus, equipment and materials for the purposes of
	laying down, installing, adjusting, altering, constructing,
	using, maintaining, repairing, renewing, upgrading,
	inspecting, removing and replacing the authorised
	development;
	e) construct, lay down, use and remove temporary access roads
	including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate,
	trackway, stone, tarmacadam, terram, temporary bridging,
	culverting or diversion of water courses and drains during any
	period during which construction, maintenance, repair,
	renewal or decommissioning of the authorised development
	is being carried out;
	f) place and use plant, machinery, structures and temporary
	structures within the Land, and to erect temporary signage
	and provide measures for the benefit of public and personnel
	safety;
	g) alter, fell, lop, cut, coppice , uproot or replant trees or hedges
	or other vegetation which now or hereafter may be present
	on the Land for the purpose of enabling the right to pass and
	re-pass to adjoining land;
	h) remove and discharge water from the Land and to install,
	retain, use, maintain, inspect, alter, remove, refurbish,
	reconstruct, replace, protect and improve sewers, drains,
	pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts,
	to lay down, install, adjust, alter, construct, create, use,
	maintain, repair, renew, upgrade, inspect, remove and
	replace a drainage scheme on the Land;
	i) install, alter, re-lay, maintain, protect, adjust or remove pipes,
	cables or conduits or apparatus including but not limited to
I	-



electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);

- j) National Grid substation connection works to lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables and ducts (including the removal of materials including spoil) in, under, over and/or on the Land, including by way of pull-through within the cable ducts and jointing bays, together with such telemetry and fibre-optic lines, structures, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such cables to the National Grid substation;
- k) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials to access adjoining land for the purposes of the Works, the inspection, use, testing, maintenance, renewal, upgrading, replacement and removal of the cables, cable ducts and jointing works and connection into any adjacent cables and associated works and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rights;
- construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway; and
- m)install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seedlings and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs, landscaping for the period of aftercare and maintenance of such planting (as set out in the landscape and ecological management plan approved under this Order).

A restrictive covenant over the Land for the benefit of the remainder of the Order land to:



- n) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto);
- o) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised development nor make it materially more difficult or expensive to maintain the authorised development);
- p) prevent anything to be done by way of mole draining or excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.7m for the purposes of arable farming) or are required to be carried out by National Grid in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the Land;
- q) prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development);
- r) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development;
- s) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development;
- t) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas, areas of habitat creation and areas of reinstatement where planting requires time to re-establish,



	 including any ploughing or grazing without the prior written consent of the undertaker; and u) prevent the uprooting, removal, destruction or damaging of any plant, bush, shrub, sapling or tree planted on the Land by the undertaker during the period within which the undertaker is reasonable for the aftercare and maintenance of such planting (as set out in the landscape and ecological management plan approved under this Order).
	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to:
F - Connection works to National Grid Substation	 a) lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables and ducts (including the removal of materials including spoil) in, under, over and/or on the Land, including by way of pull-through within the cable ducts and jointing bays; b) enter, be on, and break open and break up the surface of the Land and remain with or without plant, vehicles, machinery, apparatus and equipment which is ancillary to the purposes of transmitting electricity along the cables or use of the cables, cable ducts and jointing works; c) benefit from continuous vertical and lateral support for the authorised development; d) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials for the purposes of laying down, installing, adjusting, altering, constructing, using, maintaining, repairing, renewing, upgrading, inspecting, removing and replacing the authorised development; e) construct, lay down, use and remove temporary access roads including any necessary hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram, temporary bridging, culverting or diversion of water courses and drains during any period during which construction, maintenance, repair, renewal or decommissioning of the authorised development is being carried out; f) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage



and provide measures for the benefit of public and personnel safety;

- g) alter, fell, lop, cut, coppice, uproot or replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land;
- h) remove and discharge water from the Land and to install, retain, use, maintain, inspect, alter, remove, refurbish, reconstruct, replace, protect and improve sewers, drains, pipes, ducts, mains, conduits, flues and to drain into and manage waterflows in any drains, watercourses and culverts, to lay down, install, adjust, alter, construct, create, use, maintain, repair, renew, upgrade, inspect, remove and replace a drainage scheme on the Land;
- install, alter, re-lay, maintain, protect, adjust or remove pipes, cables or conduits or apparatus including but not limited to electricity poles, electricity pylons, electricity masts, overhead electricity lines, telecommunications cables and any ancillary equipment and apparatus public and private drains, watercourses, sewers, ponds or culverts, service media (including the pipes, cables or conduits or apparatus of statutory undertakers);
- j) National Grid substation connection works to lay down, install, retain, adjust, alter, construct, operate, erect, use, maintain, repair, renew, upgrade, inspect, remove and replace the electricity cables and ducts (including the removal of materials including spoil) in, under, over and/or on the Land, including by way of pull-through within the cable ducts and jointing bays, together with such telemetry and fibre-optic lines, structures, ducting, jointing bays and other apparatus, protection measures, cable marker posts, chambers and manholes, manhole covers and other equipment which is ancillary to the purposes of transmitting electricity along such cables to the National Grid substation;
- k) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials to access adjoining land for the purposes of the Works, the inspection, use, testing, maintenance, renewal, upgrading, replacement and removal of the cables, cable ducts and jointing works and connection into any adjacent cables and associated works and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rights; and



 I) construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway.
A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
 remainder of the Order land to: m)prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto); n) prevent anything to be done by way of hard surfacing of the Land with concrete of any kind or with any other material or surface whatsoever without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed if the proposed surfacing would not cause damage to relevant part of the authorised development nor make it materially more difficult or expensive to maintain the authorised development); o) prevent anything to be done by way of mole draining or excavation of any kind in the Land nor any activities which would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.7m for the purposes of arable farming) or are required to be carried out by National Grid in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the Land; p) prevent the planting or growing within the Land of any trees,
b) prevent the planting of growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development);
 q) prevent anything being done which may interfere with free flow and passage of electricity or telecommunications through the cables or support for the authorised development; and



	 r) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development.
	Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to:
G - Permanent Accesses	 a) pass and re-pass with or without vehicles, plant, machinery, apparatus, equipment and materials to access adjoining land for the purposes of the Works, the inspection, use, testing, maintenance, renewal, upgrading, replacement and removal of the cables, cable ducts and jointing works and connection into any adjacent cables and associated works and make such investigations in or on the Land which is ancillary for the purposes of exercise of the rights; b) construct, use, maintain and improve a permanent means of access including visibility splays, and retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway; c) place and use plant, machinery, structures and temporary structures within the Land, and to erect temporary signage and provide measures for the benefit of public and personnel safety; d) alter, fell, lop, cut, coppice, uproot and replant trees or hedges or other vegetation which now or hereafter may be present on the Land for the purpose of enabling the right to pass and re-pass to adjoining land; and e) repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs and landscaping.
	A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
	f) prevent anything to be done in or upon the Land or any part thereof for the purpose of the erection of any buildings or construction erection or works of any kind (including the foundations or footings thereto);g) prevent anything to be done by way of mole draining or
	excavation of any kind in the Land nor any activities which



	 would alter, increase or decrease ground cover or soil levels in any manner whatsoever without the consent in writing of the undertaker save as are reasonably required for agricultural activities (being ploughing to no deeper than 0.7m for the purposes of arable farming) or are required to be carried out by National Grid in order to exercise their statutory functions or rights in relation to their apparatus (if any) within the Land; h) prevent the planting or growing within the Land of any trees, shrubs or underwood without the consent in writing of the undertaker (such consent not to be unreasonably withheld or delayed provided that the proposed trees, shrubs or underwood would not cause damage to the relevant part of the authorised development nor make it materially more difficult or expensive to access the relevant part of the authorised development); and i) prevent carrying out operations or actions (including but not limited to blasting and piling) which may obstruct, interrupt, or interfere with the exercise of the rights or damage the authorised development.
н	Not used
I - Permanent landscape and ecological mitigation works	 Rights for the purposes of the construction, installation, operation, maintenance and decommissioning of the authorised development to: a) install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seedlings and other ecological measures together with the right to maintain, inspect and replant such trees, shrubs, landscaping for the period of aftercare and maintenance of such planting (as set out in the landscape and ecological management plan approved under this Order); b) install, execute, implement, retain, repair, improve, renew,
	 relocate, maintain and carry out mitigation, maintenance, remediation works for environmental or ecological mitigation or enhancement works, including the creation of hibernacula and installation of bat boxes, temporary works and the installation of temporary barriers for the protection of fauna; c) carry out environmental mitigation, remediation and enhancement works;



 d) erect fencing, gates or other boundary features; and e) take access to the Land and to use it to effect access to or egress from adjacent land, without or without vehicles and equipment, for the purposes of exercising any right vested in the undertaker, maintaining, replacing or altering for environmental or ecological mitigation or enhancement works, or for the purposes of undertaking surveying or monitoring activities required for the purposes of the authorised development.
A restrictive covenant over the Land for the benefit of the remainder of the Order land to:
 f) prevent any activity which would in the reasonable opinion of the undertaker result in the disturbance of ecological mitigation areas or areas of habitat creation or areas of reinstatement where planting requires time to re-establish including any ploughing or grazing without the prior written consent of the undertaker; g) prevent the uprooting, removal, destruction or damaging of any plant, bush, shrub, sapling or tree planted on the Land by the undertaker during the period within which the undertaker is responsible for the aftercare and maintenance of planting (as set out in the landscape and ecological management plan approved under this Order); and h) control or restrict access and use where such control or restriction is necessary for ecological or environmental mitigation or enhancement works to establish, including preventing access by the public on foot during any period where planting required under this Order is establishing (set out in the landscape and ecological management plan
out in the landscape and ecological management plan approved under this Order).



Doc. No. C282-DM-Z-GA-00003 Rev. H

5 Book of Reference

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk					
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-001	Rights A	5165 square metres of foreshore land seaward of the Mean High Water (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Kings's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted by lease dated 18 November 2009) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-002	Rights A	1523 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) rights of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-002 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right ty of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-002 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-003	Rights A	32 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1 right to use the land, and/or (iii) rights		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-003 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)			Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

				ERENCE - PART 1		
	Land which is propo	osed to be subject to: (I) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right of Norfolk	is to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-003 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-004	Rights A	1195 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG (as reputed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN	North Norfolk District Council Council Offices Holt Road CROMER Norfolk NR27 9EN Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-005	Rights A	366 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG
					Keiling HOLT Norfolk NR25 7EG	(as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure I ms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-005 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP7))	Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)
						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) ו	is Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights / of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-005 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	FERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 ulation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-006	Rights A	814 square metres of beach and public footpath (Weybourne FP7) (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk	hts to carry out protective works (Reg	
			Qualifying persons under Regula	Category 1 ation 7(1)(a) of the Infrastructure I	Planning (Applications: Prescribed	Category 2
				ms and Procedures) Regulations		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-006 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP7))	Lucid Trustee Services Limited 6th Floor Number 1 Building 1-5 London Wall Buildings London EC2M 5PG (as mortgagee for TC Dudgeon OFTO PLC) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

		·	BOOK OF RE	ns Offshore Wind Farm Developmer FERENCE - PART 1		
	Land which is prope	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) Count	right to use the land, and/or (iii) righ ty of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ulation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-006 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-007	Rights A	4767 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-007 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP7))	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1					
			Qualifying persons under Regul Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-007 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)			



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) righ ty of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-007 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-008	Rights A	765 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) rig	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regula	Category 1 $(1)(a)$ of the Infrastructure F	Planning (Applications: Prescribed	Category 2
				ms and Procedures) Regulations		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-008 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP7))	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right		ulation 7(1)(a))
			County	of Norfolk		
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-008 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-009	Rights A	2135 square metres of beach and public footpath (Weybourne FP7)(Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG (as reputed owner) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP7))	Unknown			

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-010	Temporary Possession	1680 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG (as reputed owner) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP7))	Unknown			

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-011	Temporary Possession	652 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

			County			
			Qualifying persons under Regula Fori	ation 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-011 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-011 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)
01-012	Rights B	36 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)
					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-012 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-012 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)			

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-013	Temporary Possession	57 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) ri County	of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-013 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-013 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-014	Rights B	4655 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-014 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) r	ERENCE - PART 1 right to use the land, and/or (iii) rights / of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-014 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-014 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-015	Rights B	393 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1			
				tion 7(1)(a) of the Infrastructure ns and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Extent of acquisition or Land Plans use Description of la	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-015 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Lucid Trustee Services Limited 6th Floor Number 1 Building 1-5 London Wall Buildings London EC2M 5PG (as mortgagee for TC Dudgeon OFTO PLC)



Doc. No. C282-DM-Z-GA-00003

			County	v of Norfolk Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
01-015 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for leas contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use an maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-015 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-016	Rights B	33 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-016 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-016 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
	Qualifying person			Category 1 ying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-016 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe over part)
01-017	Rights C	51 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-017 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Lucid Trustee Services Limited 6th Floor Number 1 Building 1-5 London Wall Buildings London EC2M 5PG (as mortgagee for TC Dudgeon OFTO PLC)



Doc. No. C282-DM-Z-GA-00003

			County	v of Norfolk Category 1		[
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forma and Procedures) Regulations 2009
01-017 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for leas contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use an maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-017 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-018	Rights C	7854 square metres of access track (north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order nts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-018 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dereham and District Angling Club 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads)



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) ri County	of Norfolk Category 1		
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-018 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads) Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
				ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or Uescription of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009	
01-018 cont'd						Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for leas contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use ar maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-018 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-019	Temporary Possession	10156 square metres of access track and public footpath (Weybourne FP7) (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)
					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)
					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP7))	Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)

Page 80 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (1) po	wers of compulsory acquisition, (ii) ri County	of Norfolk Category 1	s to carry out protective works (Reg	
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-019 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) ri County	of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-019 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-019 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-020	Rights C	9694 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) righ / of Norfolk Category 1	ts to carry out protective works (Reg	ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-020 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

			County Qualifying persons under Regul	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	For Owners or Reputed Owners	ms and Procedures) Regulations 20 Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-020 cont'd						Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dereham and District Angling Clu 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
			Category 1			
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-020 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) rights of Norfolk Category 1	s to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-020 cont'd						Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads) Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads) Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
	Category 1					
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-020 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) rig	RENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula Forr	tion 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-020 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-020 cont'd						The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access)
01-021	Rights B	75 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)
					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

		County of Norfolk Category 1				
				ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
01-021 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use arm maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) r	Is Offshore Wind Farm Development ERENCE - PART 1 ight to use the land, and/or (iii) rights of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-021 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop	osed to be subject to: (i) po		ty of Norfolk	nts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	ulation 7(1)(a) of the Infrastructure F orms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-022	Rights B	24 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)
					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019)
						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

			County			
			Qualifying persons under Regula Forr	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-022 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-022 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-023	Rights B	630 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			·			
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure P ms and Procedures) Regulations 2	lanning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-023 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

			County	Category 2		
				ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		0,1
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-023 cont'd						Lucid Trustee Services Limited 6th Floor Number 1 Building 1-5 London Wall Buildings London EC2M 5PG (as mortgagee for TC Dudgeon OFTO PLC) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

		County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-023 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) Count	y of Norfolk	hts to carry out protective works (Reg	julation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	For Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-024	Rights B	3359 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February

Page 99 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
			Category 1			
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-024 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-024 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ ty of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-025	Rights B	4474 square metres of grassland and access track (Weybourne Military Camp) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of rights granted)
						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) Count	y of Norfolk	hts to carry out protective works (Reg	julation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-026	Rights B	265 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019)

Page 103 of 1368



Doc. No. C282-DM-Z-GA-00003

			County	County of Norfolk Category 1		
			Qualifying persons under Regula Fori	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-026 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of an option to grant lease dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) r	is Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights / of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-026 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

		, (/)		y of Norfolk Category 1	ts to carry out protective works (Reg	
				lation 7(1)(a) of the Infrastructure Plorms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-027	Rights B	304 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-027 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Lucid Trustee Services Limited 6th Floor Number 1 Building 1-5 London Wall Buildings London EC2M 5PG (as mortgagee for TC Dudgeon OFTO PLC) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	is Offshore Wind Farm Developmen ERENCE - PART 1 ight to use the land, and/or (iii) right / of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
			Fo	rms and Procedures) Regulations 20)09	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-027 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				y of Norfolk Category 1 lation 7(1)(a) of the Infrastructure Pl	anning (Anglianting), Described	Category 2
				rms and Procedures) Regulations 2		
Number on Extent of acquisition or Land Plans use		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-028	Rights B	5462 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and
						maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) ri County	ght to use the land, and/or (iii) right of Norfolk Category 1	s to carry out protective works (Reg	ulation 7(1)(a))
				ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-028 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) r	Is Offshore Wind Farm Development ERENCE - PART 1 ight to use the land, and/or (iii) rights of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regul Fo	Category 1 ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-028 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope	osed to be subject to: (i) po		ty of Norfolk	nts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 ulation 7(1)(a) of the Infrastructure F prms and Procedures) Regulations 3	Planning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-029	Rights D	9686 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)
					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)
						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-029 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-029 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-030	Rights D	8080 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo			ght to use the land, and/or (iii) righ of Norfolk	its to carry out protective works (rreg	
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-030 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of rights granted by lease dated 13 March 2018)
						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electricity cables and junction point)
						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribe Forms and Procedures) Regulations 2009				Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-030 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) Unknown (in respect of exceptions, reservations and declarations)
01-031	Temporary Possession	4437 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-031 cont'd					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-031 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) right	RENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula Forr	tion 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-031 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order s to carry out protective works (Regi	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-032	Temporary Possession	21 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP	NONE	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)
			Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG		Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)
					Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-032 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-032 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) Unknown (in respect of right of access and right of passage of services through conducting media)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

			Count	y of Norfolk Category 1		
				lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2
Number on Extent of acquisition or Land Plans use		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-033	Temporary Possession	9105 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and junction point) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatu



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prese				Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	For Owners or Reputed Owners	rms and Procedures) Regulations 20	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-033 cont'd						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
		Category 1				
				ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2		Category 2
Number on Extent of acquisition of Land Plans use	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-033 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	used to be subject to: (i) po	owers of compulsory acquisition, (ii) right	RENCE - PART 1 ght to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula Forr	tion 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Extent of acquisition or Land Plans use Description	Description of land	escription of land Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-033 cont'd						TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
				rms and Procedures) Regulations 2		0,
Number on Extent of acquisition or Land Plans use		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-034	Rights D	22426 square metres of grassland (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG Michael Savory	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham
					Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)
						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-034 cont'd						Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) rights of Norfolk	to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-034 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-034 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-035	Rights C	15 square metres of public highway (The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right ν of Norfolk	to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-035 cont'd						Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-035 cont'd						Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

		The Shering		ns Offshore Wind Farm Developmen ERENCE - PART 1	t Consent Order	
	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii)	right to use the land, and/or (iii) right y of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
			Category 1			
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-035 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 28 February 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii) r	is Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights / of Norfolk		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-035 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-035 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)
01-036	Rights C	156 square metres of public highway (The Street, A149) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-036 cont'd			Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG (in respect of sub soil beneath half width of public highway) Louise Hubbard Home Farm The Street Weybourne HOLT			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP
			NR25 7SY (as executor of the estate of the late Monica Ann Middleton) (in respect of sub soil beneath half width of public highway)			(in respect of telecommunications)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regul For	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-036 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY (in respect of sub soil beneath half width of public highway) Unknown			Unknown			

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-037	Rights C	29 square metres of access track (north of The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	Annemarie Wharton Foxhills Weybourne HOLT NR25 7EH (as reputed owner)	NONE	Unknown	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)			
			Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG (as reputed owner)			Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016)			

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				ERENCE - PART 1		
	Land which is propo	sed to be subject to: (i) po	owers of compulsory acquisition, (ii) Count	right to use the land, and/or (iii) rights y of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
			Category 1			
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-037 cont'd			Stephen Harry Wharton Foxhills Weybourne HOLT NR25 7EH (as reputed owner) Unknown			Dereham and District Angling Club 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-037 cont'd						Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads) Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads) Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads)



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) ri County			
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-037 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-037 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access) Unknown

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-038	Rights D	707 square metres of public highway and verges (The Street, A149) (North Norfolk District Council)	Alexa Mary Josephine Kidd c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET (in respect of sub soil beneath half width of public highway) Elizabeth Jane Morgan 45 Hanson Drive FOWEY PL23 1ET (in respect of sub soil beneath half width of public highway) Louise Anne Savory Muckleburgh House Weybourne Road Kelling HOLT Norfolk NR25 7EG (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-038 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Paul Anthony Hugh Kidd c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET (in respect of sub soil beneath half width of public highway) Rupert Gervase William Kidd c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET (in respect of sub soil beneath half width of public highway)			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-038 cont'd			Sarah Bain c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET (in respect of sub soil beneath half width of public highway) Unknown			
01-039	Temporary Possession	1 square metres of access track (south of The Street, A149) (North Norfolk District Council)	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH (as reputed owner) Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH (as reputed owner)	NONE	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regul Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-039 cont'd			Unknown		Unknown				
01-040	Temporary Possession	3 square metres of access track (south of The Street, A149) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner, as executor of the estate of the late Monica Ann Middleton) Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner, as executor of the estate of the late Monica Ann Middleton)	NONE	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a))							
			County	County of Norfolk Category 1				
				lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
01-040 cont'd			Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner)		Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		gham Shoal and Dudgeon Extensior BOOK OF REF owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
	-			y of Norfolk	, , , , , , , , , , , , , , , , , , , ,	-
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-041	Temporary Possession	285 square metres of access track and land (south of The Street, A149) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	NONE	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fc	lation 7(1)(a) of the Infrastructure Planner and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-041 cont'd						Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)
01-042	Rights D	7698 square metres of hedgerow and arable land (south of The Street, A149) (North Norfolk District Council)	Alexa Mary Josephine Kidd c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-042 cont'd			Elizabeth Jane Morgan 45 Hanson Drive FOWEY PL23 1ET			
			Paul Anthony Hugh Kidd c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET			
			Rupert Gervase William Kidd c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET			
			Sarah Bain c/o Elizabeth Morgan 45 Hanson Drive FOWEY PL23 1ET			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
	Category 1					
				lation 7(1)(a) of the Infrastructure Planns and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-043	Temporary Possession	486 square metres of access track, woodland and arable land (south of The Street, A149) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	NONE	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Paul Brian Middleton and Monica Ann Middleton) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-044	Rights D	36555 square metres of arable land and hedgerow (north of Holgate Hill) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	NONE	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Paul Brian Middleton and Monica Ann Middleton)



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-044 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)
02-001	Temporary Possession	2057 square metres of arable land (north of Holgate Hill) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton)	NONE	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		gham Shoal and Dudgeon Extensior BOOK OF REF owers of compulsory acquisition, (ii) i	ERENCE - PART 1		ulation 7(1)(a))
		, ,,,		/ of Norfolk		
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-001 cont'd			Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY			Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Paul Brian Middleton and Monica Ann Middleton) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-002	Rights D	442 square metres of public highway and verges (Holt Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of sub soil beneath half width of public highway) Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-002 cont'd			Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)						

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-002 cont'd			Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY (in respect of sub soil beneath half width of public highway) Unknown				
02-003	Temporary Possession	2844 square metres of arable land, hedgerow and verges (north of Holgate Hill) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton)	NONE	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-003 cont'd			Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY			Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Paul Brian Middleton and Monica Ann Middleton)
						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-004	Rights D	583 square metres of public highway and verge (Holgate Hill) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) (in respect of sub soil beneath half width of public highway) Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Plance and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-004 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY (in respect of sub soil beneath half width of public highway)			
02-005	Rights D	816 square metres of arable land and hedgerow (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton)	NONE	Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		gham Shoal and Dudgeon Extension BOOK OF REF owers of compulsory acquisition, (ii) r	ERENCE - PART 1		(1)	
	Land which is propo			of Norfolk	s to carry out protective works (Reg		
				Category 1			
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-005 cont'd			Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as executor of the estate of the late Monica Ann Middleton) Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY			Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) National Westminster Bank PLC 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Paul Brian Middleton and Monica Ann Middleton and Monica Ann Middleton) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-006	Rights D	323 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (as reputed owner) Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner, as executor of the estate of the late Monica Ann Middleton)	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-006 cont'd			Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner, as executor of the estate of the late Monica Ann Middleton)		Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY	
			Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner) Unknown		Paul Brian Middleton Home Farm The Street Weybourne HOLT NR25 7SY Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-007	Temporary Possession	305 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (as reputed owner) Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner, as executor of the estate of the late Monica Ann Middleton)	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ Joanne Middleton Home Farm The Street Weybourne HOLT NR25 7SY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-007 cont'd			Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY (as reputed owner, as executor of the estate of the late Monica Ann Middleton) Paul Brian Middleton Home Farm The Street		Louise Hubbard Home Farm The Street Weybourne HOLT NR25 7SY Paul Brian Middleton Home Farm The Street	Unknown
			Weybourne HOLT NR25 7SY (as reputed owner) Unknown		Weybourne HOLT NR25 7SY Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-008	Temporary Possession	45 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (as reputed owner)	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus)
			Unknown		Unknown	Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)
						No Ni (in ma

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-009	Temporary Possession	730 square metres of track, arable land and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) Unknown (in respect of restrictive covenants stated in conveyance dated 23rd

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-010	Rights D	45859 square metres of arable land, grassland, hedgerow, watercourse (Spring Beck) and verges (east of Holt Road and west of Station Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted in deed dated 12th August 2011)Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted in deed dated 12th August 2011)Unknown (in respect of restrictive covenants stated in conveyance dated 23rd December 1954)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				llation 7(1)(a) of the Infrastructure PI orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-011	Rights D	74 square metres of public highway and verges (Station Road) (North Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE
02-012	Rights D	871 square metres of public highway and verges (Station Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			Unknown			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-013	Rights D	952 square metres of public highway and verges (Station Road) (North Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-014	Temporary Possession	2367 square metres of arable land, access track and verges (east of Station Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)
02-015	Rights D	60663 square metres of arable land, access track, woodland and hedgerows (east of Station Road) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-015 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)
03-001	Temporary Possession	214 square metres of public highway and verges (Sheringham Road, A149) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-001 cont'd			Unknown			
03-002	Temporary Possession	8425 square metres of access track and verges (south of Sheringham Road, A149) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)
03-003	Rights D	653 square metres of railway track, compound and verges (north east of Weybourne Train Station) (North Norfolk District Council)	North Norfolk Railway PLC Sheringham Railway Station Sheringham Norfolk NR26 8RA (as reputed owner)	NONE	North Norfolk Railway PLC Sheringham Railway Station Sheringham Norfolk NR26 8RA	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants)
			Unknown			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-004	Rights D	1829 square metres of railway track, compound and verges (north east of Weybourne Train Station) (North Norfolk District Council)	North Norfolk Railway PLC Sheringham Railway Station Sheringham Norfolk NR26 8RA (as reputed owner) Unknown	NONE	North Norfolk Railway PLC Sheringham Railway Station Sheringham Norfolk NR26 8RA	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants)
03-005	Temporary Possession	7893 square metres of arable land (east of Sandy Hill Lane) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-005 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
03-006	Temporary Possession	77 square metres of public highway and verges (Sandy Hill Lane) (North Norfolk District Council)	Anne Pope Springs Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-006 cont'd			Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of sub soil beneath half width of public highway) Keith Ranald Nichols Sandy Hill House Sandy Hill House Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of sub soil beneath half width of public highway)			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-006 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Timewell Properties Limited Bankside 300 Peachman Way Broadland Business Park NORWICH Norfolk NR7 0LB (in respect of sub soil beneath half width of public highway) Unknown			Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order its to carry out protective works (Reg	ulation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-007	Temporary Possession	22 square metres of access track (east of Sandy Hill Lane) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (as reputed owner) Keith Ranald Nichols Sandy Hill House Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Unknown	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-008	Rights D	15505 square metres of arable land, woodland, verges and track (east of Sandy Hill Lane) (North Norfolk District Council)	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE	Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ	NONE
03-009	Rights D	13406 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA	The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010	Rights D	15195 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd						Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)

Page 181 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd						Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

				v of Norfolk Category 1 ation 7(1)(a) of the Infrastructure Pla	anning (Applications: Prescribed	Category 2
				rms and Procedures) Regulations 20		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd						Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29 March 2016)



Doc. No. C282-DM-Z-GA-00003

				of Norfolk Category 1		Category 2
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	inning (Applications: Prescribed	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
03-010 cont'd						Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 25 March 2016)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right		ulation 7(1)(a))
			County	of Norfolk		
			Qualifying persons under Regul For	Category 1 ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd						The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd						William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)
03-011	Rights D	9070 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA	The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	NONE
04-001	Temporary Possession	1675 square metres of woodland and track (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA	The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	NONE

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-002	Temporary Possession	1869 square metres of woodland and track (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) rights		ulation 7(1)(a))
			Qualifying persons under Regu	y of Norfolk Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2(anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-002 cont'd						Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-002 cont'd						Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

			Qualifying persons under Regul	category 1 Category 1 ation 7(1)(a) of the Infrastructure Pla		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	For Owners or Reputed Owners	rms and Procedures) Regulations 20 Lessees or Tenants	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-002 cont'd						Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)



Doc. No. C282-DM-Z-GA-00003

			County	v of Norfolk Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
04-002 cont'd						Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 25 March 2016)



Doc. No. C282-DM-Z-GA-00003

			County	v of Norfolk Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-002 cont'd						The National Trust For Places O Historic Interest Or Natural Beau Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)



Doc. No. C282-DM-Z-GA-00003

				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed		
Number on Extent of acquisition or Land Plans use Description of land	Description of land	For Owners or Reputed Owners	ms and Procedures) Regulations 2 Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-002 cont'd						Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-003	Rights D	38365 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP6)) The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-003 cont'd						Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-003 cont'd						Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))	
				Category 1			
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-003 cont'd						Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	



Doc. No. C282-DM-Z-GA-00003

			County	r of Norfolk Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
04-003 cont'd						Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 25 March 2016)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
				rms and Procedures) Regulations 20		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-003 cont'd						The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-003 cont'd						Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-004	Temporary Possession	9 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP6)) The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	on or Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-004 cont'd						Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Extent of acquisition or Land Plans use Description of	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004 cont'd						Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media) Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

			County	v of Norfolk Category 1		1
			Qualifying persons under Regul For	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
04-004 cont'd						Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, rig to fell and transport trees and rig of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

				v of Norfolk Category 1		Category 2
				ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-004 cont'd						Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29 March 2016) Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004 cont'd						Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-004 cont'd						Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-005	Temporary Possession	13 square metres of public highway and verge (Sandy Hill Lane) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of sub soil beneath half width of public highway) Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-005 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL (in respect of sub soil beneath half width of public highway)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		llation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-005 cont'd			Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of sub soil beneath half width of public highway) Unknown			
04-006	Temporary Possession	43 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner)	NONE	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-006 cont'd			Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL (as reputed owner) Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL (as reputed owner) Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner)		Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP6)) Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-006 cont'd			Unknown		Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Unknown	
04-007	Temporary Possession	10 square metres of public highway (Sandy Hill Lane) (North Norfolk District Council)	Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-007 cont'd			Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL			Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Maureen Olive Amies and Richard John Robert Amies)	
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
						Unknown (in respect to a right of way/passage over the track)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
Number on Extent of acquisition or Land Plans use			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-008	Temporary Possession	13 square metres of public highway (Sandy Hill Lane) (North Norfolk District Council)	Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Maureen Olive Amies and Richard John Robert Amies)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-008 conťd						Unknown (in respect to a right of way/passage over the track)
04-009	Temporary Possession	2108 square metres of access track and verges (east of Sandy Hill Lane) (North Norfolk District Council)	Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT	NONE	Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Maureen Olive
			Norfolk NR25 6QL		Norfolk NR25 6QL	Amies and Richard John Robert Amies) Unknown (in respect to a right of way/passage over the track)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	it Consent Order is to carry out protective works (Regu	lation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-010	Temporary Possession	35 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Unknown	NONE	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-011	Temporary Possession	1355 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP6)) The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	is Offshore Wind Farm Developmen ERENCE - PART 1 ight to use the land, and/or (iii) right / of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-011 cont'd						Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) r	is Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights / of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	nning (Applications: Prescribed	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-011 cont'd						Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

			County Qualifying persons under Regul	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	For Owners or Reputed Owners	rms and Procedures) Regulations 20 Lessees or Tenants	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-011 cont'd						Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)



Doc. No. C282-DM-Z-GA-00003

			County	r of Norfolk Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
04-011 cont'd						Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 25 March 2016)



Doc. No. C282-DM-Z-GA-00003

				v of Norfolk Category 1 ation 7(1)(a) of the Infrastructure Pla	nning (Annlications: Proceribed	Category 2
				rms and Procedures) Regulations 20		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-011 cont'd						The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)



Doc. No. C282-DM-Z-GA-00003

				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-011 cont'd						Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and righ of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	llation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-012	Temporary Possession	67 square metres of hedgerow and verge (east of Sandy Hill Lane) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Unknown	NONE	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Unknown	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				ERENCE - PART 1				
	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) Count	right to use the land, and/or (iii) right y of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))		
				Category 1				
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
04-013	Temporary Possession	465 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weybourne FP6)) The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)		



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-013 cont'd						Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-013 cont'd						Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

			County Qualifying persons under Regul	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	For Owners or Reputed Owners	rms and Procedures) Regulations 20 Lessees or Tenants	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-013 cont'd						Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)



Doc. No. C282-DM-Z-GA-00003

	County of Norfolk Category 1						
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009	
04-013 cont'd						Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 25 March 2016)	



Doc. No. C282-DM-Z-GA-00003

				County of Norfolk Category 1		
				ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-013 cont'd						The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-013 cont'd						Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ERENCE - PART 1 right to use the land, and/or (iii) righ		lation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
Rights D	277 square metres of woodland and track (east of Sandy Hill Lane) (North Norfolk District Council)	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner)	NONE	Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	Unknown
	Extent of acquisition or use	Land which is proposed to be subject to: (i) possible Extent of acquisition or use Description of land Rights D 277 square metres of woodland and track (east of Sandy Hill Lane) (North Norfolk	Extent of acquisition or use Description of land Qualifying persons under Regu Gualifying persons under Regu Powers Rights D 277 square metres of woodland and track (east of Sandy Hill Lane) (North Norfolk District Council) Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner)	BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) right county of Norfolk Extent of acquisition or use Description of land Rights D Description of land Woodland and track (east of Sandy Hill Lane) (North Norfolk District Council) Christopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW	Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulations County of Norfolk Extent of acquisition or use Description of land Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Rights D 277 square metres of woodland and track (east of Sandy Hill Lane) (North Norfolk District Council) Ohristopher Robert Esh Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) NONE Christopher Rubert Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner) Suzanne Ruth Tansley Wild Wood Lodge Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (as reputed owner)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Regi	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-015	Rights D	46631 square metres of arable land, hedgerow and verges (east of Sandy Hill Lane) (North Norfolk District Council)	Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL	NONE	Maureen Olive Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL Richard John Robert Amies Breck Farm Weybourne Road Bodham HOLT Norfolk NR25 6QL	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Maureen Olive Amies and Richard John Robert Amies) Unknown (in respect to a right of way/passage over the track)
04-016	Rights D	2459 square metres of arable land (east of Gipsies' Lane) (North Norfolk District Council)	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-017	Rights D	38597 square metres of arable land, grassland and hedgerow (east of Gipsies' Lane) (North Norfolk District Council)	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
05-001	Temporary Possession	182 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	Andrew Wegg Uphill Church Lane Gresham NORWICH NR11 8RT (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-001 cont'd			David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (in respect of sub soil beneath half width of public highway) Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-001 cont'd			Steven Wegg 90 Drabblegate Aylsham NORWICH NR11 6LR (in respect of sub soil beneath half width of public highway) Susan Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (in respect of sub soil beneath half width of public highway) Trevor Wegg 13 Campion Way SHERINGHAM NR26 8UN (in respect of sub soil beneath half width of public highway) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-002	Temporary Possession	642 square metres of arable land, hedgerow and verge (east of Sandy Hill Lane) (North Norfolk District Council)	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
05-003	Temporary Possession	14 square metres of arable land, hedgerow and verge (east of Sandy Hill Lane) (North Norfolk District Council)	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-004	Rights D	2000 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (in respect of sub soil beneath half width of public highway) Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-004 cont'd			Unknown			
05-005	Temporary Possession	360 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (in respect of sub soil beneath half width of public highway) Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-005 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown
05-006	Rights D	4033 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (as reputed owner) Unknown	NONE	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-007	Rights D	1530 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	Lemur UK LLC 10960 Wilshire Blvd 11th Floor LOS ANGELES CA 90024 UNITED STATES OF AMERICA (in respect of right of access for maintenance)
05-008	Temporary Possession	6023 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	NONE	Elizabeth Jean Thurtle Laburnhum Farm Cromer Road Bodham HOLT Norfolk NR25 6QQ	Lemur UK LLC 10960 Wilshire Blvd 11th Floor LOS ANGELES CA 90024 UNITED STATES OF AMERICA (in respect of right of access for maintenance)
05-009	Rights D	34484 square metres of arable land and hedgerow (south of Holt Road, A148 and north of The Street) (North Norfolk District Council)	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (as reputed owner)	NONE	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-009 cont'd			Unknown		Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
05-010	Temporary Possession	655 square metres of access track and hedgerow (north of The Street) (North Norfolk District Council)	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (as reputed owner) Unknown	NONE	David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	l and which is prop		gham Shoal and Dudgeon Extensior BOOK OF REF owers of compulsory acquisition, (ii)	ERENCE - PART 1		(1)
	Land which is prope			y of Norfolk	is to carry out protective works (rieg	
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-011	Temporary Possession	27 square metres of public highway (The Street) (North Norfolk District Council)	AWE (Holdings) Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH (in respect of sub soil beneath half width of public highway) David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-011 cont'd			Unknown			
05-012	Rights D	811 square metres of public highway and verges (The Street) (North Norfolk District Council)	AWE (Holdings) Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH (in respect of sub soil beneath half width of public highway) David George Wegg Avenue Farm The Street Bodham HOLT NR25 6NW (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-012 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
05-013	Rights D	306 square metres of verges and hedgerow (south of The Street) (North Norfolk District Council)	AWE (Holdings) Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH (as reputed owner)	NONE	AWE (Holdings) Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)
			Unknown		Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Page 245 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-013 cont'd						Unknown
05-014	Temporary Possession	355 square metres of arable land, verges and hedgerow (south of The Street) (North Norfolk District Council)	AWE (Holdings) Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH Unknown (in respect of mines and minerals)	John McNeil Wilson Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH	John McNeil Wilson Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)
05-015	Rights D	39060 square metres of arable land and hedgerow (south of The Street) (North Norfolk District Council)	AWE (Holdings) Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH	John McNeil Wilson Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH	John McNeil Wilson Limited Manor Farm East Winch KINGS LYNN Norfolk PE32 1NH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-015 cont'd			Unknown (in respect of mines and minerals)			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
05-016	Rights D	23118 square metres of arable land and hedgerow (north of Norfolk Coast, A149) (North Norfolk District Council)	C J C Lee (Saxthorpe) Limited Great Farm Saxthorpe NORWICH Norfolk NR11 7BX	NONE	C J C Lee (Saxthorpe) Limited Great Farm Saxthorpe NORWICH Norfolk NR11 7BX	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for C J C Lee (Saxthorpe) Limited) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted) Unknown (in respect of right of access to lay and maintain electricity cables contained in transfer dated 28 July 1995)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-001	Temporary Possession	242 square metres of arable land and hedgerow (north of Norfolk Coast, A149) (North Norfolk District Council)	C J C Lee (Saxthorpe) Limited Great Farm Saxthorpe NORWICH Norfolk NR11 7BX	NONE	C J C Lee (Saxthorpe) Limited Great Farm Saxthorpe NORWICH Norfolk NR11 7BX	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for C J C Lee (Saxthorpe) Limited) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted) Unknown (in respect of right of access to lay and maintain electricity cables contained in transfer dated 28 July 1995)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1 right to use the land, and/or (iii) righ		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-002	Rights D	466 square metres of verges and public highway (Norfolk Coast, A149) (North Norfolk District Council)	C J C Lee (Saxthorpe) Limited Great Farm Saxthorpe NORWICH Norfolk NR11 7BX (in respect of sub soil beneath half width of public highway) John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-002 cont'd			Unknown			
06-003	Rights D	65 square metres of verge (Norfolk Coast, A149) (North Norfolk District Council)	John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP (as reputed owner) Unknown	NONE	John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP Unknown	Unknown
06-004	Temporary Possession	7320 square metres of private road and public footpath (Bodham FP8) (west of New Road, A149) (North Norfolk District Council)	John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP	NONE	John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-004 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Bodham FP8))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
06-005	Rights D	61348 square metres of arable land and public footpath (Bodham FP8) (Pond Farm, west of New Road, A149) (North Norfolk District Council)	John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP	NONE	John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pl rrms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-005 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Bodham FP8))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
06-006	Temporary Possession	920 square metres of arable land (south of New Road, A149) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Christine Frances Wright Hill House Kelling Road Lower Bodham HOLT NR25 6QH	Christine Frances Wright Hill House Kelling Road Lower Bodham HOLT NR25 6QH	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-006 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			
06-007	Rights D	549 square metres of public highway (New Road, Holt, A149) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) (in respect of public footpath (Bodham FP8))	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regu	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-007 cont'd			John Edwin Mack Pond Farm New Road Bodham HOLT NR25 6PP (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown						

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-007 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of sub soil beneath half width of public highway)			
06-008	Rights D	5637 square metres of arable land and hedgerow (south of New Road, Holt, A149) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Christine Frances Wright Hill House Kelling Road Lower Bodham HOLT NR25 6QH	Christine Frances Wright Hill House Kelling Road Lower Bodham HOLT NR25 6QH	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-008 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			
07-001	Rights D	16979 square metres of arable land and hedgerow (south east side of New Road, A149) (North Norfolk District Council)	David John Roger Hall Upwood Lodge Baconsthorpe Holt Norfolk NR25 6NB (as reputed owner)	NONE	David John Roger Hall Upwood Lodge Baconsthorpe Holt Norfolk NR25 6NB	Arthur Richard Gee The Bungalow Chestnut Farm Castle Road Baconsthorpe HOLT NR25 6L (in respect of right of way) BW Sipp Trustees Limited St. James House St. James House St. James Square CHELTENHAM Gloucestershire GL50 3PR (in respect of right of way)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
	Category 1					
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-001 cont'd						Margaret Thelma Gee The Bungalow Chestnut Farm Castle Road Baconsthorpe HOLT NR25 6L (in respect of right of way)
07-002	Rights D	18238 square metres of arable land and hedgerow (south of New Road, A149) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-003	Rights D	373 square metres of arable land and hedgerow (south of New Road, A149) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fc	lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-004	Rights D	778 square metres of public highway being Gresham Road (north of Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of sub soil beneath half width of public highway) Henry Charles Steel Thistley House Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-004 cont'd			Unknown William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of sub soil beneath half width of public highway)			
07-005	Rights D	5577 square metres of arable land and hedgerow (south east Side Of New Road, A149) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-006	Rights D	16784 square metres of arable land and hedgerow (north of Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield)
			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		Category 2
			Qualifying persons under Regul	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed	- 3 ,
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-006 cont'd						Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-006 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter) Unknown (in respect of drainage rights reserved by transfer dated 25 March 2008)
07-007	Temporary Possession	1133 square metres of arable land and private track situated (south east Side Of New Road, A149) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-008	Temporary Possession	101 square metres of private track (north of Gresham Road and on the south side of New Road, A149) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	NONE	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Unknown		Unknown	Unknown
07-009	Rights D	392 square metres of verges and public highway (Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regul For	Category 1 ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-009 cont'd			Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-009 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of sub soil beneath half width of public highway)			
07-010	Temporary Possession	606 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

		The Sherin	gham Shoal and Dudgeon Extension	ns Offshore Wind Farm Development ERENCE - PART 1	t Consent Order	
	Land which is propo	osed to be subject to: (i) p	owers of compulsory acquisition, (ii)		s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-010 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield)
						Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-010 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)
07-011	Temporary Possession	1788 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right / of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-011 cont'd						Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-011 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-012	Temporary Possession	4 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT (as reputed owner) Unknown	NONE	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT Unknown	Unknown
07-013	Temporary Possession	2 square metres of arable land (south of Gresham Road) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT (as reputed owner) Unknown	NONE	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT Unknown	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-014	Temporary Possession	2345 square metres of arable land (south of Gresham Road) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	NONE	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electric) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electric) Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-014 cont'd						William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)
07-015	Rights D	24769 square metres of arable land and hedgerow (south of Gresham Road) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	NONE	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-015 conťď						William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)
07-016	Rights D	8641 square metres of arable land, hedgerow and private track (north of Church Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

		County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed				Category 2
				ms and Procedures) Regulations 20		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-016 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			Barclays Security Trustee Limite 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louis Suffield)
						Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right t enter)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication cables)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-016 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)
07-017	Temporary Possession	194 square metres of arable land (north of Church Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				of Norfolk Category 1		Category 2
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		3 3 7
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-017 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			Barclays Security Trustee Limiter 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louis Suffield)
						Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right enter)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication cables)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-017 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)
07-018	Rights D	29 square metres of agricultural land (north of Church Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Cletheroe and Sons Manor Farm Church Lane Edgefield MELTON CONSTABLE Norfolk NR24 2AF	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Reg County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed							
Number on Land Plans	Extent of acquisition or use	Description of land	Fo Owners or Reputed Owners	rms and Procedures) Regulations 20 Lessees or Tenants	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
07-018 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to William Robert Bartle Edwards and Emma Louise Suffield) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)		



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fc	Category 1 Ilation 7(1)(a) of the Infrastructure Pla orms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-018 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)
07-019	Rights D	452 square metres of hedgerow and public highway (Church Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-019 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of sub soil beneath public highway)			Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

		The Sherin		ns Offshore Wind Farm Developmen FERENCE - PART 1	t Consent Order	
	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii)	right to use the land, and/or (iii) right y of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				llation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-020	Temporary Possession	435 square metres of arable land (south of Church Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Emma Louise Suffield and William Robert Bartle Edwards) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Emma Louise Suffield and William Robert Bartle Edwards) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	l and which is propo		BOOK OF REF	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
				y of Norfolk	s to daily out protocitive works (ridg	
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-021	Rights D	41848 square metres of arable land, hedgerow and public footpath (Matlask FP1) (south of Church Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Emma Louise Suffield and William Robert Bartle Edwards)
			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Matlask FP1))	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Emma Louise Suffield and William Robert Bartle Edwards)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
	Category 1					
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-001	Rights D	52920 square metres of arable land and hedgerow (north of Northfield Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
08-002	Temporary Possession	1129 square metres of arable land and hedgerow (Barningham Park, north of Northfield Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 ualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-002 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
09-001	Rights D	892 square metres of public highway and verges (Northfield Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-001 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of sub soil beneath public highway)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-002	Temporary Possession	519 square metres of arable land and public footpath (Plumstead FP11) (south east of Northfield Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Plumstead FP11))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)			



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-002 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)
09-003	Rights D	35596 square metres of arable land and public footpath (Plumstead FP11) (south east of Northfield Lane) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-003 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Plumstead FP11))	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-003 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)
09-004	Rights D	15597 square metres of arable land and hedgerow (west of The Street, A149) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-004 conťď			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
09-005	Temporary Possession	124 square metres of arable land (north of Matlaske Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Barningham Farms Limited 7 The Close NORWICH Norfolk NR1 4DJ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-006	Rights D	777 square metres of public highway and verges (Matlaske Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii) ו	ns Offshore Wind Farm Development ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-006 cont'd			William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of sub soil beneath public highway)			

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-007	Temporary Possession	196 square metres of arable land (south of Matlaske Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-008	Temporary Possession	2035 square metres of arable land (south of Matlaske Road t) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-009	Rights D	41722 square metres of arable land and hedgerow (south of Matlaske Road t) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Mott Radclyffe Farms Barningham Hall Barningham Park Matlaske NORWICH NR11 7HY	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-001	Rights D	31524 square metres of arable land and hedgerow (Ferndale Farm, south west of The Street, A149) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)	NONE	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Harry John Betts)
			Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG		Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables)
			Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)		Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-002	Rights D	88 square metres of hedgerow (Ferndale Farm, south west of The Street, A149) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner) Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Harry John Betts) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-003	Temporary Possession	550 square metres of arable land (Ferndale Farm, north of Little Barningham Road) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)	NONE	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Harry John Betts)
			Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG		Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables)
			Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)		Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-004	Rights D	347 square metres of public highway and verges (Little Barningham Road) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath public highway) Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-004 cont'd			Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-005	Rights D	71 square metres of verge (south west of Little Barningham Road) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner) Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner) Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)	NONE	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH NORWICH NORWICH NORWICH NORWICH NORMIA	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-005 conťď			Unknown		Unknown	
10-006	Temporary Possession	685 square metres of arable land (Ferndale Farm, south of Little Barningham Road) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner) Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	NONE	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Harry John Betts) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-006 cont'd			Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)		Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)
10-007	Rights D	19814 square metres of arable land and hedgerow (Ferndale Farm, south of Little Barningham Road) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)	NONE	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-007 cont'd			Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG		Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Harry John Betts)
			Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)		Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-008	Rights D	400 square metres of public highway and verge (Sweetbriar Lane) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath half width of public highway) Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-008 cont'd			Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT (in respect of sub soil beneath half width of public highway) Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)			Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-008 cont'd			Unknown			
10-009	Temporary Possession	1104 square metres of arable land and access track (south of Sweetbriar Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Arthur Thomas Steel Flat 29 Charleville Mansions Charleville Road LONDON W14 9JA (in respect of rights granted) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Christopher Edward Steel 56 Nasmyth Street LONDON W6 0HB



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-009 cont'd						Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of legal easements reserved) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right / of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-009 cont'd						The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant) Unknown (in respect of rights of way, wayleaves, drainage and watercourse easements) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of legal easements reserved)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-010	Rights D	26850 square metres of arable land and hedgerow (south of Sweetbriar Lane and west of Matlaske Road) (excluding all interests of the Crown) (North Norfolk District Council)	Henry Charles Steel Thistley House Thistley Lane Gosmore HITCHIN SG4 7QT	Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Arthur Thomas Steel Flat 29 Charleville Mansions Charleville Road LONDON W14 9JA (in respect of rights granted) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Christopher Edward Steel 56 Nasmyth Street LONDON W6 0HB (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-010 cont'd						Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of legal easements reserved) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-010 cont'd						The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant) Unknown (in respect of rights of way, wayleaves, drainage and watercourse easements) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of legal easements reserved)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-011	Rights D	1095 square metres of verge and public highway (Matlaske Road) (North Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE
10-012	Rights D	462 square metres of verge and public highway (Matlaske Road) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Page 314 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-012 cont'd			Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath half width of public highway) Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (in respect of sub soil beneath half width of public highway)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-012 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
10-013	Rights D	10291 square metres of arable land and hedgerow (Ferndale Farm, Matlaske Road) (North Norfolk District Council)	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner) Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	NONE	Elinor Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Harry John Betts)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
		Category 1				Cotomer 2
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-013 cont'd			Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG (as reputed owner)		Harry John Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG Margaret Elizabeth Betts Ferndale Farm The Street Little Barningham NORWICH Norfolk NR11 7AG	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)
10-014	Rights D	53753 square metres of arable land, hedgerow and public footpath (Itteringham FP3) (south east of Matlaske Road) (North Norfolk District Council)	Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Itteringham FP3))	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Oak Park Farms Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) ri County	ght to use the land, and/or (iii) rig of Norfolk	hts to carry out protective works (Reg	ulation 7(1)(a))
				ation 7(1)(a) of the Infrastructure ms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-014 cont'd					Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	esed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
10-014 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to I and maintain gas pipeline) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved an granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-001	Temporary Possession	12 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Oak Park Farms Limited)			
						Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline)			



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-001 cont'd						Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe) Michael John Chantler Compass House Lypiatt Road CHELTENHAM GL50 2QJ (in respect of rights reserved) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-001 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved and granted)
11-002	Temporary Possession	1298 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Oak Park Farms Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-002 cont'd						Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-002 cont'd						Michael John Chantler Compass House Lypiatt Road CHELTENHAM GL50 2QJ (in respect of rights reserved) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-002 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved and granted)
11-003	Rights D	357 square metres of hedgerow (south east of Matlaske Road) (North Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH Unknown (in respect of mines and minerals)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-004	Rights D	672 square metres of public highway and verge (Matlaske Road) (North Norfolk District Council)	Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
			(as highway authority) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	lation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-005	Rights D	620 square metres of verge and hedgerow (Matlaske Road) (North Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH Unknown (in respect of mines and minerals)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
11-006	Rights D	28631 square metres of arable land and hedgerow (Mere north west of Matlaske Road) (North Norfolk District Council)	Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	NONE	Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of easement contained in deed dated 12 September 2011)



Doc. No. C282-DM-Z-GA-00003

			· · · · · ·	of Norfolk Category 1		Category 2
				tion 7(1)(a) of the Infrastructure F ns and Procedures) Regulations	Planning (Applications: Prescribed 2009	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-006 conťd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus)
						Nicholas Benjamin Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of rights stated in conveyance dated 5 November 1982)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-006 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of easement contained in deed dated 12 September 2011)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	ilation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-006 cont'd						Seth Elton Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of right of access to conduct drainage maintenance stated in conveyance dated 5 November 1982)
11-007	Rights D	40407 square metres of arable land and hedgerow Mere Farm and Wood Farm, north west of Matlaske Road) (North Norfolk District Council)	Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	NONE	Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of easement contained in deed dated 12 September 2011)



Doc. No. C282-DM-Z-GA-00003

		The Shering	gham Shoal and Dudgeon Extensior BOOK OF REE	s Offshore Wind Farm Developmen ERENCE - PART 1	t Consent Order	
	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii)		s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-007 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Nicholas Benjamin Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of rights stated in conveyance dated 5 November 1982) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) rights of Norfolk Category 1	s to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-007 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of easement contained in deed dated 12 September 2011) Seth Elton Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of right of access to conduct drainage maintenance stated in conveyance dated 5 November 1982)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rrms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-008	Rights D	5466 square metres of arable land and hedgerow (north west of Matlaske Road) (North Norfolk District Council)	Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right ty of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-008 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)
12-001	Temporary Possession	865 square metres of arable land (north west of Matlaske Road) (North Norfolk District Council)	Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-001 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)
12-002	Rights D	1592 square metres of public highway, verges and hedgerow (Matlaske Road) (North Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) p	owers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-002 cont'd			Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of sub soil beneath public highway)			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus
			Unknown			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication: Scira Offshore Energy Limited
						Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to la and maintain electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ sy of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Ilation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2	anning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-002 cont'd						Unknown
12-003	Rights D	173 square metres of wooded land (south east of Matlaske Road) (North Norfolk District Council)	Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to la and maintain electricity cables) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Degu	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed		
				rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-003 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)
12-004	Rights D	8386 square metres of arable land and hedgerow (south east of Matlaske Road) (North Norfolk District Council)	Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-004 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-004 cont'd						Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)
12-005	Temporary Possession	792 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	it Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-005 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)
12-006	Rights D	40552 square metres of arable land, hedgerow and public footpath (Corpusty FP17) (Wood Farm, south of Matlaske Road) (North Norfolk District Council)	Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	NONE	Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) right	ERENCE - PART 1 ght to use the land, and/or (iii) rig of Norfolk	hts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure ms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-006 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Corpusty FP17))	Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting and sporting rights) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Sharon Brooks Wood Farm Saxthorpe Road Itteringham NORWICH NR11 7BG (in respect of private water supply

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-007	Rights D	18286 square metres of arable land and public footpath (Corpusty FP16) (south east of Matlaske Road) (North Norfolk District Council)	Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU	NONE	Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Corpusty FP16))	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-007 conťd						Unknown (in respect of restrictive covenants)
12-008	Rights D	50061 square metres of arable land, hedgerow and public footpath (Corpusty FP16) (Lake Farm, north of Aylsham Road, B1354) (North Norfolk District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Corpusty FP16)) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				right to use the land, and/or (iii) righ ty of Norfolk Category 1		
				ulation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-001	Rights D	4758 square metres of arable land and hedgerow (Hall Farm, Aylsham Road, B1354) (North Norfolk District Council)	Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU	NONE	Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) (in respect of a right of pre- emption contained in an Agreement dated 15 October 2010)
13-002	Temporary Possession	1189 square metres of private highway and arable land (Lake Farm, north of Aylsham Road, B1354) (North Norfolk District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed					Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Fo Owners or Reputed Owners	rms and Procedures) Regulations 2	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-003	Rights D	479 square metres of arable land and hedgerow (Hall Farm, north of Aylsham Road, B1354) (North Norfolk District Council)	Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Bolton Agnew)
13-004	Rights D	1112 square metres of verge and public highway (Blickling Road, B1354) (North Norfolk District Council)	Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-004 cont'd			Elizabeth Glover Saxthorpe Hall Aylsham Road Saxthorpe NR11 7DE Norfolk (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Robert Glover Saxthorpe Hall Aylsham Road Saxthorpe NR11 7DE Norfolk (in respect of sub soil beneath half width of public highway)			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-004 cont'd			Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of sub soil beneath half width of public highway) Unknown			
13-005	Rights D	76 square metres of verge, hedgerow and arable land (Lake Farm, north of Aylsham Road, B1354) (North Norfolk District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-006	Rights D	6505 square metres of hedgerow and arable land (Saxthorpe Hall, south of Aylsham Road, B1354) (North Norfolk District Council)	Elizabeth Glover Saxthorpe Hall Aylsham Road Saxthorpe NR11 7DE Norfolk	NONE	Elizabeth Glover Saxthorpe Hall Aylsham Road Saxthorpe NR11 7DE Norfolk	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Robert Glover Saxthorpe Hall Aylsham Road Saxthorpe NR11 7DE Norfolk		Robert Glover Saxthorpe Hall Aylsham Road Saxthorpe NR11 7DE Norfolk	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Robert Glover and Elizabeth Glover)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

		1	Coun	y of Norfolk Category 1		
				ulation 7(1)(a) of the Infrastructure P prms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-007	Rights D	27 square metres of public highway and verge (Aylsham Road, B1354) (Broadland District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Saltcarr Farms Limited)
						Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU (as beneficiary of pre-emption agreement dated 28 September 2010)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

			BOOK OF REF	ns Offshore Wind Farm Developmen ERENCE - PART 1		
	Land which is propo	osed to be subject to: (I) po	owers of compulsory acquisition, (ii) Count	right to use the land, and/or (iii) right y of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-007 cont'd						Unknown (in respect of rights reserved and granted)
13-008	Temporary Possession	30 square metres of public highway and verge (Aylsham Road, B1354) (Broadland District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Saltcarr Farms Limited) Bolton Agnew Oulton Hall
						Hall Road Oulton NORWICH Norfolk NR11 6NU (as beneficiary of pre-emption agreement dated 28 September 2010)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-008 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights reserved and granted)
13-009	Temporary Possession	2481 square metres of arable land and hedgerow (south of Aylsham Road, B1354) (Broadland District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Saltcarr Farms Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-009 cont'd						Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU (as beneficiary of pre-emption agreement dated 28 September 2010) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) Unknown (in respect of rights reserved and granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-010	Rights D	56523 square metres of arable land and hedgerow (north of Spa Lane and south of Aylsham Road, B1354) (Broadland District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Saltcarr Farms Limited)
						Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU (as beneficiary of pre-emption agreement dated 28 September 2010)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-010 cont'd						Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas transmission pipeline)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-010 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to la and maintain pipeline and restrictive covenants) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications Unknown (in respect of rights reserved and granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-011	Temporary Possession	194 square metres of arable land (north of Spa Lane and south of Aylsham Road, B1354) (Broadland District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	NONE	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Saltcarr Farms Limited)
						Bolton Agnew Oulton Hall Hall Road Oulton NORWICH Norfolk NR11 6NU (as beneficiary of pre-emption agreement dated 28 September 2010)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-011 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) Unknown (in respect of rights reserved and granted)
13-012	Temporary Possession	6 square metres of arable land (north of Spa Lane) (Broadland District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (as reputed owner) Unknown	NONE	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-013	Rights D	694 square metres of verges and public highway (Spa Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas transmission pipeline)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regul Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-013 cont'd			Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of sub soil beneath half width of public highway) Unknown			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-014	Temporary Possession	879 square metres of arable land (Irmingland Hall Estate, south of Spa Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	NONE	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-015	Temporary Possession	58 square metres of arable land (Church Farm, south of Spa Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	NONE	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
13-016	Rights D	32735 square metres of arable land and hedgerow (Irmingland Hall Estate, south of Spa Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	NONE	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

			County	r of Norfolk Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
13-016 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to la and maintain gas pipeline and restrictive covenants)
						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	it Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-001	Rights D	35234 square metres of arable land, hedgerow and public footpath (Oulton FP4) (Church Farm, north of Spink's Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	NONE	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpaths (Oulton FP4))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
14-002	Temporary Possession	269 square metres of arable land (Church Farm, north of Spink's Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	NONE	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-003	Rights D	797 square metres of verge, hedgerow and public highway (Spink's Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-004	Temporary Possession		Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (I) p		right to use the land, and/or (III) rig y of Norfolk Category 1	hts to carry out protective works (Reg	
			Qualifying persons under Regu	ulation 7(1)(a) of the Infrastructure orms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-005	Temporary Possession	29 square metres of public highway (Spink's Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-006	Temporary Possession	895 square metres of arable land and hedgerow (Church Farm, south of Spink's Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	NONE	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
14-007	Rights D	56335 square metres of arable land, hedgerow and public footpath (Heydon FP2 and Oulton FP9) (Church Farm, south east of Spink's Lane) (Broadland District Council)	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	NONE	Ollands Investments (Farming) Ltd Barratt & Cooke 5 Opie Street Norwich NR1 3DW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	ent Consent Order nts to carry out protective works (Regu	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure F rms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-007 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpaths (Oulton FP9 and Heydon FP2))	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-001	Temporary Possession	271 square metres of verge and arable land (The Grange, B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE	William Long Farms c/o Estate Office Salle Reepham NORFOLK NR10 4SF	William Long Farms c/o Estate Office Salle Reepham NORFOLK NR10 4SF	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road Andover Hampshire SP10 2NQ (as mortgagee for Alexander Gavin Angell Lane, Patrick Edward Henley Buscall and Rhona Jane Kirwan Bulwer-Long)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-002	Temporary Possession	577 square metres of verge mad public highway (B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of sub soil beneath half width of public highway) Heydon Nominee Company 1 Limited Heydon Hall Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of sub soil beneath half width of public highway) Heydon Nominee Company 2 Limited Heydon Hall Heydon Mominee Company 2 Limited Heydon Mominee Company 2 Limited Heydon Mall Heydon Sori beneath half of RE (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-002 cont'd			Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of sub soil beneath half width of public highway) Unknown			

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-003	Temporary Possession	2295 square metres of arable land (The Grange, north of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ	William Long Farms c/o Estate Office Salle Reepham NORFOLK NR10 4SF	William Long Farms c/o Estate Office Salle Reepham NORFOLK NR10 4SF	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right to maintain and use service media and right of entry) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road Andover Hampshire SP10 2NQ (as mortgagee for Alexander Gavin Angell Lane, Patrick Edward Henley Buscall and Rhona Jane Kirwan Bulwer-Long)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-003 cont'd			Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE			
15-004	Rights D	87228 square metres of arable land and hedgerow (The Grange, north of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ	William Long Farms c/o Estate Office Salle Reepham NORFOLK NR10 4SF	William Long Farms c/o Estate Office Salle Reepham NORFOLK NR10 4SF	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right to maintain and use service media and right of entry) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-004 cont'd			Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE			The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road Andover Hampshire SP10 2NQ (as mortgagee for Alexander Gavin Angell Lane, Patrick Edward Henley Buscall and Rhona Jane Kirwan Bulwer-Long)
16-001	Rights D	957 square metres of verge and public highway (Heydon Road) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-001 cont'd			E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of sub soil beneath half width of public highway) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of electricity apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-001 cont'd			Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of sub soil beneath half width of public highway) Unknown			
16-002	Rights D	1 square metres of verge (Heydon Road) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (as reputed owner) Norman Earnest Durrant Bluestone Cottage Heydon NORWICH NR11 6RA (as reputed owner)	NONE	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH Norman Earnest Durrant Bluestone Cottage Heydon NORWICH NR11 6RA	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-002 cont'd			Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (as reputed owner) Unknown		Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE Unknown	
16-003	Rights D	23468 square metres of arable land and hedgerow (north west of Beerhouse Farm, north east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-003 cont'd						Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media) Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of rights granted by Transfer dated 7th April 2009)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 lation 7(1)(a) of the Infrastructure Pla	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Fc Owners or Reputed Owners	rms and Procedures) Regulations 20	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-003 cont'd						Docking Farm Solar Limited 361 - 373 City Road London EC1V 1LR (as beneficiary of option for lease dated 12 June 2020)
16-004	Temporary Possession	912 square metres of arable land and hedgerow (north east of Beerhouse Farm, east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media)
						Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

			County Qualifying persons under Regul Foi	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
16-004 cont'd						Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of rights granted by Transfer dated 7th April 2009)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-004 cont'd						Docking Farm Solar Limited 361 - 373 City Road London EC1V 1LR (as beneficiary of option for lease dated 12 June 2020)
16-005	Temporary Possession	838 square metres of wooded area and public highway (B1149 and Heydon Road) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of sub soil beneath half width of public highway) E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-005 cont'd			Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Rhona Jane Kirwan Bulwer- Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of sub soil beneath half width of public highway) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			Count	y of Norfolk Category 1		
				llation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-006	Temporary Possession	1390 square metres of verge and private highway (east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (as reputed owner) Unknown	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF Unknown	Unknown
16-007	Temporary Possession	158 square metres of arable land and hedgerow (north east of Beerhouse Farm, B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-007 cont'd						Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)
						Rhona Jane Kirwan Bulwer-Lon Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-007 cont'd						Docking Farm Solar Limited 361 - 373 City Road London EC1V 1LR (as beneficiary of option for lease dated 12 June 2020)
16-008	Temporary Possession	311 square metres of arable land (east of B1149) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ	NONE	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)
			M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU		M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-008 cont'd			Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ		Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Po Owners or Reputed Owners	rms and Procedures) Regulations 20 Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-008 cont'd						Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-008 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Docking Farm Solar Limited 361 - 373 City Road London EC1V 1LR (as beneficiary of option for lease dated 12 June 2020) Unknown (in respect of right reserved by deed dated 30 December 1969)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-009	Rights D	55342 square metres of arable land and hedgerow (east of B1149) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ	NONE	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)
			M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU		M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe)
			Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ		Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-009 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-009 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Docking Farm Solar Limited 361 - 373 City Road London EC1V 1LR (as beneficiary of option for lease dated 12 June 2020) Unknown (in respect of right reserved by deed dated 30 December 1969)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-010	Temporary Possession	1847 square metres of arable land and hedgerow (north east of B1149) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ	NONE	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)
			M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU		M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ		Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-010 cont'd						Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-010 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Docking Farm Solar Limited 361 - 373 City Road London EC1V 1LR (as beneficiary of option for lease dated 12 June 2020) Unknown (in respect of right reserved by deed dated 30 December 1969)

Page 395 of 1368

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-011	Rights D	890 square metres of verge and public highway (The Street) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of sub soil beneath half width of public highway) E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of sub soil beneath half width of public highway) M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning Forms and Procedures) Regulations 2009			anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-011 cont'd			Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
16-012	Rights D	18282 square metres of arable land and hedgerow (Docking Farm, east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-012 conťd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
16-013	Temporary Possession	232 square metres of verge, hedgerow and public highway (The Street) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of sub soil beneath half width of public highway) E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	l and which is propo		gham Shoal and Dudgeon Extensior BOOK OF REF owers of compulsory acquisition, (ii) i	ERENCE - PART 1		(1)
	Land which is prope			/ of Norfolk	s to carry out protective works (neg	
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-013 cont'd			M.W. Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU (in respect of sub soil beneath half width of public highway) Nicholas Ernest Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii)	FERENCE - PART 1 right to use the land, and/or (iii) rig ry of Norfolk	hts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed		
Number on Land Plans	Extent of acquisition or use	Description of land	Fo	orms and Procedures) Regulations	2009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-014	Temporary Possession	259 square metres of verge, hedge row and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited) Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-014 cont'd						Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-014 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987) Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)
16-015	Rights D	502 square metres of hedgerow and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) rig County	of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula Forr	tion 7(1)(a) of the Infrastructure Pla ns and Procedures) Regulations 20	nning (Applications: Prescribed 09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-015 cont'd						Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
				ms and Procedures) Regulations 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-015 cont'd						Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-015 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987) Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)
16-016	Temporary Possession	2337 square metres of arable land (Docking Farm, east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-016 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
16-017	Temporary Possession	258 square metres of verge, arable land and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited)
						Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
				ing persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-017 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) rig	RENCE - PART 1 ght to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				tion 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 20		Category 2
Number on Extent of acquisition or De Land Plans use De	Description of land Own	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-017 cont'd						Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) The Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenan contained in assent dated 2 October 1987)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-017 cont'd						Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)
16-018	Temporary Possession	2496 square metres of arable land and hedgerow (Docking Farm, east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited)
16-019	Temporary Possession	114 square metres of arable land (Docking Farm, east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited)
16-020	Rights D	78814 square metres of arable land and hedgerow (Docking Farm, east of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	NONE	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for E.F. Harrold Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-020 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
17-001	Rights D	2108 square metres of tree and hedge line (east of Marriott Way, B1149) (Broadland District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH Unknown (in respect of mines and minerals)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		Category 2
				llation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		outogory 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-002	Rights D	15638 square metres of arable land and hedgerow (east of B1149) (Broadland District Council)	Nicholas Gerald Youngs Horseshoe Barn Matlaske Road Mannington NORWICH Norfolk NR11 7BE	NONE	Nicholas Gerald Youngs Horseshoe Barn Matlaske Road Mannington NORWICH Norfolk NR11 7BE	NONE
					WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	
17-003	Rights D	87 square metres of hedgerow (east of B1149) (Broadland District Council)	Nicholas Gerald Youngs Horseshoe Barn Matlaske Road Mannington NORWICH Norfolk NR11 7BE	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1								
	Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1					
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-003 cont'd					WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW				
17-004	Rights D	20 square metres of hedgerow (Dixs House, west of B1149) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-005	Rights D	1726 square metres of public highway and verge (B1149) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS (in respect of sub soil beneath half width of public highway) Nicholas Gerald Youngs Horseshoe Barn Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-005 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
17-006	Temporary Possession	2065 square metres of arable land (east of B1149) (Broadland District Council)	Nicholas Gerald Youngs Horseshoe Barn Matlaske Road Mannington NORWICH Norfolk NR11 7BE	NONE	Nicholas Gerald Youngs Horseshoe Barn Matlaske Road Mannington NORWICH Norfolk NR11 7BE WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Page 414 of 1368

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-007	Rights D	19627 square metres of arable land and hedgerow (Dixs House, north of Bird's Lane and south west of B1149) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
17-008	Temporary Possession	963 square metres of arable land (Dixs House, north of Bird's Lane and south west of B1149) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE

Page 415 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-009	Rights D	560 square metres of verge, hedgerow and public highway (Birds Lane) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-010	Temporary Possession	75 square metres of arable land (Dixs House, south of Bird's Lane and south west of B1149) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE
17-011	Rights D	7668 square metres of arable land and hedgerow (Dixs House, south of Bird's Lane and south west of B1149) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE
17-012	Rights D	51650 square metres of arable land and hedgerow (west of Holt Road, B1149) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	Unknown (in respect of a right to enter the land and maintain and repair the water pipes in respect of rights reserved)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-012 cont'd						Unknown (in respect of a right to enter the land and maintain and repair the water supply)
18-001	Rights D	27689 square metres of arable land and hedgerow (north of Aylsham Road, B1145) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	Matthew Paul Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-001 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Pauline Susan Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)
18-002	Temporary Possession	252 square metres of arable land and verge (north of Aylsham Road, B1145) (Broadland District Council)	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	Matthew Paul Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-002 cont'd						Pauline Susan Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)
18-003	Rights D	813 square metres of verges and public highway (B1145) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-003 cont'd			Edward Sinclair Pye Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown
18-004	Rights D	8 square metres of verge (south of B1145) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Clifford Pye Limited)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-005	Temporary Possession	422 square metres of arable land (south of B1145) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Clifford Pye Limited)
18-006	Rights D	17343 square metres of arable land and public footpath (Cawston FP3) (south of B1145) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Cawston FP3))	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Clifford Pye Limited)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-007	Rights D	515 square metres of hedgerow and public highway (Old Friendship Lane) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-007 cont'd			The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-008	Temporary Possession	217 square metres of arable land and hedgerow (south west of Old Friendship Lane) (Broadland District Council)	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ	NONE	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-009	Rights D	20048 square metres of arable land and hedgerow(south of Old Friendship Lane) (Broadland District Council)	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ	NONE	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			Qualifying persons under Regu	County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-009 cont'd						The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)
18-010	Rights D	11543 square metres of arable land and hedgerow (north of Norwich Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	llation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-011	Rights D	1 square metres of arable land (north of Norwich Road) (Broadland District Council)	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ	NONE	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ	The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)
18-012	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-013	Rights D	450 square metres of arable land and hedgerow (north of Norwich Road) (Broadland District Council)	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ	NONE	The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road AyIsham NORWICH Norfolk NR11 6AJ	The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)
18-014	Rights D	547 square metres of arable land and public footpath (Cawton FP7 and FP8) (north of Norwich Road) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS (as reputed owner)	NONE	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		llation 7(1)(a))
				anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-014 cont'd			The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ (as reputed owner) Unknown		The Executor of the Estate of the Late David Ayton Key c/o Mr Martin Smith ClearView Financial Consultants Ltd 1 Burgh Road Aylsham NORWICH Norfolk NR11 6AJ Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Cawston FP7 and FP8))	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				y of Norfolk Category 1	ts to carry out protective works (Reg	
				llation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-015	Rights D	9454 square metres of arable land (north east of Norwich Road) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	NONE	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications
18-015 cont'd						Unknown (in respect of a right to enter the land and maintain and repair the water pipes in respect of rights reserved)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-016	Rights D	998 square metres of verge and public highway (Norwich Road) (Broadland District Council)	Clifford Pye Limited Dixs House Holt Road Cawston NORWICH Norfolk NR10 4HS (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 Iation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-016 cont'd			Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB (in respect of sub soil beneath half width of public highway) Unknown			
18-017	Rights D	14565 square metres of arable land (west of Norwich Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE
19-001	Rights D	42844 square metres of arable land and hedgerow (Church Farm, north of Reepham Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-002	Temporary Possession	3127 square metres of arable land (Church Farm, north of Reepham Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE
19-003	Rights D	507 square metres of arable land and hedgerow (north of Reepham Road) (Broadland District Council)	Tina Heybourn Hayward 2 Harrolds Grove Cottage Heydon NORWICH Norfolk NR11 6RJ	NONE	Tina Heybourn Hayward 2 Harrolds Grove Cottage Heydon NORWICH Norfolk NR11 6RJ	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB (in respect of rights of access)
19-004	Rights D	1535 square metres of Section of verge, hedgerow and public highway (Reepham Road) (Broadland District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-004 cont'd			Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB (in respect of sub soil beneath public highway) Tina Heybourn Hayward 2 Harrolds Grove Cottage Heydon NORWICH Norfolk NR11 6RJ (in respect of sub soil beneath public highway) Unknown			Unknown

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	l and which is prop		BOOK OF REF	ns Offshore Wind Farm Developmen ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	ration 7(1)(a)
	Land which is prop			y of Norfolk	s to carry out protective works (rtegt	
	Category 1					
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-005	Rights D	209 square metres of hedgerow (Reepham Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB (as reputed owner) Unknown	NONE	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
19-006	Rights D	30564 square metres of arable land and hedgerow (Church Farm, south of Reepham Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	NONE	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
19-007	Rights D	7269 square metres of arable land and hedgerow (south of Reepham Road) (Broadland District Council)	Graeme Clive Stuart Gates The Old School House Brandiston NORWICH Norfolk NR10 4PJ (as trustee of the William Gurney Charity)	Fiona Mary Heybourn Hambidge 1 Clay Lane Haveringland NORWICH Norfolk NR10 4PP	Fiona Mary Heybourn Hambidge 1 Clay Lane Haveringland NORWICH Norfolk NR10 4PP	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-007 cont'd			Guy Marcus Adams Brandiston Hall Brandiston NORWICH Norfolk NR10 4PJ (as trustee of the William Gurney Charity) Josephine Greta Adams Brandiston Hall Brandiston NORWICH Norfolk NR10 4PJ (as trustee of the William Gurney Charity)	Tina Heybourn Hayward 2 Harrolds Grove Cottage Heydon NORWICH Norfolk NR11 6RJ	Tina Heybourn Hayward 2 Harrolds Grove Cottage Heydon NORWICH Norfolk NR11 6RJ	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-007 cont'd			Malcolm Alexander Guton Hall Farm Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH (as trustee of The William Gurney Charity) Peter Thomas Day The Old Rectory The Street SWANNINGTON Norfolk NR9 5NW (as trustee of the William Gurney Charity)						

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-007 cont'd			Roger Martin Fry Hill House Cawston Road Brandiston NORWICH Norfolk NR10 4PQ (as trustee of the William Gurney Charity) Ruth Mary Fry Hill House Cawston Road Brandiston NORWICH Norfolk NR10 4PQ (as trustee of the William Gurney Charity)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-008	Temporary Possession	458 square metres of arable land (south of Reepham Road) (Broadland District Council)	Graeme Clive Stuart Gates The Old School House Brandiston NORWICH Norfolk NR10 4PJ (as trustee of the William Gurney Charity)	Fiona Mary Heybourn Hambidge 1 Clay Lane Haveringland NORWICH Norfolk NR10 4PP	Fiona Mary Heybourn Hambidge 1 Clay Lane Haveringland NORWICH Norfolk NR10 4PP	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			Guy Marcus Adams Brandiston Hall Brandiston NORWICH Norfolk NR10 4PJ (as trustee of the William Gurney Charity) Josephine Greta Adams Brandiston Hall Brandiston NORWICH Norfolk	Tina Heybourn Hayward 2 Harrolds Grove Cottage Heydon NORWICH Norfolk NR11 6RJ	Tina Heybourn Hayward 2 Harrolds Grove Cottage Heydon NORWICH Norfolk NR11 6RJ	
			NR10 4PJ (as trustee of the William Gurney Charity)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-008 cont'd			Malcolm Alexander Guton Hall Farm Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH (as trustee of The William Gurney Charity) Peter Thomas Day The Old Rectory The Street SWANNINGTON Norfolk NR9 5NW (as trustee of the William Gurney Charity)						

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-008 cont'd			Roger Martin Fry Hill House Cawston Road Brandiston NORWICH Norfolk NR10 4PQ (as trustee of the William Gurney Charity) Ruth Mary Fry Hill House Cawston Road Brandiston NORWICH Norfolk NR10 4PQ (as trustee of the William Gurney Charity)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-009	Temporary Possession	15 square metres of public highway (Unnamed Road, south of Reepham Road) (Broadland District Council)	Graeme Clive Stuart Gates The Old School House Brandiston NORWICH Norfolk NR10 4PJ (in respect of sub soil beneath public highway) Guy Marcus Adams Brandiston Hall Brandiston NORWICH Norfolk NR10 4PJ (in respect of sub soil beneath public highway) Josephine Greta Adams Brandiston NORWICH Norfolk NR10 4PJ (in respect of sub soil beneath public highway) (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
		Category 1				
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Plance rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-009 cont'd			Malcolm Alexander Guton Hall Farm Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Peter Thomas Day The Old Rectory The Street SWANNINGTON Norfolk NR9 5NW (in respect of sub soil beneath public highway)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-009 cont'd			Roger Martin Fry Hill House Cawston Road Brandiston NORWICH Norfolk NR10 4PQ (in respect of sub soil beneath public highway) Ruth Mary Fry Hill House Cawston Road Brandiston NORWICH Norfolk NR10 4PQ (in respect of sub soil beneath public highway) Unknown			

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-010	Rights D	24898 square metres of arable land and hedgerow (north of Church Lane) (Broadland District Council)	Anthea Bridget Borrett Guton Hall Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH Unknown (in respect of mines and minerals)	NONE	Anthea Bridget Borrett Guton Hall Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Anthea Bridget Borrett) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-010 cont'd						P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)	
20-001	Rights D	465 square metres of Section of verge, hedgerow and public highway (Church Lane) (Broadland District Council)	Anthea Bridget Borrett Guton Hall Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))		
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
20-001 conťď			Unknown					
20-002	Temporary Possession	266 square metres of arable land and hedgerow (south of Church Lane) (Broadland District Council)	Anthea Bridget Borrett Guton Hall Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH	NONE	Anthea Bridget Borrett Guton Hall Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Anthea Bridget Borrett)		
			Unknown (in respect of mines and minerals)			P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)		

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
20-003	Rights D	75441 square metres of arable land and hedgerow (south of Church Lane) (Broadland District Council)	Anthea Bridget Borrett Guton Hall Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH Unknown (in respect of mines and	NONE	Anthea Bridget Borrett Guton Hall Guton Hall Lane Brandiston NORWICH Norfolk NR10 4PH	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Anthea Bridget Borrett) P.D. Chapman Limited Guton Hall
			minerals)			Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)
21-001	Rights D	31318 square metres of arable land and hedgerow (west of Clay Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	Unknown (in respect of right of way restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-002	Rights D	584 square metres of verge, hedgerow and public highway (Clay Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown
21-003	Temporary Possession	394 square metres of arable land (south east of Clay Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-004	Rights D	35430 square metres of arable land and hedgerow (east of Clay Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE
21-005	Rights D	8955 square metres of wooded area (east of Clay Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	Unknown (in respect of shooting rights)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-006	Rights D	33434 square metres of wooded land, arable land, hedgerow and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP4))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
21-007	Temporary Possession	3032 square metres of arable land and tree line (north east of Church Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-007 conťď						Unknown (in respect of right of way restrictive covenants)
21-008	Temporary Possession	189 square metres of wooded land (north east of Church Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE
21-009	Temporary Possession	75 square metres of wooded land and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	Unknown (in respect of right of way restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	it Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
				Category 1		
				llation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-009 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP4))	
21-010	Temporary Possession	5780 square metres of wooded land, arable land and hedgerow and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP4))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of way stated in Conveyance dated 03 April 1965)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	llation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-011	Temporary Possession	53 square metres of wooded land (east of Church Lane) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for The Mutimer Partnership Limited)
21-012	Temporary Possession	1392 square metres of hedgerow, private road (Church Lane) and public footpath (Swannington FP6) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po		of Norfolk	nts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure F ms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-012 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP6))	George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed				Category 2
			For	ms and Procedures) Regulations 2	009	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-012 cont'd						Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of way stated in Conveyance dated 03 April 1965)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r County	/ of Norfolk	is to carry our protective works (Negr	
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-013	Rights D	43782 square metres of arable land, private road (Church Lane), hedgerow and public footpath (Swannington FP6 and Swannington FP7) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpaths (Swannington FP6 and Swannington FP7)) The Mutimer Partnership Limited	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Barclays Security Trustee Limited
					Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	1 Churchill Place LONDON E14 5HP (as mortgagee for The Mutimer Partnership Limited)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatu



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) r County Qualifying persons under Regul Foi	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-013 cont'd						Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	is Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-013 cont'd						Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo			ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	ulation 7(1)(a))
		2 ().		/ of Norfolk		
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-014	Temporary Possession	1210 square metres of arable land (north of Church Lane) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for The Mutimer Partnership Limited)
21-015	Temporary Possession	197 square metres of private road and public footpath (Swannington FP5) (Woodlands Farm, north of Church Lane) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP5)) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to The Mutimer Partnership Limited) Unknown (in respect of rights of access reserved)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po		right to use the land, and/or (iii) right y of Norfolk Category 1	s to carry out protective works (Regu	ılation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-016	Temporary Possession	669 square metres of private road and public footpath (Swannington FP5) (north east of Church Lane) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP5))	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for The Mutimer Partnership Limited)
					The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	lation 7(1)(a))
				anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-017	Temporary Possession	183 square metres of private road, farm courtyard and public footpath (Swannington FP5) (Woodlands Farm, Church Lane) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP5)) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee to The Mutimer Partnership Limited) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passage of services granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right of Norfolk Category 1	s to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-017 cont'd						Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passge of services granted) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of rights of access reserved)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				anning (Applications: Prescribed	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-018	Temporary Possession	2025 square metres of arable land, private road (Church Lane) and public footpath (Swannington FP5 and FP6) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpaths (Swannington FP5 and FP6)) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Barclays Security Trustee Limiter 1 Churchill Place LONDON E14 5HP (as mortgagee for The Mutimer Partnership Limited) Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1			
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-018 cont'd						George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to la and maintain conducting media and right of passage of services through conducting media) Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to la and maintain conducting media and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo	·	owers of compulsory acquisition, (ii)	ERENCE - PART 1		llation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-018 cont'd						Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track)
21-019	Temporary Possession	142 square metres of hedgerow, private road (Church Lane) and public footpath (Swannington FP6) (Broadland District Council)	The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP6)) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for The Mutimer Partnership Limited)

Page 467 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) rights of Norfolk Category 1	s to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-019 cont'd						Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right		ulation 7(1)(a))
	1		County	/ of Norfolk		1
			Qualifying persons under Regul For	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
21-019 cont'd						Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of access and right of passage of services through conducting media)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-001	Rights D	337 square metres of Section of watercourse (Trout Stream) (south west of Church Lane) (Broadland District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (as reputed owner) Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (as reputed owner) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN (as reputed owner)	NONE	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-001 cont'd			Unknown		Unknown	
22-002	Rights D	32927 square metres of wooded land, arable land, hedgerow and public footpath (Swannington FP8) (north of School Road) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP8))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-003	Rights D	982 square metres of of hedgerow and public highway (Upgate Road) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-004	Temporary Possession	394 square metres of arable land and public footpath (Swannington FP8) (north of Upgate Road) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpaths (Swannington FP8))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-005	Temporary Possession	19 square metres of verge, public highway (Upgate Road) and public footpath (Swannington FP8) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) (in respect of public footpath (Swannington FP8))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) p		right to use the land, and/or (iii) righ y of Norfolk Category 1	nts to carry out protective works (Reg	
			Qualifying persons under Regu Fo	ulation 7(1)(a) of the Infrastructure F prms and Procedures) Regulations :	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-006	Temporary Possession	29 square metres of public highway (Upgate Road) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order ts to carry out protective works (Regu	ulation 7(1)(a))
	Category 1					
				llation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-007	Temporary Possession	316 square metres of verge, arable land (west of Upgate Road) and public footpath (Swannington FP13) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP13))	Unknown (in respect of right of access and right of light and air and to lay water pipes)
22-008	Rights D	39526 square metres of arable land and hedgerow and public footpath (Swannington FP13) and restricted public byway (Swannington RB12) (south of Upgate Road) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	NONE	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW	Unknown (in respect of right of access and right of light and air and to lay water pipes)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order s to carry out protective works (Regi	ulation 7(1)(a))
				Category 1		
				llation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-008 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swannington FP13) and restricted byway (Swannington RB12))	
22-009	Rights D	28761 square metres of arable land and hedgerow (north of Reepham Road) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement) Eastern Power Networks PLC
						Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) rights		ulation 7(1)(a))
			-	County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed		
				rms and Procedures) Regulations 20		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-009 cont'd						Graham Martin Dacre Drayton Hall Hall Lane Drayton NORWICH Norfolk NR8 6DP (in respect of personal covenants contained in transfer dated 3 April 2018) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-009 cont'd						Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to lay and maintain service media)
22-010	Rights D	1259 square metres of verge and public highway (Reepham Road) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-010 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
22-011	Rights D	38380 square metres of arable land and hedgerow (south of Reepham Road) and restricted byway (Attlebridge RB1) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-011 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of restricted byway (Attlebridge RB1))	Graham Martin Dacre Drayton Hall Hall Lane Drayton NORWICH Norfolk NR8 6DP (in respect of personal covenants contained in transfer dated 3 April 2018) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
		Category 1				
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-011 cont'd						Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to lay and maintain service media)
23-001	Rights D	1359 square metres of public highway (Marriott's Way) (Broadland District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of heritage trail (Marriott's Way))	NONE	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of heritage trail (Marriott's Way))	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of disused railway) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-001 cont'd			Unknown		Unknown	
23-002	Rights D	327 square metres of arable land and hedgerow (south of Marriott's Lane) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 4 Mach 2020)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 4 Mach 2020)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-003	Rights D	31610 square metres of arable land and hedgerow (south of Marriott's Lane and north of Felthorpe Road) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Unknown (in respect of right of way)
23-004	Rights D	445 square metres of verge and public highway (Felthorpe Road) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-004 conťď			Unknown			
23-005	Temporary Possession	585 square metres of arable land and hedgerow (north of Felthorpe Road) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Unknown (in respect of right of way)
23-006	Temporary Possession	2 square metres of verge (Felthorpe Road) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	FERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-006 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
23-007	Rights D	22093 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
						Unknown (in respect of right of way)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-008	Temporary Possession	30747 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
						Unknown (in respect of right of way)
23-009	Temporary Possession	258 square metres of verge and public highway (Old Fakenham Road) (Broadland District Council)	Jeremy Harold Riley Church Farmhouse Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-009 cont'd			Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown
23-010	Temporary Possession	180 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Jeremy Harold Riley Church Farmhouse Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST	NONE	Jeremy Harold Riley Church Farmhouse Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST	Unknown (in respect of right to take water from pump and well and restrictive covenants)

Page 488 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-011	Temporary Possession	108 square metres of hedgerow (east of Fakenham, A1067) (Broadland District Council)	Jeremy Harold Riley Church Farmhouse Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST (in respect of sub soil beneath of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-012	Rights D	1300 square metres of verge and public highway (Old Fakenham Road) (Broadland District Council)	Jeremy Harold Riley Church Farmhouse Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST (in respect of sub soil beneath half width of public highway) Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-012 cont'd			Unknown			
23-013	Rights D	6142 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Jeremy Harold Riley Church Farmhouse Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST	NONE	Jeremy Harold Riley Church Farmhouse Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST	Unknown (in respect of right to take water from pump and well and restrictive covenants)
23-014	Rights D	3726 square metres of verge and public highway (Fakenham Road, A1067) (Broadland District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-015	Temporary Possession	113 square metres of verge and public highway (Ivy Cottages, south west of Fakenham Road, A1067) (Broadland District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications)
23-016	Temporary Possession	724 square metres of private highway and arable land (south west of Fakenham Road, A1067) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Keith Robert Miller 1 Ivy Cottages Fakenham Road ATTLEBRIDGE Norfolk NR9 5TG (in respect of right of way)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forma and Procedures) Regulations 2009
23-016 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication Simon James Cram 2 Ivy Cottages Fakenham Road ATTLEBRIDGE Norfolk NR9 5TG (in respect of right of way) Unknown (in respect of right of way)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	llation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
	Extent of acquisition or use	· Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-017	Rights D	20334 square metres of arable land and hedgerow (south west of Fakenham Road, A1067) (Broadland District Council)	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	NONE	Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Unknown (in respect of right of way)
23-018	Rights D	622 square metres of watercourse (River Wensum) (south west of Fakenham Road, A1067) (Broadland District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (as reputed owner) Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA (as reputed owner)	NONE	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU Julia May Dacre Ashtree Farm Station Road Attlebridge NORWICH Norfolk NR9 5AA	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-018 cont'd			Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (as reputed owner) Unknown		Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	
23-019	Rights D	914 square metres of hedgerow (south of Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	NONE	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-019 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)
23-020	Rights D	9158 square metres of arable land (south of Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (as reputed lessee)	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)
				William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed		
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	rms and Procedures) Regulations 2 Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-021	Rights D	4489 square metres of woodland, drain and land (south of Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	NONE	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)
23-022	Temporary Possession	989 square metres of arable land and private road (south of Fakenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (as reputed owner)	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ ty of Norfolk		ulation 7(1)(a))	
				Category 1		Category 2	
				Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Exte Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-022 cont'd				William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)	



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
Number on Land Plans Extent of acquisition or use Description of land			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
23-022 cont'd						Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access)
23-023	Temporary Possession	946 square metres of private road (south of Fakenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	NONE	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-023 cont'd						Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (in respect of right of access) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019) Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
Number on Extent of acquisition Land Plans use			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
	Extent of acquisition or use	^{n or} Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-023 cont'd						William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ (in respect of right of access)
23-024	Temporary Possession	11 square metres of land and private road (south of Fakenham Road, A1067) (Broadland District Council)	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP	NONE	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Extent of acquisition or Descrip Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-025	Temporary Possession	1 square metres of private road (south of Fakenham Road, A1067) (Broadland District Council)	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP	NONE	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP	NONE
23-026	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE
23-027	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				ty of Norfolk Category 1	ghts to carry out protective works (Reg	
					Planning (Applications: Prescribed ; 2009	Category 2
Number on Extent of acquisition or Land Plans use		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-028	Temporary Possession	165 square metres of public highway (Fackenham Road, A1067) (Broadland District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary in respect of an Option of Lease dated 5 May 2021) Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (in respect of personal covenants relating to erection of boundary fence) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option of the lease dated 5th May 2021)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	FERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Deg	Category 1	Dianning (Applications, Dressrihad	Category 2
			Qualitying persons under Regu	Ilation 7(1)(a) of the Infrastructure F orms and Procedures) Regulations	2009	
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-028 cont'd						Unknown (in respect of rights reserved) Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications)
23-029	Rights D	862 square metres of woodland (west of Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	NONE	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-030	Rights D	56 square metres of woodland (west of Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	NONE	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)
23-031	Rights D	85136 square metres of arable land, woodland and hedgerow (west of Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (as reputed owner)	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	-	BOOK OF RE wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right ty of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-031 cont'd				William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-001	Temporary Possession	827 square metres of arable land (east of Ringland Lane) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (as reputed owner)	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)
				William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)
24-002	Temporary Possession	19 square metres of access track and verge (to the east of Ringland Lane) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (as reputed owner)	NONE	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Unknown

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-002 cont'd			Unknown			
24-003	Temporary Possession	106 square metres of verge and public highway (Ringland Lane) (Broadland District Council)	Elizabeth Ann Earl Heathgreen House Heath Green Medstead ALTON Hampshire GU34 5NG (in respect of sub soil beneath half width of public highway) Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-003 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
24-004	Rights D	3945 square metres of woodland and verge (east of Ringland Lane) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	NONE	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 26 September 2019)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-005	Rights D	821 square metres of public highway (Ringland Lane) (Broadland District Council)	Elizabeth Ann Earl Heathgreen House Heath Green Medstead ALTON Hampshire GU34 5NG (in respect of sub soil beneath half width of public highway) Margaret Ann Prince-Smith Morton Hall Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-005 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
24-006	Temporary Possession	608 square metres of arable land and private access road (west of Ringland Lane) (Broadland District Council)	Elizabeth Ann Earl Heathgreen House Heath Green Medstead ALTON Hampshire GU34 5NG	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (as reputed owner)	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 18 February 2020)
			Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 18 February 2020)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-006 cont'd						Unknown (in respect of rights reserved by conveyance dated 6 January 1983)
24-007	Rights D	103393 square metres of hedgerow and arable land (north east of Church Hill Lane) (Broadland District Council)	Elizabeth Ann Earl Heathgreen House Heath Green Medstead ALTON Hampshire GU34 5NG Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (as reputed owner) William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk	Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 18 February 2020) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osea to be subject to: (1) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) rights of Norfolk Category 1	s to carry out protective works (Reg	uiauon /(1)(a))
				ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-007 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 18 February 2020) Unknown (in respect of rights reserved by conveyance dated 6 January 1983)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure PI orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-001	Rights D	891 square metres of ditch and thicket (east of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (as reputed owner)	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Elizabeth Ann Earl Heathgreen House Heath Green Medstead ALTON Hampshire GU34 5NG (as reputed owner)		Elizabeth Ann Earl Heathgreen House Heath Green Medstead ALTON Hampshire GU34 5NG	Unknown
			Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (as reputed owner)		Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-001 cont'd			Unknown		Unknown	
25-002	Temporary Possession	4 square metres of private road (east of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (as reputed owner) Unknown	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ Unknown	Unknown
25-003	Temporary Possession	85 square metres of verge and public highway (Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-003 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown
25-004	Temporary Possession	16 square metres of verge (east of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (as reputed owner) Unknown	NONE	Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-005	Temporary Possession	1524 square metres of arable land (east of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-005 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ebony Holding Limited) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to la and maintain gas main and restrictive covenants) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 9 October 2019



Doc. No. C282-DM-Z-GA-00003

		County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
Number on Land Plans	Description of land	Forr Owners or Reputed Owners	ns and Procedures) Regulations 20 Lessees or Tenants	09 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-005 cont'd					Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to la and maintain gas main and restrictive covenants) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 9 October 2019



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-005 conťď						Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)
25-006	Rights D	61433 square metres of hedgerow, arable land and verge (east of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

				ERENCE - PART 1		
	Land which is propo	osed to be subject to: (I) po	owers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-006 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ebony Holdings Limited) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 9 October 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-006 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 9 October 2019) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-007	Rights D	57 square metres of verge (south of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (as reputed owner) Unknown	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ Unknown	Unknown
25-008	Rights D	539 square metres of verge and public highway (Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-008 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown
25-009	Temporary Possession	1111 square metres of private road and arable land (north of Weston Road) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-009 cont'd						Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ebony Holdings Limited) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) rig	RENCE - PART 1 ght to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula Forr	tion 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-009 cont'd						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 9 October 2019) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas transmission pipeline)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-009 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 9 October 2019) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)
25-010	Temporary Possession	57 square metres of verge and public highway (Weston Road) (Broadland District Council)	David Robert Pett 2 Breck Barn Cottage Ringland NORWICH Norfolk NR8 6JL (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of right of access)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-010 cont'd			Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway) Michelle Nicola Pett (nee) McConkey 2 Breck Barn Cottage Ringland NORWICH Norfolk NR8 6JL (in respect of sub soil beneath half width of public highway)		Unknown	Unknown

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-010 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
25-011	Temporary Possession	16 square metres of public highway (Church Hill Lane), verge and public footpath (Weston Longville FP9) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) (in respect of public footpath (Weston Longville FP9))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1		gulation 7(1)(a))
			Qualifying persons under Regula For	Category 1 ation 7(1)(a) of the Infrastructure F ms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-011 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Richard Alexander Millar Flint Cottage 12 Top Road Belaugh NORWICH Norfolk NR12 8XB (in respect of sub soil beneath half width of public highway) SLT Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU (in respect of sub soil beneath half width of public highway)		Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-011 cont'd			Terence Davies Manor Farm 45 The Street Ringland NORWICH Norfolk NR8 6JH (in respect of sub soil beneath half width of public highway) Unknown			
25-012	Temporary Possession	54 square metres of public highway (Church Hill Lane), verge and public footpath (Weston Longville FP9) (Broadland District Council)	Richard Alexander Millar Flint Cottage 12 Top Road Belaugh NORWICH Norfolk NR12 8XB	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) (in respect of public footpath (Weston Longville FP9))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-012 cont'd			SLT Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU Terence Davies Manor Farm 45 The Street Ringland NORWICH Norfolk NR8 6JH			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
25-013	Temporary Possession	5 square metres of verge and public footpath (Weston Longville FP9) (South of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (as reputed owner)	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order nts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 ation 7(1)(a) of the Infrastructure F rms and Procedures) Regulations 3		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-013 cont'd			Richard Alexander Millar Flint Cottage 12 Top Road Belaugh NORWICH Norfolk NR12 8XB (as reputed owner) SLT Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU (as reputed owner) Terence Davies Manor Farm 45 The Street Ringland NORWICH Norfolk NR8 6JH (as reputed owner)		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weston Longville FP9)) Richard Alexander Millar Flint Cottage 12 Top Road Belaugh NORWICH Norfolk NR12 8XB SLT Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-013 cont'd			Unknown		Terence Davies Manor Farm 45 The Street Ringland NORWICH Norfolk NR8 6JH Unknown	
25-014	Temporary Possession	261 square metres of public footpath (Weston Longville FP9), land and verge (south of Church Hill Lane) (Broadland District Council)	Richard Alexander Millar Flint Cottage 12 Top Road Belaugh NORWICH Norfolk NR12 8XB	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weston Longville FP9))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-014 cont'd			SLT Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Terence Davies Manor Farm 45 The Street Ringland NORWICH Norfolk NR8 6JH			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
25-015	Temporary Possession	219 square metres of verge and public footpath (Weston Longville FP9) (south of Church Hill Lane) (Broadland District Council)	David Robert Pett 2 Breck Barn Cottage Ringland NORWICH Norfolk NR8 6JL (as reputed owner)	NONE	David Robert Pett 2 Breck Barn Cottage Ringland NORWICH Norfolk NR8 6JL	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-015 cont'd			Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (as reputed owner)		Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Michelle Nicola Pett (nee) McConkey 2 Breck Barn Cottage Ringland NORWICH Norfolk NR8 6JL (as reputed owner)		Michelle Nicola Pett (nee) McConkey 2 Breck Barn Cottage Ringland NORWICH Norfolk NR8 6JL	Unknown
			Richard Alexander Millar Flint Cottage 12 Top Road Belaugh NORWICH Norfolk NR12 8XB (as reputed owner)		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Weston Longville FP9))	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-015 cont'd			SLT Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU (as reputed owner) Terence Davies Manor Farm 45 The Street Ringland NORWICH Norfolk NR8 6JH (as reputed owner) Unknown		Richard Alexander Millar Flint Cottage 12 Top Road Belaugh NORWICH Norfolk NR12 8XB SLT Trustees Limited 1 New Walk Place LEICESTER Leicestershire LE1 6RU Terence Davies Manor Farm 45 The Street Ringland NORWICH Norfolk NR8 6JH	
					Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-016	Temporary Possession	1101 square metres of hedgerow and arable land (south of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (I) po	wers of compulsory acquisition, (ii) ri County	of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-016 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ebony Holdings Limited) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) rights of Norfolk Category 1	s to carry out protective works (Reg	
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-016 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lat and maintain gas main and restrictive covenants) Unknown (in respect of rights of service media and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii)	FERENCE - PART 1 right to use the land, and/or (iii) righ ty of Norfolk Category 1	ts to carry out protective works (Reg	ulation 7(1)(a))
				ulation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-017	Rights D	47124 square metres of arable land (south of Church Hill Lane) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-017 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ebony Holdings Limited) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul Foi	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-017 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights of service media and restrictive covenants)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

			Count	y of Norfolk Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-001	Rights D	208 square metres of public highway (The Broadway) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ebony Holding Limited)
						Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to la and maintain gas main and restrictive covenants)
						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to la and maintain gas main and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-001 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)
26-002	Rights D	654 square metres of woodland, verge and public highway (The Broadway) (Broadland District Council)	Ebony Holdings Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
26-002 cont'd			Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown						

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
				rms and Procedures) Regulations 20		, ,
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-003	Temporary Possession	83 square metres of verge and public highway (The Broadway) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sub soil beneath half width of public highway) Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-003 conťd			Unknown			
26-004	Rights D	13 square metres of woodland and verge (The Broadway) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ (as reputed owner) Unknown	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ Unknown	Unknown
26-005	Temporary Possession	66 square metres of woodland and verge (The Broadway) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ (as reputed owner)	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)
			Unknown		Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-006	Temporary Possession	380 square metres of hedgerow and woodland (The Broadway) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-006 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of right of way and shooting rights)
26-007	Rights D	111007 square metres of hedgerow, arable land, woodland, private road, verge and public highway (The Broadway) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-007 cont'd						Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regul Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
26-007 cont'd						Unknown (in respect of right of way granted and shooting rights)			
26-008	Temporary Possession	937 square metres of arable land and verge (west of Taverham Road) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Matthew John Rampton Kingfisher House 1 Gilders Way NORWICH Norfolk NR3 1UB (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right of way and shooting rights)			
27-001	Temporary Possession	70 square metres of public highway (Taverham Road) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF wers of compulsory acquisition, (ii) i		t Consent Order s to carry out protective works (Regu	ulation 7(1)(a))
			Qualifying persons under Regu	County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-001 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown
27-002	Temporary Possession	1835 square metres of arable land and private road (east of Taverham Road) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Unknown (in respect of right of access and right of passage of services through conducting media)
27-003	Rights D	806 square metres of hedgerow and public highway (Taverham Road) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-003 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			Unknown
27-004	Rights D	64154 square metres of arable land, hedgerow and private road (east of Taverham Road) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-004 cont'd						Unknown (in respect of right of access and right of passage of services through conducting media)
27-005	Rights D	36546 square metres of watercourse (River Tud), arable land and hedgerow (north of the A47) (Broadland District Council)	Anne Denise Gowing Brook House Honingham NORWICH NR9 5BT (as reputed owner, trustees of the A47 Partnership) Heather Jane Harrison Vale Farm 22 Bridge Street Stiffkey WELLS-NEXT-THE-SEA NR23 1QP (as reputed owner, trustees of the A47 Partnership)	Joanne Kate Gowing Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ Nicholas Charles Gowing Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ	E B Gowing & Son Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-005 cont'd			Nicholas Charles Gowing Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ (as reputed owner, trustees of the A47 Partnership) Unknown			
27-006	Temporary Possession	2245 square metres of arable land and hedgerow (north of A47) (Broadland District Council)	Anne Denise Gowing Brook House Honingham NORWICH NR9 5BT (as reputed owner, trustees of the A47 Partnership)	Joanne Kate Gowing Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ	E B Gowing & Son Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of right of access to lay and maintain apparatus)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-006 cont'd			Heather Jane Harrison Vale Farm 22 Bridge Street Stiffkey WELLS-NEXT-THE-SEA NR23 1QP (as reputed owner, trustees of the A47 Partnership) Nicholas Charles Gowing Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ (as reputed owner, trustees of the A47 Partnership) Unknown	Nicholas Charles Gowing Kenningham Hall Brick Kiln Road Mulbarton NORWICH Norfolk NR14 8AJ (as trustees of the A47 Partnership)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-001	Rights D	20155 square metres of arable land and hedgerow (north of the A47) (Broadland District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of access and right of passage of services through conducting media)
28-002	Rights D	845 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-003	Temporary Possession	901 square metres of arable land and access track (north of the A47) (South Norfolk District Council)	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	NONE	Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of access and right of passage of services through conducting media)
28-004	Temporary Possession	121 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-004 cont'd						The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)
28-005	Temporary Possession	3 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown (in respect of restrictive covenants and rentcharges as may have been imposed before 12 February 2009)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-006	Temporary Possession	298 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway)	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Honingham Aktieselskab Easton Estates Locksash Farm House Chichester PO18 9DZ (in respect of sub soil beneath half width of public highway)		Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority)			The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-006 cont'd			Unknown			Unknown
28-007	Temporary Possession	14 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of rights granted)	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of rights granted)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)
28-008	Rights D	589 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-008 cont'd						The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown (in respect of restrictive covenants and rentcharges as may have been imposed before 12 February 2009)
28-009	Rights D	1231 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway)	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
28-009 cont'd			National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (as highway authority) Unknown		Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-010	Rights D	649 square metres of public highway, lay by and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)
28-011	Rights D	49638 square metres of arable land, woodland, hedgerow and verges (south of the A47 and north of Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) right	ERENCE - PART 1 ght to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula Fori	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-011 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Food Enterprise Park Limited) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary to an Option Agreement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-011 cont'd						Unknown (in respect of right of access and right of passage of services through conducting media)
28-012	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE
28-013	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii)	y of Norfolk	nts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-014	Rights D	243 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Food Enterpris Park Limited) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain and in respect of right to maintain woodland path) Unknown (in respect of right of access and right of passage of services through conducting media)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-015	Rights D	403 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-015 cont'd			Unknown			
28-016	Rights D	203 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Norwich City College of Further and Higher Education)
28-017	Temporary Possession	1458 square metres of arable land, track and hedgerow (south of Church Road) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 27th August 2020) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Page 570 of 1368



Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2(anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-017 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Norwich City College of Further and Higher Education) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 27th August 2020)
28-018	Temporary Possession	1075 square metres of arable land and hedgerow (south of Church Lane) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 27th August 2020)

Page 571 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-018 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Norwich City College of Further and Higher Education) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 27th August 2020)
28-019	Rights D	17488 square metres of arable land and hedgerow (south of Church Lane) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norwich City College of Further and Higher Education c/o Martin Colbourne lpswich Road Norwich NR2 2LJ	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 27th August 2020)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-019 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Norwich City College of Further and Higher Education) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option agreement dated 27th August 2020)
28-020	Temporary Possession	1598 square metres of arable land and hedgerow (south of Church Road) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Norwich City College of Further and Higher Education)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order s to carry out protective works (Regu	ulation 7(1)(a))
			Qualifving persons under Regu	Category 1 lation 7(1)(a) of the Infrastructure Pla	anning (Applications: Prescribed	Category 2
				rms and Procedures) Regulations 20		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-021	Rights D	358 square metres of public highway and verges (Broom Lane) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of sub soil beneath public highway) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown
28-022	Rights D	31129 square metres of arable land, grassland and hedgerow (south of Broom Lane) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Norwich City College of Further and Higher Education)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-023	Temporary Possession	1 square metre of arable land and access track (north of Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton Norwich NR9 5BZ	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton Norwich NR9 5BZ	Barclays Security Trustee Limited 1 Churchill Place London E14 5HP (as mortgagee for Food Enterprise Park Limited) Broadland District Council Thorpe Lodge 1 Yarmouth Road Norwich NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021)
28-023 cont'd						Norfolk County Council County Hall Martineau Lane Norwich NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
						Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016)
28-023 cont'd						Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect rights granted) Condimentum Limited Food Enterprise Park Colmans Close Easton Noriwch NR9 5FG (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-023 cont'd						Eurotek Engineering Limited Centrum Noriwch Research Park Colney Lane Colney Norwich NR4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Property Vision (Norwich) Limited



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu	lation 7(1)(a) of the Infrastructure Pla orms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-023 cont'd						c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)
28-024	Temporary Possession	2 square metres of arable land and access track (north of Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton Norwich NR9 5BZ	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton Norwich NR9 5BZ	Barclays Security Trustee Limited 1 Churchill Place London E14 5HP (as mortgagee for Food Enterprise Park Limited)
						Broadland District Council Thorpe Lodge 1 Yarmouth Road Norwich NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021)



Doc. No. C282-DM-Z-GA-00003

			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-024 cont'd						Norfolk County Council County Hall Martineau Lane Norwich NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		I
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-024 cont'd						Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016)



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) ri County	of Norfolk Category 1		
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure PI ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-024 cont'd						Broadland Food Innovation Cent Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)
						Condimentum Limited Food Enterprise Park Colmans Close Easton Noriwch NR9 5FG (in respect of rights granted)
						Eurotek Engineering Limited Centrum Noriwch Research Park Colney Lane Colney Norwich NR4 7UG (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 lation 7(1)(a) of the Infrastructure P	lanning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	orms and Procedures) Regulations 2	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-024 cont'd						Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)
28-025	Temporary Possession	123 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton Norwich NR9 5BZ (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown



Doc. No. C282-DM-Z-GA-00003

		The Sherin	gham Shoal and Dudgeon Extension		t Consent Order	
	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) rights / of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
			Norwich City College of Further and Higher Education c/o Martin Colbourne lpswich Road Norwich NR2 2LJ (in respect of sub soil beneath half width of public highway)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))	
			Qualifying persons under Regu	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-025 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown				
28-026	Temporary Possession	6041 square metres of arable land and private road (Coleman Road) (north of Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Food Enterprise Park Limited) Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021)	

Page 584 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-026 cont'd						Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)
						Unknown (in respect of right of access and right of passage of services through conducting media)
28-026 cont'd						Condimentum Limited Food Enterprise Park Colmans Close Easton NORWICH NR9 5FG (in respect of rights granted)
						Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
						NR9 5FX (in respect of rights granted) Eurotek Engineering Limited Centrum Norwich Research Park Colney Lane Colney Norwich NE4 7UG (in respect of rights granted)
28-026 cont'd						Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-027	Temporary Possession	483 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Food Enterprise Park Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Food Enterprise Park Limited) Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021)



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) r County	v of Norfolk Category 1	a to carry out protective works (iteg	
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-027 cont'd						Condimentum Limited Food Enterprise Park Colmans Close Easton NORWICH NR9 5FG (in respect of rights granted) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
			Qualifying persons under Regula Form	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-027 cont'd						Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regula For	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-027 cont'd						Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Eurotek Engineering Limited Centrum Norwich Research Park Colney Lane Colney Norwich NE4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	it Consent Order is to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-027 cont'd						Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)
29-001	Temporary Possession	2802 square metres of access track and verges (east of Colton Road) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	Annette Lucie Alston Honingham Thorpe Farm Norwich Road Colton NORWICH NR9 5BZ (in respect of right of access) Ian James Alston Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)



Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-001 cont'd						William Young (Dereham) Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)
29-002	Temporary Possession	95 square metres of public highway and verges (Colton Road) (South Norfolk District Council)	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Page 592 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-002 cont'd			Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of sub soil beneath half width of public highway) Unknown William Young (Dereham) Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of sub soil beneath half width of public highway)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	ed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regu County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			
				rms and Procedures) Regulations 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-003	Rights D	60617 square metres of arable land, woodland, hedgerow and track (east of Colton Road) (South Norfolk District Council)	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	NONE	Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ	Annette Lucie Alston Honingham Thorpe Farm Norwich Road Colton NORWICH NR9 5BZ (in respect of right of access) lan James Alston Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access) William Young (Dereham) Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-004	Rights D	8190 square metres of arable land and hedgerows (east of Colton Road) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe) Unknown (in respect of rights reserved)
29-005	Rights D	685 square metres of public highway and verges (Colton Road) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	nning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-005 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of sub soil beneath half width of public highway)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-005 cont'd			Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of sub soil beneath half width of public highway) Unknown			
29-006	Rights D	32592 square metres of arable land, grassland and hedgerows (west of Colton Road) (South Norfolk District Council)	Norwich Diocesan Board Of Finance Limited Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

			BOOK OF REF	ns Offshore Wind Farm Developmen ERENCE - PART 1		
	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii) Count	right to use the land, and/or (iii) right y of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-006 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
29-007	Rights D	3726 square metres of arable land and hedgerows (west of Colton Road) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe) Unknown (in respect of rights reserved)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-008	Rights D	43135 square metres of arable land, grassland, woodland and hedgerows (west of Chapel Street) (South Norfolk District Council)	Colin Herbert Bothway Cherry Tree Farm Salle NORWICH Norfolk NR10 4SF (in respect of mines and minerals) Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Honingham Farms Limited in respect of floating charge) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Honingham Farms Limited)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-008 cont'd						Colin Herbert Bothway Cherry Tree Farm Salle NORWICH Norfolk NR10 4SF (in respect of right of entry to extract mines and minerals, timber rights, positive covenants and restrictive covenants) Unknown (in respect of light and air)
30-001	Rights D	608 square metres of watercourse (River Yare) (South Norfolk District Council)	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (as reputed owner)	NONE	Honingham Farms Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-001 cont'd			Janine Elizabeth Curtis The Ruffetts Honingham Road Barnham Broom NORWICH Norfolk NR9 4DB (as reputed owner) Unknown		Janine Elizabeth Curtis The Ruffetts Honingham Road Barnham Broom NORWICH Norfolk NR9 4DB Unknown	
30-002	Rights D	23868 square metres of arable land, hedgerow, drain and private road (south of the River Yare) (South Norfolk District Council)	Janine Elizabeth Curtis The Ruffetts Honingham Road Barnham Broom NORWICH Norfolk NR9 4DB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-003	Rights D	20021 square metres of arable land, hedgerows and verge (north of Chapel Street) (South Norfolk District Council)	Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)
30-004	Temporary Possession	120 square metres of private road (north of Chapel Street) (South Norfolk District Council)	Janine Elizabeth Curtis The Ruffetts Honingham Road Barnham Broom NORWICH Norfolk NR9 4DB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)
30-005	Temporary Possession	235 square metres of arable land (north of Chapel Street) (South Norfolk District Council)	Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	NONE

Page 602 of 1368

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-006	Temporary Possession	484 square metres of land and buildings (north of Chapel Street) (South Norfolk District Council)	Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	E C Leaper St. Michaels Honingham Road Barnham Broom NORWICH NR9 4DB	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)
30-007	Temporary Possession	7 square metres of private access road and verge (north of Chapel Street) (South Norfolk District Council)	Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB (as reputed owner) Unknown	NONE	Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-008	Temporary Possession	25 square metres of public highway (Chapel Street) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB (in respect of sub soil beneath half width of public highway) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-009	Rights D	18 square metres of verge (north of Chapel Street) (South Norfolk District Council)	Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB (as reputed owner) Unknown	NONE	Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB Unknown	Unknown
30-010	Rights D	959 square metres of public highway (Chapel Street) (South Norfolk District Council)	Graham Gamble Common Farm Chapel Street Barford NORWICH NR9 4BJ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-010 cont'd			Rodney Neville Lee Brown 46 Chapel Street Barford NORWICH Norfolk NR9 4AB (in respect of sub soil beneath half width of public highway) Unknown			Unknown
30-011	Rights D	35447 square metres of arable land, hedgerows and tree line (south of Chapel Street) (South Norfolk District Council)	Graham Gamble Common Farm Chapel Street Barford NORWICH NR9 4BJ	NONE	Graham Gamble Common Farm Chapel Street Barford NORWICH NR9 4BJ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)
			Unknown		Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-011 cont'd						Unknown
30-012	Rights D	532 square metres of arable land and hedgerow (south of Chapel Street) (South Norfolk District Council)	Janet Elizabeth Green Waterways Farm Church Road East Harling NORWICH NR16 2NB (as reputed owner) Robert George Green Waterways Farm Church Road East Harling NORWICH NR16 2NB (as reputed owner) Unknown	NONE	Janet Elizabeth Green Waterways Farm Church Road East Harling NORWICH NR16 2NB Robert George Green Waterways Farm Church Road East Harling NORWICH NR16 2NB Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order its to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-013	Rights D	1227 square metres of watercourse (River Tiffey) (north of Watton Road, B1108) (South Norfolk District Council)	Gi-Fay Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ (as reputed owner) Sophie Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ (as reputed owner) Graham Gamble Common Farm Chapel Street Barford NORWICH NR9 4BJ (as reputed owner)	NONE	Gi-Fay Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ Sophie Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ Graham Gamble Common Farm Chapel Street Barford NORWICH NR9 4BJ	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul	Category 1 ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-013 cont'd			Janet Elizabeth Green Waterways Farm Church Road East Harling NORWICH NR16 2NB (as reputed owner)		Janet Elizabeth Green Waterways Farm Church Road East Harling NORWICH NR16 2NB	
			Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (as reputed owner)		Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	
			Robert George Green Waterways Farm Church Road East Harling NORWICH NR16 2NB (as reputed owner)		Robert George Green Waterways Farm Church Road East Harling NORWICH NR16 2NB	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-013 conťď			Unknown		Unknown	
30-014	Rights D	9178 square metres of woodland (south of River Tiffey) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe) Unknown (in respect of rights reserved)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-015	Rights D	174 square metres of woodland (south of River Tiffey) (South Norfolk District Council)	Gi-Fay Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ (as reputed owner) Sophie Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ (as reputed owner) Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (as reputed owner)	NONE	Gi-Fay Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ Sophie Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Unknown

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	is Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right / of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-015 conťd			Unknown		Unknown	
30-016	Rights D	693 square metres of woodland (south of the River Tiffey) (South Norfolk District Council)	Gi-Fay Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ (as reputed owner) Sophie Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ (as reputed owner)	NONE	Gi-Fay Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ Sophie Mok Turnpike Cottage South Norwich Road Barford Norwich NR9 4BQ	Helga Erika Jalil 1A Gilbert Way Cringleford NORWICH Norfolk NR4 7RN (in respect of a right of access, right of passage and running of surface water and provision of light and air)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-017	Rights D	11929 square metres of hedgerow and arable land (north of Watton Road B1108) (South Norfolk District Council)	Helga Erika Jalil 1A Gilbert Way Cringleford NORWICH Norfolk NR4 7RN (as the executor of the estate of the late Syed Maswood Jalil)	NONE	Grahame Green Edgefield Longwater Lane NORWICH NR5 0TD	NONE
31-001	Rights D	1209 square metres of verge and public highway (Watton Road, B1108) (South Norfolk District Council)	Helga Erika Jalil 1A Gilbert Way Cringleford NORWICH Norfolk NR4 7RN (in respect of sub soil beneath half width of public highway) Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
	Category 1					
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-001 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
31-002	Rights D	17700 square metres of immature trees and hedgerow (south of Watton Road, B1108) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe)
31-003	Temporary Possession	702 square metres of private road (south of Watton Road, B1108) (South Norfolk District Council)	Punter Southall Sipp Trustees 11 Strand LONDON WC2N 5HR	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-003 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ			
31-004	Rights D	25870 square metres of private road, arable land and hedgerow (north of Burdock Lane) (South Norfolk District Council)	Punter Southall Sipp Trustees 11 Strand LONDON WC2N 5HR	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			Count Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-004 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ			Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)
31-005	Rights D	679 square metres of woodland (north of Burdock Lane) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe) Unknown (in respect of rights reserved)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
31-006	Temporary Possession	2467 square metres of private road (north of Burdock Lane) (South Norfolk District Council)	Punter Southall Sipp Trustees 11 Strand LONDON WC2N 5HR The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-007	Rights D	600 square metres of verge and public highway (Burdock Lane) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Punter Southall Sipp Trustees 11 Strand LONDON WC2N 5HR (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-007 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Unknown			
31-008	Temporary Possession	33 square metres of verge (south of Burdock Lane) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-008 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
31-009	Temporary Possession	146 square metres of hedgerow (south of Burdock Lane) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Plance rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-010	Temporary Possession	14 square metres of public highway (Burdock Lane) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (as reputed owner) Punter Southall Sipp Trustees 11 Strand LONDON WC2N 5HR (as reputed owner) The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner)	NONE	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP Punter Southall Sipp Trustees 11 Strand LONDON WC2N 5HR The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Unknown
			Unknown		Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-011	Rights D	35462 square metres of arable land and hedgerow (south of Burdock Lane) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe)
31-012	Rights D	31953 square metres of arable land, drain, hedgerow and public footpath (Great Melton FP1) (west of Pockthorpe Road) (South Norfolk District Council)	Punter Southall Sipp Trustees 11 Strand LONDON WC2N 5HR	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to use, lay and maintain electricity cables)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				ERENCE - PART 1		
	Land which is propo	osed to be subject to: (I) po	owers of compulsory acquisition, (ii) Count	right to use the land, and/or (iii) right y of Norfolk	s to carry out protective works (Regi	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
31-012 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Great Melton FP1))	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)
						Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)
32-001	Rights D	26466 square metres of hedgerow, drain and arable land (north of Skoyles Lane) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-001 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nicholas Edward Evans-Lombe)
32-002	Rights D	13733 square metres of hedgerow and arable land (north of Skoyles Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted)
			Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE		Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of right of way over access track)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-002 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ		The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
32-003	Rights D	520 square metres of verge and public highway (Skoyles Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (In respect to water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-003 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Sarah Frances Buxton Algarsthorpe Farm Marlingford NORWICH NR9 5JA (in respect of sub soil beneath half width of public highway) The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
	Category 1					
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-003 cont'd			Unknown			
32-004	Temporary Possession	110 square metres of verge and public highway (Skoyles Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-004 cont'd			Sarah Frances Buxton Algarsthorpe Farm Marlingford NORWICH NR9 5JA (in respect of sub soil beneath half width of public highway) The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF RE owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	llation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-005	Temporary Possession	390 square metres of arable land and track (south of Skoyles Lane) (South Norfolk District Council)	Sarah Frances Buxton Algarsthorpe Farm Marlingford NORWICH NR9 5JA	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Susanne Eva Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted) Thomas Charles Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted) Norfolk NR18 0SA (in respect of rights granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	owers of compulsory acquisition, (ii)	FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				llation 7(1)(a) of the Infrastructure F orms and Procedures) Regulations		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-006	Rights D	28012 square metres of arable land, ditch and hedgerows (south of Skoyles Lane) (South Norfolk District Council)	Sarah Frances Buxton Algarsthorpe Farm Marlingford NORWICH NR9 5JA	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (In respect to water apparatus) Susanne Eva Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
	County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-006 cont'd						Thomas Charles Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)
33-001	Rights D	15856 square metres of arable land, watercourse and hedgerow (north west of Melton Road) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE	David Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-001 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ			
33-002	Temporary Possession	29 square metres of public highway (Pockthorpe Road) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-002 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-003	Temporary Possession	4471 square metres of verge and hedgerow (west of Melton Road) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	David Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	NONE
33-003 cont'd			Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE	Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX		
			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-004	Temporary Possession	11 square metres of public highway (Melton Road) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	l and which is propo		gham Shoal and Dudgeon Extensior BOOK OF REF owers of compulsory acquisition, (ii) i	ERENCE - PART 1		(1)
	Land which is prope			y of Norfolk	s to carry out protective works (rieg	
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-004 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Unknown			
33-005	Temporary Possession	594 square metres of private road (west of Melton Road) (South Norfolk District Council)	Dawn Marie Sheldrake White Cottage Melton Road WYMONDHAM Norfolk NR18 0SE (as reputed owner)	NONE	Dawn Marie Sheldrake White Cottage Melton Road WYMONDHAM Norfolk NR18 0SE	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-005 cont'd			Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner) Great Melton Parish Council Matsall House Church Lane Mattishall DEREHAM NR20 3QZ (as reputed owner) Henrietta Katherine Norman 90 Christchurch Avenue		Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ Great Melton Parish Council Matsall House Church Lane Mattishall DEREHAM NR20 3QZ Henrietta Katherine Norman 90 Christchurch Avenue	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-005 cont'd			James Christopher Martin Starling Town Cottage Melton Road Great Melton NORWICH Norfolk NR9 3BT (as reputed owner)		James Christopher Martin Starling Town Cottage Melton Road Great Melton NORWICH Norfolk NR9 3BT	
			Richard Terry Sheldrake White Cottage Melton Road WYMONDHAM Norfolk NR18 0SE (as reputed owner)		Richard Terry Sheldrake White Cottage Melton Road WYMONDHAM Norfolk NR18 0SE	
			Sally Jane Starling Town Cottage Melton Road Great Melton NORWICH Norfolk NR9 3BT (as reputed owner)		Sally Jane Starling Town Cottage Melton Road Great Melton NORWICH Norfolk NR9 3BT	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-005 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner) Unknown		The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ Unknown	
33-006	Temporary Possession	1295 square metres of hedgerow (west of Melton Road) (South Norfolk District Council)	Great Melton Parish Council Matsall House Church Lane Mattishall DEREHAM NR20 3QZ	Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	Unknown (in respect of restrictive covenants as may have been imposed before 16 September 2011)
33-007	Rights D	13970 square metres of arable land and hedgerow (west of Melton Road) (South Norfolk District Council)	Great Melton Parish Council Matsall House Church Lane Mattishall DEREHAM NR20 3QZ	Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	Robert Richardson Whiterails Farm Great Melton NORWICH Norfolk NR9 3BX	Unknown (in respect of restrictive covenants as may have been imposed before 16 September 2011)

Page 639 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	it Consent Order is to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu Fc	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-008	Rights D	17097 square metres of arable land, ditch and verge (west of Melton Road) (South Norfolk District Council)	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	NONE	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
33-009	Temporary Possession	476 square metres of arable land and verge (west of Melton Road) (South Norfolk District Council)	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	NONE	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	NONE
33-010	Rights D	826 square metres of public highway (Melton Road) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-010 cont'd			Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU (in respect of sub soil beneath public highway) Unknown			Unknown
33-011	Temporary Possession	439 square metres of arable land and verge (east of Melton Road) (South Norfolk District Council)	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	NONE	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) i	is Offshore Wind Farm Development ERENCE - PART 1 ight to use the land, and/or (iii) rights / of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	nning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-011 cont'd						Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-012	Rights D	23630 square metres of arable land, hedgerow and verge (east of Melton Road) (South Norfolk District Council)	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	NONE	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-012 cont'd						Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)
33-013	Rights D	820 square metres of arable land (east of Melton Road) (South Norfolk District Council)	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) rights / of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-013 cont'd						Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Persimmon Homes Limited Persimmon House Fulford YORK North Yorkshire YO19 4FE (in respect of rights granted)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			Count	y of Norfolk Category 1		
			Qualifying persons under Regu Fc	lation 7(1)(a) of the Infrastructure Plance Infrastructure Plance and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-013 cont'd						Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)
33-014	Rights D	39755 square metres of arable land, drain and hedgerow (east of Melton Road) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-014 cont'd			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ			
33-015	Rights D	10988 square metres of arable land (north of Ketts Oak, B1172) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
			Qualifying persons under Regul Fo	Category 1 ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
33-015 cont'd			Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE						
			The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ						

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-016	Rights D	2637 square metres of woodland (north of Ketts Oak, B1172) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	NONE	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP	NONE
33-017	Rights D	28370 square metres of arable land, drain and hedgerow (north of Ketts Oak, B1172) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Great Melton Farms Limited Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Great Melton Farms Hall Farm Pockthorpe Road Great Melton NORWICH NR9 3BW	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of right of way over access track)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-017 cont'd			Henrietta Katherine Norman 90 Christchurch Avenue LONDON NW6 7PE The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ			Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU (in respect of right of passage of services through conducting media)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-001	Rights D	19636 square metres of arable land and hedgerow (north of Ketts Oaks, B1172) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY	NONE	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right of access to lay and maintain apparatus)
34-002	Rights D	1243 square metres of verge (Ketts Oak, B1172) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-003	Temporary Possession	2192 square metres of arable land (north of Ketts Oak, B1172) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY	NONE	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right of access to lay and maintain apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-004	Temporary Possession	117 square metres of public highway (Ketts Oaks, B1172) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-004 cont'd			Peter Gowing and Partners Park Farm Hethersett NORWICH Norfolk NR9 3DL (in respect of sub soil beneath half width of public highway)			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)			
			Unknown			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-005	Temporary Possession	139 square metres of public highway (Ketts Oaks, B1172) (South Norfolk District Council)	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of street apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			(as highway authority) Unknown			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-006	Rights D	2059 square metres of public highway (Ketts Oaks, B1172) and verge (South Norfolk District Council)	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-006 cont'd						Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of street apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
34-007	Temporary Possession	38 square metres of private road (south of Ketts Oaks, B1172) (South Norfolk District Council)	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU	NONE	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-008	Temporary Possession	2987 square metres of arable land and hedgerow (south of Ketts Oaks, B1172) (South Norfolk District Council)	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU	NONE	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
34-009	Rights D	34350 square metres of arable land and hedgerow (south of Ketts Oaks, B1172) (South Norfolk District Council)	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU	NONE	Martin Paul Kemp Thickthorn Farm Norwich Road Hethersett NORWICH Norfolk NR9 3AU	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	it Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-010	Rights D	50248 square metres of arable land and hedgerow (north of the A11) (South Norfolk District Council)	Peter Gowing and Partners Park Farm Hethersett NORWICH Norfolk NR9 3DL Unknown	NONE	Peter Gowing and Partners Park Farm Hethersett NORWICH Norfolk NR9 3DL Unknown	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown
35-001	Rights D	20393 square metres of arable land (west of Ketteringham Lane and north of the A11) (South Norfolk District Council)	Foster Harrison Farms Limited High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	NONE	Foster Harrison Farms Limited High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Foster Harrison Farms Limited)
35-002	Rights D	4801 square metres of public highway (A11), verge and hedgerows (west of Ketteringham Lane) (excluding all interests of the Crown) (South Norfolk District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	NONE	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-002 cont'd						
35-003	Rights D	2626 square metres of rail line and verge (south of the A11) (South Norfolk District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner)	NONE	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW	Unknown
			Unknown		Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-004	Temporary Possession	435 square metres of private road (west of Ketteringham Lane and south of A11) (South Norfolk District Council)	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	NONE	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	Unknown (in respect of rights granted as contained in a Deed dated 27th June 1983)
35-005	Temporary Possession	34 square metres of private road (west of Ketteringham Lane) (South Norfolk District Council)	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP (as reputed owner)	NONE	Unknown	Chris Thrower Forgestone House Low Street Ketteringham WYMONDHAM NR18 9RY (in respect of rights of access and to draw water from an ancient well) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-006	Rights D	40884 square metres of Land and buildings on the north side of arable land, private road and verge (west of High Street) (South Norfolk District Council)	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	NONE	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	Unknown (in respect of rights granted as contained in a Deed dated 27th June 1983)
35-007	Rights D	1076 square metres of public highway (High Street) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-007 cont'd			Unknown			
35-008	Temporary Possession	87 square metres public highway (High Street) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Timothy Mark Hornor High Ash Farm High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP (in respect of sub soil beneath public highway) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-009	Temporary Possession	308 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	NONE	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Fo	orms and Procedures) Regulations 20	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-009 cont'd						Unknown (in respect of right to use drainage system, right to enter and right to lay and maintain water mains and pipes stated in conveyance dated 10th June 1959)
35-010	Rights D	24020 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	NONE	Timothy Mark Hornor High Ash Farm High Ash Road Ketteringham WYMONDHAM Norfolk NR18 9RP	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-010 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948) Unknown (in respect of right to use drainage system, right to enter and right to lay and maintain water mains and pipes stated in conveyance dated 10th June 1959)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-011	Rights D	107807 square metres of private road (Church Road), woodland, hedgerow, verge and public footpath (Ketteringham FP4) (excluding all interests of the Crown) (South Norfolk District Council)	Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Ketteringham FP4)) Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
				ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2		Category 2
Number on Extent of acquisition or Des Land Plans use	Description of land Owners or Reputed Ov	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009	
35-011 cont'd						Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication Sharon Gowing Park Lodge Ketteringham Park Ketteringham WYMONDHAM NR18 9PF (in respect of right of way over access track and drainage rights as contained in a Conveyance dated 5th January 1959)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Requ	Category 1 Ilation 7(1)(a) of the Infrastructure Pla	anning (Applications: Prescribed	Category 2
			Fc	orms and Procedures) Regulations 2	009	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
35-011 cont'd						The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)
36-001	Temporary Possession	4970 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	NONE	Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights)



Doc. No. C282-DM-Z-GA-00003

				County of Norfolk Category 1		
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure P ms and Procedures) Regulations 2	lanning (Applications: Prescribed 2009	Category 2
Number on Exte Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-001 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications Sharon Gowing Park Lodge Ketteringham Park Ketteringham Mark Ketteringham WYMONDHAM NR18 9PF (in respect of right of way over access track and drainage rights as contained in a Conveyance dated 5th January 1959) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-002	Temporary Possession	226 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU (in respect of sub soil beneath public highway) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-003	Temporary Possession	266 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU (in respect of sub soil beneath public highway) Unknown	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-004	Rights D	1002 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			Unknown			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-005	Temporary Possession	7623 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	NONE	Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-005 cont'd						Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)
36-006	Rights D	56249 square metres of verge, arable land and private road (east of Hethersett Road) (South Norfolk District Council)	Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	NONE	Simon Peter Moores Ivyhouse Farm High Street Ketteringham WYMONDHAM Norfolk NR18 9RU	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) rig County	of Norfolk Category 1	· · · · · · · · · · · · · · · · · · ·	
			Qualifying persons under Regula Forr	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Extent of acquisition or Land Plans use Descripti	Description of land Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
36-006 cont'd						Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of gas main and right way) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fc	anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-006 cont'd						Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)
36-007	Temporary Possession	4812 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Charles Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF	NONE	Charles Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-007 cont'd			Judith Molly Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF		Judith Molly Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipes)



Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	For	ms and Procedures) Regulations 20	009 Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-007 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights of access to lay and maintain electricity cable and restrictive covenants)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparate
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-007 cont'd						Unknown (in respect of right of access and timber rights) Unknown (in respect of rights reserved)
36-008	Temporary Possession	254 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Charles Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF (in respect of sub soil beneath public highway) Judith Molly Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-008 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Unknown			
36-009	Rights D	19618 square metres of hedgerow and arable land (east of Hethersett Road) (South Norfolk District Council)	Charles Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF	NONE	Charles Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights of access to lay and maintain electricity cables and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Preso Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-009 cont'd			Judith Molly Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF		Judith Molly Bussey Dimora Chandler Road Stoke Holy Cross NORWICH Norfolk NR14 8RF	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of access and timber rights) Unknown (in respect of rights reserved)
36-010	Rights D	9643 square metres of hedgerow and arable land (east of Hethersett Road) (South Norfolk District Council)	Clive Kennedy Chapman Manor Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8JY	NONE	Clive Kennedy Chapman Manor Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8JY	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii) i	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-010 cont'd			Hazel Patricia Chapman Manor Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8JY		Hazel Patricia Chapman Manor Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8JY	
36-011	Rights D	55959 square metres of arable land, hedgerows, verge and public footpath (East Carleton FP4) (west of Intwood Lane) (South Norfolk District Council)	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	NONE	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	it Consent Order is to carry out protective works (Reg	ulation 7(1)(a))
				Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
36-011 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (East Carleton FP4))	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
37-001	Temporary Possession	903 square metres of arable land and public footpath (East Carleton FP4) (west of Intwood Lane) (South Norfolk District Council)	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	NONE	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	it Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
37-001 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (East Carleton FP4))	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
37-002	Rights D	808 square metres of verge and public highway (Intwood Lane) (South Norfolk District Council)	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
37-002 conťd			Unknown			Unknown	
37-003	Temporary Possession	553 square metres of arable land (east of Intwood Lane) (South Norfolk District Council)	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	NONE	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fc	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
37-004	Temporary Possession	25 square metres verge (south of Intwood Lane) (South Norfolk District Council)	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA (as reputed owner) Unknown	NONE	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA (as reputed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown
37-005	Rights D	27480 square metres of hedgerow, arable land and verge (east of Intwood Lane) (South Norfolk District Council)	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	NONE	John Edward Crane North Farm Intwood Lane East Carleton NORWICH Norfolk NR14 8LA	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) r County	of Norfolk Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed)09	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009
37-005 cont'd						Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to b and maintain gas mains)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Plance Inns and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
37-006	Rights D	82917 square metres of public footpath (Swardeston FP2, Swardeston FP3 and Swardeston FP4), hedgerow, watercourse (unnamed) and arable land (west of Norwich Road, B1113) (South Norfolk District Council)	Hethel Farming Limited Ketteringham Hall Church Road Ketteringham WYMONDHAM Norfolk NR18 9RS	NONE	Hethel Farming Limited Ketteringham Hall Church Road Ketteringham WYMONDHAM Norfolk NR18 9RS	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)
					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpaths (Swardeston FP2, Swardeston FP3 and Swardeston FP4))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
						Unknown (in respect of rights as contained in a Conveyance dated 21st October 1902)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-001	Temporary Possession	276 square metres of access track, verge and arable land (west of Main Road, B1113) (South Norfolk District Council)	Hethel Farming Limited Ketteringham Hall Church Road Ketteringham WYMONDHAM Norfolk NR18 9RS	NONE	Hethel Farming Limited Ketteringham Hall Church Road Ketteringham WYMONDHAM Norfolk NR18 9RS	Unknown (in respect of rights as contained in a Conveyance dated 21st October 1902)
38-002	Rights D	1135 square metres of verge and public highway (Main Road, B1113) (South Norfolk District Council)	Hethel Farming Limited Ketteringham Hall Church Road Ketteringham WYMONDHAM Norfolk NR18 9RS (in respect of sub soil beneath half width of public highway) Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-002 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF (in respect of sub soil beneath half width of public highway) Unknown			Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ERENCE - PART 1 right to use the land, and/or (iii) righ		ulation 7(1)(a))
					Category 2
Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
Temporary Possession	160 square metres of verge and arable land (east of Main Road, B1113) (South Norfolk District Council)	Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF	NONE	Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)
		Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF		Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF	
	Extent of acquisition or use	Land which is proposed to be subject to: (i) possible Extent of acquisition or use Description of land Temporary Possession 160 square metres of verge and arable land (east of Main Road, B1113) (South Norfolk	Extent of acquisition or use Description of land Qualifying persons under Reguler Temporary Possession 160 square metres of verge and arable land (east of Main Road, B1113) (South Norfolk District Council) Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH	BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) righ County of Norfolk Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) righ County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure PI Forms and Procedures) Regulations 2 Category 1 Extent of acquisition or use Description of land Owners or Reputed Owners Lessees or Tenants Temporary Possession 160 square metres of verge and arable land (east of Main Road, Bi1113) (South Norfolk District Council) Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF NONE Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH NORWICH	Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Reg Courty of Norfolk Extent of acquisition or use Description of land Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 Temporary Possession 160 square metres of verge and arable land (east of Main Road, B1113) (South Norfolk Margaret Elizabeth Bobbin Swardeston NORWICH NONE Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH NORWICH Patrick Bobbin Nordok Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH NORWICH NORWICH NORWICH

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-004	Rights D	16683 square metres of arable land and hedgerow (east of Main Road, B1113) (South Norfolk District Council)	Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF	NONE	Margaret Elizabeth Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)
			Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF		Patrick Bobbin Benchmark Cottage Main Road Swardeston NORWICH NR14 8DF	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
38-005	Rights D	8712 square metres of arable land and hedgerow (east of Main Road, B1113) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Nicola McDonnell 19 Pheasant Close Mulbarton NORWICH NR14 8BL	NONE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-005 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP			
38-006	Rights D	63248 square metres of public footpaths (Mulbarton FP6 and Mulbarton FP7), arable land and hedgerows (south east of Main Road, B1113) (South Norfolk District Council)	Peddars Pigs Limited Unit 9 Dalton House 60 Windsor Avenue LONDON SW19 2RR	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpaths (Mulbarton FP6 and Mulbarton FP7))	Clydesdale Bank PLC 30 St. Vincent Place GLASGOW G1 2HL (as mortgagee for Peddars Pigs Limited)



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-006 cont'd					Peddars Pigs Limited Unit 9 Dalton House 60 Windsor Avenue LONDON SW19 2RR	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
38-007	Rights D	3193 square metres of arable land and hedgerow (west of Swardeston Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-007 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
38-008	Temporary Possession	1398 square metres of arable land (west of Swardeston Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-008 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	
38-009	Temporary Possession	83 square metres of public highway (Swardeston Lane and Gowthorpe Lane) and verge (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath public highway)	NONE		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) righ		ulation 7(1)(a))
	County of Norfolk Category 1					
				lation 7(1)(a) of the Infrastructure P rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-009 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of sub soil beneath public highway) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of sub soil beneath public highway) Unknown		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
38-010	Temporary Possession	17 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) (in respect of public byway (Swainsthorpe BOAT6))	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) (in respect of public byway (Swainsthorpe BOAT6))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk							
				Category 1 ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
38-010 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of sub soil beneath public highway) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of sub soil beneath public highway) Unknown					

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
					Category 2			
ttent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
	public highway (Swardeston Lane) and public byway	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as bichway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) (in respect of public byway (Swainsthorpe BOAT6))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to			
	use orary Possession	use Description of land prary Possession 2 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District	The of acquisition or use Description of land Owners or Reputed Owners orary Possession 2 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ Norfolk County Council County Hall Martineau Lane NORWICH NORWICH Norfolk County Council County Hall Norfolk	Int of acquisition or use Description of land Owners or Reputed Owners Lessees or Tenants Drary Possession 2 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ NONE Norfolk District Council) Norfolk County Council County Hall Martineau Lane NORWICH NORVICH NR12DH Norfolk County Council County Hall Martineau Lane NORWICH NR12DH	Autor of acquisition or use Description of land Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 orary Possession Description of land Owners or Reputed Owners Lessees or Tenants Occupiers orary Possession 2 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH Norfolk District Council) NONE Norfolk County Council (Swainsthorpe BOAT6) (South Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR14 8DZ Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR12DH Norfolk NR12DH Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR12DH Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR12DH Norfolk NR12DH			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2(Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-011 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock			Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
			CHIPPENHAM SN15 2PP			
38-012	Temporary Possession	42 square metres of private road (east of Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				ERENCE - PART 1		
	Land which is propo	sed to be subject to: (i) po	owers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) right / of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-012 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public byway (Swainsthorpe BOAT6))	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1	it Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-013	Temporary Possession	1149 square metres of public footpath (Swainsthorpe FP1), verge and arable land (east of Swardeston Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swainsthorpe FP1))	
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-013 conťď					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	
38-014	Rights D	604 square metres of public highway (Swardeston Lane) and verge (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of sub soil beneath half width of public highway) Norfolk County Council County Hall Martineau Lane NORWICH	NONE	Norfolk County Council County Hall Martineau Lane NORWICH	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON
			Norfolk NR1 2DH (as highway authority)		Norfolk NR1 2DH (as highway authority)	WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		Category 2
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-014 cont'd			Peddars Pigs Limited Unit 9 Dalton House 60 Windsor Avenue LONDON SW19 2RR (in respect of sub soil beneath half width of public highway)			Unknown
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of sub soil beneath half width of public highway)			
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of sub soil beneath half width of public highway)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fc	lation 7(1)(a) of the Infrastructure Planner and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-014 conťd			Unknown			
38-015	Rights D	6 square metres of verge (east of Swardeston Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner) Jennifer Ruth Sheldrake Vale Farm Swainsthorpe NORWICH NR14 8PT (as reputed owner) Mark Sheldrake Vale Farm Swainsthorpe NORWICH NR14 8PT (as reputed owner)	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ Jennifer Ruth Sheldrake Vale Farm Swainsthorpe NORWICH NR14 8PT Mark Sheldrake Vale Farm Swainsthorpe NORWICH NR14 8PT	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ılation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-015 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP Unknown	
38-016	Rights D	61 square metres of verge (east of Swardeston Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))	
			Qualifying persons under Regul For	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
38-016 cont'd			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP			Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Regu	ılation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Plance and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-017	Rights D	22746 square metres of arable land (north of Church Road) and public footpath (Swainsthorpe FP1) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swainsthorpe FP1))	
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	

Page 710 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
38-017 cont'd					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	
39-001	Rights D	19624 square metres of arable land (south of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner) Unknown	Paul David Watkinson Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	Paul David Watkinson Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS Unknown	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus Unknown



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-002	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE
39-003	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE
39-004	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pla orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-005	Rights D	4101 square metres of arable land and verge (south of Hickling Lane and west of Ipswich, A140) (South Norfolk District Council)	Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)	Paul David Watkinson Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	Paul David Watkinson Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	Unknown
			Unknown		Unknown	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)
						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights granted)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	rms and Procedures) Regulations 20 Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-006	Rights D	8397 square metres of arable land and verge (south of Hickling Lane and west of Ipswich, A140) (South Norfolk District Council)	Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)	Paul David Watkinson Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	Paul David Watkinson Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	Unknown
			Unknown		Unknown	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights granted)
						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)
39-007	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE

Page 714 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Ilation 7(1)(a) of the Infrastructure Pl orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-008	Freehold Acquisition	308 square metres of verge, hedgerow and public byway open to all traffic (Swainsthorpe BOAT6) (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner)	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)
			Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)		Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order nts to carry out protective works (Regu	ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 ation 7(1)(a) of the Infrastructure F rms and Procedures) Regulations 3	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-008 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of byway (Swainsthorpe BOAT6)) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Swittana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REI	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ ry of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-009	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE	NUMBER NOT IN USE
39-010	Rights D	1 square metres of land and public byway open to all traffic (Swainsthorpe BOAT6) (Hickling Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner) Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Regu	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-010 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of byway (Swainsthorpe BOAT6)) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prop		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	lanning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-011	Rights D	455 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner)	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Unknown
			Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)		Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of byway (Swainsthorpe BOAT6))	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Regu	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-011 cont'd			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-012	Freehold Acquisition	464 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner)	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Unknown
			Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner) Rosemary Margaret Watkinson		Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS Norfolk County Council	
			Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)		County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of byway (Swainsthorpe BOAT6))	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-012 cont'd			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-013	Freehold Acquisition	67 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner)	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Unknown
			Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)		Griffin Farms Limited Hall Farm Gowthorpe Lane Swardeston NORWICH NR14 8DS	
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of byway (Swainsthorpe BOAT6))	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ılation 7(1)(a))
			Qualifying persons under Regul Fo	Category 1 lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-013 cont'd			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

				ERENCE - PART 1		
	Land which is propo	osed to be subject to: (i) po		right to use the land, and/or (iii) right y of Norfolk	s to carry out protective works (Reg	llation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-014	Freehold Acquisition	112837 square metres of arable land and part of public byway open to all traffic (Swainsthorpe BOAT6) (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 17 December 2019)
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public byway (Swainsthorpe BOAT6))	Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-014 cont'd					Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 17 December 2019)
					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	Unknown (in respect of right of way over Hickling Lane reserved)
39-015	Freehold Acquisition	52057 square metres of arable land and hedgerow (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order nts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 ation 7(1)(a) of the Infrastructure F rms and Procedures) Regulations		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-015 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Swainsthorpe BR7))	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)

Page 727 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1					
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-016	Rights G	936 square metres of hedgerow and arable land (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)			
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Swainsthorpe BR7))	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)			
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	t Consent Order s to carry out protective works (Regu	ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-016 cont'd					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	
39-017	Rights I	1005 square metres of hedgerow and arable land (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Swainsthorpe BR7))	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	·	BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-017 cont'd			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	
39-018	Freehold Acquisition	5791 square metres of arable land (north of Hickling Lane) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed		
				ms and Procedures) Regulations 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-018 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Swainsthorpe BR7))	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	t Consent Order s to carry out protective works (Reg	ulation 7(1)(a))	
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-019	Rights E	83657 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)	

Page 732 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-019 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
39-020	Rights E and G	1227 square metres of arable land and hedgerow (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	Category 1 ation 7(1)(a) of the Infrastructure F ms and Procedures) Regulations 2	lanning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-020 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk								
				Category 1 lation 7(1)(a) of the Infrastructure PI rms and Procedures) Regulations 2		Category 2			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-021	Freehold Acquisition	5569 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)			
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				lation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-021 cont'd			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
39-022	Freehold Acquisition	5093 square metres of arable land (west of lpswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-022 cont'd			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
39-023	Rights E and I	3018 square metres of woodland (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	NONE	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	owers of compulsory acquisition, (ii) ri	ERENCE - PART 1 ght to use the land, and/or (iii) righ of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure P ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-023 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Swainsthorpe BR7))	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 10 September 2020)
					43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-024	Rights G	605 square metres of arable land (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 17 December 2019)
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 17 December 2019)
					Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP	Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ (as beneficiary of option agreement dated 22 December 2004)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk	it Consent Order is to carry out protective works (Regi	ulation 7(1)(a))
			Qualifying persons under Regu	Category 1 Ilation 7(1)(a) of the Infrastructure PI orms and Procedures) Regulations 2	anning (Applications: Prescribed	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-025	Rights I	15 square metres of public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)
					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Stoke Holy Cross BR3))	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Plance ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-025 cont'd						Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-025 cont'd						Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track) Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-026	Rights I	56 square metres of public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleway (Stoke Holy Cross BR3))	Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-026 cont'd			National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (as reputed owner) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner)		Unknown	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of a right of way) Unknown
			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)			Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
		Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-026 cont'd			Unknown			
39-027	Temporary Possession	24 square metres of public bridleway (Stoke Holy Cross BR3) and hedgerow (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleway (Stoke Holy Cross BR3))	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo	owers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Extent of acquisition or use	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-027 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (as reputed owner) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as reputed owner)		Unknown	Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of a right of way)
		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)			Unknown



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right		ulation 7(1)(a))
			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed			Category 2
Number on Land Plans		Owners or Reputed Owners	orms and Procedures) Regulations 2	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-027 cont'd			Unknown			Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)
39-028	Rights E and I	5153 square metres of woodland, hedgerow and public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Stoke Holy Cross BR3))	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-028 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) r County	of Norfolk Category 1	с се салу салросские поло (тез	
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Plans and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-028 cont'd						Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunication



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-028 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-029	Rights I	9833 square metres of woodland, hedgerow and public bridleways (Stoke Holy Cross BR3 and Swardeston BR12) (south of Norwich Main Substation and west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 ation 7(1)(a) of the Infrastructure P ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-029 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Stoke Holy Cross BR3 and Swardeston BR12))	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul Fo	ation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-029 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane
						Swardeston NORWICH NR14 8DS (in respect of right of way over access track)
						Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-029 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-030	Rights I	203 square metres of public bridleway (Swardeston BR12) (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (as reputed owner)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleway (Swardeston BR12)) Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	l and which is propo		gham Shoal and Dudgeon Extension BOOK OF REF owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
	Eand Which is prope		County	of Norfolk		
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-030 cont'd			Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Land Plans	Locorintion of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-031	Rights E and I	654 square metres of public bridleway (Swardeston BR12) (west of Ipswich Road, A140) (South Norfolk District Council)	Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (as reputed owner) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (as reputed owner) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (as reputed owner)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleway (Swardeston BR12)) Unknown	Unknown

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regu Fo	llation 7(1)(a) of the Infrastructure PI orms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-031 cont'd			Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (as reputed owner) Unknown			
39-032	Rights E and I	1979 square metres of arable land and hedgerow (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation) Centrica PLC Millstream Maidenhead Road WINDSOR
						Berkshire SL4 5GD (in respect of easement)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-032 cont'd						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for leas contained in option agreement dated 29 March 2019)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				Category 1		
				lation 7(1)(a) of the Infrastructure Plance Inns and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-032 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-033	Rights E and I	1925 square metres of arable land, hedgerow and public bridleway (Swardeston BR12) (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media)
			Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public footpath (Swardeston BR12))	Cerulea Limited 5 Howick Place LONDON SW1P 1WG (in respect of an option to grant lease as contained in an Option Agreement dated 9th September 2021)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure ms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-033 cont'd					Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
				tion 7(1)(a) of the Infrastructure Pla ns and Procedures) Regulations 20		Category 2
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-033 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to I and maintain gas pipelines and restrictive covenants) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of an option to grant lease as contained in an Option Agreement dated 9th Septembe 2021) Unknown (in respect of rights granted, reserved and restrictive covenants)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-034	Rights F	40840 square metres of arable land (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (in respect of an option to grant lease as contained in an Option Agreement dated 9th September 2021) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)



Doc. No. C282-DM-Z-GA-00003

			wers of compulsory acquisition, (ii) r County	of Norfolk		
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Extent of acquisition or Description of Land Plans	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Form and Procedures) Regulations 2009	
39-034 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right fell or lop trees and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (1) po	wers of compulsory acquisition, (ii) rig County	of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Extent of acquisition or Land Plans use Desc	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-034 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatu National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to la and maintain gas pipelines and restrictive covenants) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of an option to grant lease as contained in an Option Agreement dated 9th September 2021)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-034 cont'd						Unknown (in respect of rights granted, reserved and restrictive covenants)
39-035	Rights F	15126 square metres of arable land and hedgerow (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-035 cont'd						Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)
						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for leas contained in option agreement dated 29 March 2019)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparate



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	ms and Procedures) Regulations 2 Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-035 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)
						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-036	Temporary Possession	5908 square metres of arable land (west of lpswich Road, A140) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of right to maintain gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r County	ight to use the land, and/or (iii) righ of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-036 cont'd						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (in respect of an option to grant lease as contained in an Option Agreement dated 9th September 2021) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-036 cont'd						National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
				llation 7(1)(a) of the Infrastructure Planner and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-036 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of an option to grant lease as contained in an Option Agreement dated 9th September 2021) Unknown (in respect of rights granted, reserved and restrictive covenants)
39-037	Rights F	25343 square metres of Norwich Main Substation and private road (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation)



Doc. No. C282-DM-Z-GA-00003

Rev. H

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	anning (Applications: Prescribed 009	Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-037 cont'd						Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)
						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)

Page 770 of 1368



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	esed to be subject to: (i) po	owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) righ [,] of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-037 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)
						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REI	ns Offshore Wind Farm Developmer FERENCE - PART 1 right to use the land, and/or (iii) righ y of Norfolk	nt Consent Order ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 lation 7(1)(a) of the Infrastructure P orms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-038	Rights G	4098 square metres of verge, hedgerow and public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleway (Stoke Holy Cross BR3))	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-038 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-038 cont'd						Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-038 cont'd						Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu Fo	Category 1 lation 7(1)(a) of the Infrastructure Pla rms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-038 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-039	Temporary Possession	2117 square metres of hedgerow and verge (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) ri	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Plants and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-039 cont'd						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-039 cont'd						Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)
						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)
						Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) rights y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-039 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-040	Temporary Possession	634 square metres of hedgerow and verge (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE		Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) ri	ERENCE - PART 1	ent Consent Order hts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1 ation 7(1)(a) of the Infrastructure F ms and Procedures) Regulations	Planning (Applications: Prescribed 2009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-040 cont'd					Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of public bridleways (Stoke Holy Cross BR3 and Swardeston BR12)) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Cerulea Limited
						5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	owers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right r of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-040 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-040 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications)
						Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Regu	Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-040 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-041	Temporary Possession	7492 square metres of private road and grassland (Norwich Main Substation and west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) rig County	of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applica Forms and Procedures) Regulations 2009		anning (Applications: Prescribed 009	Category 2
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-041 cont'd						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-041 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications) Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

			Count	cy of Norfolk Category 1		
			Qualifying persons under Regu Fo	Category 2		
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-042	Temporary Possession	3223 square metres of private road and verge (west of Ipswich Road, A140 and south of Mangreen Road) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) ri	of Norfolk	ts to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-042 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
				ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-042 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications Vodafone Limited Vodafone House



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REI owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-042 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
39-043	Temporary Possession	14336 square metres of verge, grassland and hedgerows (Norwich Main Substation, west of Ipswich Road, A140 and south of Mangreen Road) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	sed to be subject to: (i) po	wers of compulsory acquisition, (ii) rig	RENCE - PART 1 ght to use the land, and/or (iii) rights of Norfolk	to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Number on Extent of acquisition or Land Plans use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-043 cont'd						Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019)



Doc. No. C282-DM-Z-GA-00003

			County	of Norfolk Category 1		
			Qualifying persons under Regula For	ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-043 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)
						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparate
						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		wers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-043 cont'd						Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications) Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

			County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-044	Rights G	2901 square metres of private road (west of Ipswich Road and south of Mangreen Road) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN (in respect of right of access) Brockwell Energy Limited Caledonian Exchange 19a Canning Street Edinburgh EH3 8EG (as beneficiary in respect of an Option Agreement to connect a cable to the substation) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo	osed to be subject to: (i) po	wers of compulsory acquisition, (ii) r	ERENCE - PART 1 ight to use the land, and/or (iii) right of Norfolk	s to carry out protective works (Reg	ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pl ms and Procedures) Regulations 2		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-044 cont'd						Cerulea Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii) r	is Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right / of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-044 cont'd						Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (as beneficiary of option for lease contained in option agreement dated 29 March 2019) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-044 cont'd						Ashgreen Energyfarm Limited Connect House 133-137 Alexandra Road LONDON SW19 7JY (as beneficiary of option agreement dated 13 November 2019)
40-001	Temporary Possession	16719 square metres of grassland (west of Ipswich Road, A140) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of right to maintain gas apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-001 cont'd			Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ	Sadie Ann Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media)



Doc. No. C282-DM-Z-GA-00003

	County of Norfolk Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Plannin Forms and Procedures) Regulations 2009					Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-001 cont'd						Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right f fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right f fell or lop trees and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF wers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Qualifying persons under Degu	Category 1	onning (Applications: Drocerihod	Category 2
				lation 7(1)(a) of the Infrastructure Pla orms and Procedures) Regulations 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-001 cont'd						National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)
40-002	Temporary Possession	5925 square metres of public highway (Ipswich Road, A140 and Mangreen Road) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ (in respect of sub soil beneath half width of public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-002 cont'd			National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of sub soil beneath half width of public highway)			Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)
			Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of sub soil beneath half width of public highway)			National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)
			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-002 cont'd			Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN (in respect of sub soil beneath half width of public highway) Unknown			Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-003	Temporary Possession	126 square metres of verge (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (as reputed owner)	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			Unknown		Unknown	Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-004	Rights G	19 square metres of private road (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (as reputed owner) Unknown	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
40-005	Temporary Possession	12 square metres of verge (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (as reputed owner)	NONE	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is prope		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-005 cont'd			Unknown		Unknown	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown
40-006	Temporary Possession	27 square metres of Mangreen Hall Farm, Swardeston, Norwich, NR14 8DD (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ	Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-006 cont'd						Unknown (in respect of rights granted, reserved and restrictive covenants)
40-007	Temporary Possession	161 square metres of verge and public highway (Mangreen Road) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ (in respect of sub soil beneath public highway)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)
			Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)			National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen ERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-007 cont'd			Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN (in respect of sub soil beneath public highway)			Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)
			Unknown			Unknown
40-008	Temporary Possession	5875 square metres of hedgerow and verge (Mangreen Hall Farm, west of Ipswich Road, A140) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ	Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ	Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii) r	ERENCE - PART 1		ulation 7(1)(a))
				Category 1		
				ation 7(1)(a) of the Infrastructure Pla ms and Procedures) Regulations 20		Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-008 cont'd			Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN			Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk					
				Category 1		
			Qualifying persons under Regul For	ation 7(1)(a) of the Infrastructure Pl rms and Procedures) Regulations 2	anning (Applications: Prescribed 009	Category 2
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-008 cont'd						GRS Building Products Limited 10 Goldsmith Way Eliot Business Park NUNEATON Warwickshire CV10 7RJ (as beneficiary of lease dated 29 September 2015) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	Land which is propo		BOOK OF REF owers of compulsory acquisition, (ii)	ns Offshore Wind Farm Developmen FERENCE - PART 1 right to use the land, and/or (iii) right y of Norfolk		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-008 cont'd						Unknown (in respect of rights granted, reserved and restrictive covenants)
40-009	Temporary Possession	219 square metres of verge and private road (west of Ipswich Road, A140) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN	Tarmac Aggregates Limited Portland House Bickenhill Lane BIRMINGHAM West Midlands B37 7BQ	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) GRS Building Products Limited 10 Goldsmith Way Eliot Business Park NUNEATON Warwickshire CV10 7RJ (as beneficiary of lease dated 29 September 2015)



Doc. No. C282-DM-Z-GA-00003

	Land which is propo		owers of compulsory acquisition, (ii)	ERENCE - PART 1		ulation 7(1)(a))
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2	
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-009 cont'd						Unknown (in respect of rights granted, reserved and restrictive covenants)
40-010	Temporary Possession	353 square metres of public highway (Ipswich Road, A140) (South Norfolk District Council)	Kerry Ann Gray Barley View Greenways Newton Flotman NORWICH Norfolk NR15 1QJ (in respect of sub soil beneath public highway) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	NONE	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (as highway authority)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Norfolk						
			Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		Category 2		
Number on Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7 (1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-010 cont'd			Sarah Jane Haistead Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN (in respect of sub soil beneath public highway) Unknown				

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
01-001	5165 square metres of foreshore land seaward of the Mean High Water (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted by lease dated 18 November 2009) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted by lease dated 18 November 2009)			
01-002	1523 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
01-002 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)			
01-003	32 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
01-003 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence		
01-004	1195 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access) Unknown		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
01-005	366 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
01-005 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)			
01-006	814 square metres of beach and public footpath (Weybourne FP7) (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)			

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-006 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-007	4767 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-007 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-008	765 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-008 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-009	2135 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Unknown	
01-010	1680 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-011	652 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018)	

Page 820 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-011 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-012	36 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-012 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-013	57 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-013 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-014	4655 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-014 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-015	393 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-016	33 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Ministry of Defence Ministry of Defence Ministry of Defence Ministry of Defence Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe over part)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-017	51 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-018	7854 square metres of access track (north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dereham and District Angling Club 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads) Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-018 cont'd		Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-018 cont'd		The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access)	
01-019	10156 square metres of access track and public footpath (Weybourne FP7) (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-019 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)
		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-020	9694 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dereham and District Angling Club 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-020 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads) Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk N28 9HZ (in respect of right of way over access roads) Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-020 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-021	75 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	

Page 835 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-021 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-022	24 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-022 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018)	
		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-023	630 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	
		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-023 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-024	3359 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-024 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-025	4474 square metres of grassland and access track (Weybourne Military Camp) (North Norfolk District Council)	Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-025 conťď		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	
01-026	265 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-026 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-027	304 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-027 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-028	5462 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-028 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	
		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-029	9686 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	
		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-029 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-030	8080 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electricity cables and junction point)	

Page 844 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-030 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of rights granted by lease dated 13 March 2018) Soira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-031	4437 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	

Page 846 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-031 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-032	21 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-032 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
		Unknown (in respect of right of access and right of passage of services through conducting media)	
01-033	9105 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-033 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-034	22426 square metres of grassland (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-034 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-035	15 square metres of public highway (The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-035 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-036	156 square metres of public highway (The Street, A149) (North Norfolk District Council)	Unknown	
01-037	29 square metres of access track (north of The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-037 cont'd		Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dereham and District Angling Club 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-037 cont'd		Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads) Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-037 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access) Unknown	
01-038	707 square metres of public highway and verges (The Street, A149) (North Norfolk District Council)	Unknown	
01-039	1 square metres of access track (south of The Street, A149) (North Norfolk District Council)	Unknown	
01-040	3 square metres of access track (south of The Street, A149) (North Norfolk District Council)	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-040 cont'd		Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables) Unknown	
01-041	285 square metres of access track and land (south of The Street, A149) (North Norfolk District Council)	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-041 cont'd		Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
01-043	486 square metres of access track, woodland and arable land (south of The Street, A149) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
01-043 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
01-044	36555 square metres of arable land and hedgerow (north of Holgate Hill) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
02-001	2057 square metres of arable land (north of Holgate Hill) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-001 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
02-002	442 square metres of public highway and verges (Holt Road) (North Norfolk District Council)	Unknown	
02-003	2844 square metres of arable land, hedgerow and verges (north of Holgate Hill) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
02-005	816 square metres of arable land and hedgerow (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-005 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
02-006	323 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Unknown	
02-007	305 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) Unknown	
02-008	45 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-009	730 square metres of track, arable land and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) Unknown (in respect of restrictive covenants stated in conveyance dated 23rd December 1954)	
02-010	45859 square metres of arable land, grassland, hedgerow, watercourse (Spring Beck) and verges (east of Holt Road and west of Station Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted in deed dated 12th August 2011) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted in deed dated 12th August 2011) Unknown (in respect of restrictive covenants stated in conveyance dated 23rd December 1954)	
02-012	871 square metres of public highway and verges (Station Road) (North Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-014	2367 square metres of arable land, access track and verges (east of Station Road) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)	
02-015	60663 square metres of arable land, access track, woodland and hedgerows (east of Station Road) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)	
03-001	214 square metres of public highway and verges (Sheringham Road, A149) (North Norfolk District Council)	Unknown	
03-002	8425 square metres of access track and verges (south of Sheringham Road, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)	
03-003	653 square metres of railway track, compound and verges (north east of Weybourne Train Station) (North Norfolk District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-004	1829 square metres of railway track, compound and verges (north east of Weybourne Train Station) (North Norfolk District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants) Unknown	
03-006	77 square metres of public highway and verges (Sandy Hill Lane) (North Norfolk District Council)	Unknown	
03-007	22 square metres of access track (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	
03-010	15195 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-010 cont'd		Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 75Z (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media) Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Rediand Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
03-010 cont'd		Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turmbull 9 Cowslip Lane Gamilingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) Susan Rodda Burton 60 Fitzwalter Road Fitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-010 cont'd		Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wittshire SN2 2NA (in respect of right of way) Theresa Patricia Mason 41 Arington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of right and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-002	1869 square metres of woodland and track (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Fitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-002 cont'd		Ivy Elizabeth Black S8A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Garnlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-002 cont'd		Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 ZNA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SWIP 4DF (in respect of right of way)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-002 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-003	38365 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-003 cont'd		Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media) Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-003 cont'd		Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-003 cont'd		Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 ZNA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SWIP 4DF (in respect of right of way)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-003 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-004	9 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-004 cont'd		Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media) Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-004 cont'd		Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-004 cont'd		Susan Rodda Burton 60 Fitzwalter Road Filtch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wittshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-004 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-005	13 square metres of public highway and verge (Sandy Hill Lane) (North Norfolk District Council)	Unknown	
04-006	43 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	
04-007	10 square metres of public highway (Sandy Hill Lane) (North Norfolk District Council)	Unknown (in respect to a right of way/passage over the track)	
04-008	13 square metres of public highway (Sandy Hill Lane) (North Norfolk District Council)	Unknown (in respect to a right of way/passage over the track)	
04-009	2108 square metres of access track and verges (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown (in respect to a right of way/passage over the track)	
04-010	35 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-011	1355 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Filtch Green DUMMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-011 cont'd		Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-011 cont'd		Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 ZNA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SWIP 4DF (in respect of right of way)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-011 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown	
		(in respect of right of way and service medias) Unknown (in respect of rights and easements)	
		William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-012	67 square metres of hedgerow and verge (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	
04-013	465 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-013 cont'd		Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media) Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-013 cont'd		Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
04-013 cont'd		Susan Rodda Burton 60 Fitzwalter Road Filtch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-013 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-014	277 square metres of woodland and track (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	
04-015	46631 square metres of arable land, hedgerow and verges (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown (in respect to a right of way/passage over the track)	
05-001	182 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	Unknown	
05-004	2000 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	Unknown	
05-005	360 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	Unknown	
05-006	4033 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-007	1530 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Lemur UK LLC 10960 Wilshire Blvd 11th Floor LOS ANGELES CA 90024 UNITED STATES OF AMERICA (in respect of right of access for maintenance)	
05-008	6023 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Lemur UK LLC 10960 Wilshire Blvd 11th Floor LOS ANGELES CA 90024 UNITED STATES OF AMERICA (in respect of right of access for maintenance)	
05-009	34484 square metres of arable land and hedgerow (south of Holt Road, A148 and north of The Street) (North Norfolk District Council)	Unknown	
05-010	655 square metres of access track and hedgerow (north of The Street) (North Norfolk District Council)	Unknown	
05-011	27 square metres of public highway (The Street) (North Norfolk District Council)	Unknown	
05-012	811 square metres of public highway and verges (The Street) (North Norfolk District Council)	Unknown	
05-013	306 square metres of verges and hedgerow (south of The Street) (North Norfolk District Council)	Unknown	
05-016	23118 square metres of arable land and hedgerow (north of Norfolk Coast, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-016 cont'd		Unknown (in respect of right of access to lay and maintain electricity cables contained in transfer dated 28 July 1995)	
06-001	242 square metres of arable land and hedgerow (north of Norfolk Coast, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted) Unknown (in respect of right of access to lay and maintain electricity cables contained in transfer dated 28 July 1995)	
06-002	466 square metres of verges and public highway (Norfolk Coast, A149) (North Norfolk District Council)	Unknown	
06-003	65 square metres of verge (Norfolk Coast, A149) (North Norfolk District Council)	Unknown	
06-007	549 square metres of public highway (New Road, Holt, A149) (North Norfolk District Council)	Unknown	
07-001	16979 square metres of arable land and hedgerow (south east side of New Road, A149) (North Norfolk District Council)	Arthur Richard Gee The Bungalow Chestnut Farm Castle Road Baconsthorpe HOLT NR25 6LL (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-001 cont'd		BW Sipp Trustees Limited St. James House St. James Square CHELTENHAM Gloucestershire GL50 3PR (in respect of right of way) Margaret Thelma Gee The Bungalow Chestnut Farm Castle Road Baconsthorpe HOLT NR25 6L (in respect of right of way)
07-004	778 square metres of public highway being Gresham Road (north of Gresham Road) (North Norfolk District Council)	Unknown
07-006	16784 square metres of arable land and hedgerow (north of Gresham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-006 cont'd		Unknown (in respect of drainage rights reserved by transfer dated 25 March 2008)	
07-008	101 square metres of private track (north of Gresham Road and on the south side of New Road, A149) (North Norfolk District Council)	Unknown	
07-009	392 square metres of verges and public highway (Gresham Road) (North Norfolk District Council)	Unknown	
07-010	606 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-011	1788 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-011 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)
07-012	4 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Unknown
07-013	2 square metres of arable land (south of Gresham Road) (North Norfolk District Council)	Unknown
07-014	2345 square metres of arable land (south of Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-014 cont'd		William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)	
07-015	24769 square metres of arable land and hedgerow (south of Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)	
07-016	8641 square metres of arable land, hedgerow and private track (north of Church Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-016 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-017	194 square metres of arable land (north of Church Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-018	29 square metres of agricultural land (north of Church Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-018 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-019	452 square metres of hedgerow and public highway (Church Lane) (North Norfolk District Council)	Unknown	
09-001	892 square metres of public highway and verges (Northfield Lane) (North Norfolk District Council)	Unknown	
09-002	519 square metres of arable land and public footpath (Plumstead FP11) (south east of Northfield Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-003	35596 square metres of arable land and public footpath (Plumstead FP11) (south east of Northfield Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
09-006	777 square metres of public highway and verges (Matlaske Road) (North Norfolk District Council)	Unknown	
09-007	196 square metres of arable land (south of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-008	2035 square metres of arable land (south of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
09-009	41722 square metres of arable land and hedgerow (south of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-001	31524 square metres of arable land and hedgerow (Ferndale Farm, south west of The Street, A149) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-002	88 square metres of hedgerow (Ferndale Farm, south west of The Street, A149) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-003	550 square metres of arable land (Ferndale Farm, north of Little Barningham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-004	347 square metres of public highway and verges (Little Barningham Road) (North Norfolk District Council)	Unknown	
10-005	71 square metres of verge (south west of Little Barningham Road) (North Norfolk District Council)	Unknown	
10-006	685 square metres of arable land (Ferndale Farm, south of Little Barningham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-006 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-007	19814 square metres of arable land and hedgerow (Ferndale Farm, south of Little Barningham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-008	400 square metres of public highway and verge (Sweetbriar Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-008 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables) Unknown	
10-009	1104 square metres of arable land and access track (south of Sweetbriar Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Arthur Thomas Steel Flat 29 Charleville Mansions Charleville Road LONDON W14 9JA (in respect of rights granted) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Christopher Edward Steel 56 Nasmyth Street LONDON W6 0HB (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-009 cont'd		Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of legal easements reserved) Saltcarr Farms Limited Saltcarr Flouse Bure Valley Lakes Outton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) The Forestry Commission 620 Bristol Business Park Coldnarbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-009 cont'd		Unknown (in respect of rights of way, wayleaves, drainage and watercourse easements) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of legal easements reserved)	
10-010	26850 square metres of arable land and hedgerow (south of Sweetbriar Lane and west of Matlaske Road) (excluding all interests of the Crown) (North Norfolk District Council)	Arthur Thomas Steel Flat 29 Charleville Mansions Charleville Road LONDON W14 9JA (in respect of rights granted) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Christopher Edward Steel 56 Nasmyth Street LONDON W6 0HB (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-010 cont'd		Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of legal easements reserved) Saltcarr Farms Limited Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-010 cont'd		Unknown (in respect of rights of way, wayleaves, drainage and watercourse easements) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of legal easements reserved)	
10-012	462 square metres of verge and public highway (Matlaske Road) (North Norfolk District Council)	Unknown	
10-013	10291 square metres of arable land and hedgerow (Ferndale Farm, Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-014	53753 square metres of arable land, hedgerow and public footpath (Itteringham FP3) (south east of Matlaske Road) (North Norfolk District Council)	Centrica PLC Milistream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline) Mere Farm (Mannington) Limited Sankence Farm Sankence Farm Sankence A Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Outton NORWICH Norfolk NR11 6NW (in respect of shooting rights)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-014 cont'd		Unknown (in respect of rights reserved and granted)	
11-001	12 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline) Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe) Michael John Chantler Compass House Lypiatt Road CHELTENHAM GL50 2QJ (in respect of rights reserved) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N SEH (in respect of right of access to lay and maintain gas pipeline)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-001 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Outton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved and granted)	
11-002	1298 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline) Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-002 cont'd		Michael John Chantler Compass House Lypiatt Road CHELTENHAM GL50 2QJ (in respect of rights reserved) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved and granted)	
11-004	672 square metres of public highway and verge (Matlaske Road) (North Norfolk District Council)	Unknown	
11-006	28631 square metres of arable land and hedgerow (Mere north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of easement contained in deed dated 12 September 2011)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-006 cont'd		Nicholas Benjamin Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of rights stated in conveyance dated 5 November 1982) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Outon NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of easement contained in deed dated 12 September 2011) Seth Elton Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of right of access to conduct drainage maintenance stated in conveyance dated 5 November 1982)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-007	40407 square metres of arable land and hedgerow Mere Farm and Wood Farm, north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of easement contained in deed dated 12 September 2011) Nicholas Benjamin Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of rights stated in conveyance dated 5 November 1982) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH NORVICH NORVICH NORVICH NORVICH NORTofik NR11 16 NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of easement contained in deed dated 12 September 2011)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-007 cont'd		Seth Elton Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of right of access to conduct drainage maintenance stated in conveyance dated 5 November 1982)	
11-008	5466 square metres of arable land and hedgerow (north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NRT11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-001	865 square metres of arable land (north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-002	1592 square metres of public highway, verges and hedgerow (Matlaske Road) (North Norfolk District Council)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-002 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables) Unknown	
12-003	173 square metres of wooded land (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Saltcarr Farms Limited Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-004	8386 square metres of arable land and hedgerow (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-005	792 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-005 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-006	40552 square metres of arable land, hedgerow and public footpath (Corpusty FP17) (Wood Farm, south of Matlaske Road) (North Norfolk District Council)	Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting and sporting rights)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-006 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Sharon Brooks Wood Farm Saxthorpe Road Itteringham NORWICH NR11 7BG (in respect of private water supply)	
12-007	18286 square metres of arable land and public footpath (Corpusty FP16) (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-007 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) Unknown (in respect of restrictive covenants)	
13-001	4758 square metres of arable land and hedgerow (Hall Farm, Aylsham Road, B1354) (North Norfolk District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) (in respect of a right of pre-emption contained in an Agreement dated 15 October 2010)	
13-004	1112 square metres of verge and public highway (Blickling Road, B1354) (North Norfolk District Council)	Unknown	
13-007	27 square metres of public highway and verge (Aylsham Road, B1354) (Broadland District Council)	Unknown (in respect of rights reserved and granted)	
13-008	30 square metres of public highway and verge (Aylsham Road, B1354) (Broadland District Council)	Unknown (in respect of rights reserved and granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-009	2481 square metres of arable land and hedgerow (south of Aylsham Road, B1354) (Broadland District Council)	National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) Unknown (in respect of rights reserved and granted)	
13-010	56523 square metres of arable land and hedgerow (north of Spa Lane and south of Aylsham Road, B1354) (Broadland District Council)	National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) Unknown (in respect of rights reserved and granted)	
13-011	194 square metres of arable land (north of Spa Lane and south of Aylsham Road, B1354) (Broadland District Council)	National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) Unknown (in respect of rights reserved and granted)	
13-012	6 square metres of arable land (north of Spa Lane) (Broadland District Council)	Unknown	
13-013	694 square metres of verges and public highway (Spa Lane) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-014	879 square metres of arable land (Irmingland Hall Estate, south of Spa Lane) (Broadland District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline and restrictive covenants)	
13-016	32735 square metres of arable land and hedgerow (Irmingland Hall Estate, south of Spa Lane) (Broadland District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline and restrictive covenants)	
14-003	797 square metres of verge, hedgerow and public highway (Spink's Lane) (Broadland District Council)	Unknown	
14-004	19 square metres of verge and public highway (Spink's Lane) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
14-005	29 square metres of public highway (Spink's Lane) (Broadland District Council)	Unknown	
15-002	577 square metres of verge mad public highway (B1149) (Broadland District Council)	Unknown	
15-003	2295 square metres of arable land (The Grange, north of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right to maintain and use service media and right of entry)	
15-004	87228 square metres of arable land and hedgerow (The Grange, north of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right to maintain and use service media and right of entry)	
16-001	957 square metres of verge and public highway (Heydon Road) (Broadland District Council)	Unknown	
16-002	1 square metres of verge (Heydon Road) (Broadland District Council)	Unknown	
16-003	23468 square metres of arable land and hedgerow (north west of Beerhouse Farm, north east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-003 conťď		Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)	
		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media)	
		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of rights granted by Transfer dated 7th April 2009)	
16-004	912 square metres of arable land and hedgerow (north east of Beerhouse Farm, east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media)	

Page 921 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-004 cont'd		Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)	
		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media)	
		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of rights granted by Transfer dated 7th April 2009)	
16-005	838 square metres of wooded area and public highway (B1149 and Heydon Road) (Broadland District Council)	Unknown	
16-006	1390 square metres of verge and private highway (east of B1149) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-007	158 square metres of arable land and hedgerow (north east of Beerhouse Farm, B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)	
		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-008	311 square metres of arable land (east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-008 cont'd		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right reserved by deed dated 30 December 1969)	
16-009	55342 square metres of arable land and hedgerow (east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
16-009 cont'd		E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)
		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-009 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right reserved by deed dated 30 December 1969)	
16-010	1847 square metres of arable land and hedgerow (north east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-010 cont'd		Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	
		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	
		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media)	
		Unknown (in respect of right reserved by deed dated 30 December 1969)	
16-011	890 square metres of verge and public highway (The Street) (Broadland District Council)	Unknown	
16-013	232 square metres of verge, hedgerow and public highway (The Street) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-014	259 square metres of verge, hedge row and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr Farms Limited Saltcarr Fause Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-014 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987) Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)	
16-015	502 square metres of hedgerow and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-015 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Outon NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987) Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)	
16-017	258 square metres of verge, arable land and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-017 cont'd		Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 602 (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Outon NORWICH NORTOR NORTOR Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) The Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987) Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-001	2108 square metres of tree and hedge line (east of Marriott Way, B1149) (Broadland District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants)	
17-003	87 square metres of hedgerow (east of B1149) (Broadland District Council)	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of rights granted)	
17-005	1726 square metres of public highway and verge (B1149) (Broadland District Council)	Unknown	
17-009	560 square metres of verge, hedgerow and public highway (Birds Lane) (Broadland District Council)	Unknown	
17-012	51650 square metres of arable land and hedgerow (west of Holt Road, B1149) (Broadland District Council)	Unknown (in respect of a right to enter the land and maintain and repair the water pipes in respect of rights reserved)	
18-001	27689 square metres of arable land and hedgerow (north of Aylsham Road, B1145) (Broadland District Council)	Matthew Paul Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-001 cont'd		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of rights granted) Pauline Susan Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)	
18-002	252 square metres of arable land and verge (north of Aylsham Road, B1145) (Broadland District Council)	Matthew Paul Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights) Pauline Susan Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)	
18-003	813 square metres of verges and public highway (B1145) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-007	515 square metres of hedgerow and public highway (Old Friendship Lane) (Broadland District Council)	Unknown	
18-008	217 square metres of arable land and hedgerow (south west of Old Friendship Lane) (Broadland District Council)	The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	
18-009	20048 square metres of arable land and hedgerow(south of Old Friendship Lane) (Broadland District Council)	The Diocess Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	
18-011	1 square metres of arable land (north of Norwich Road) (Broadland District Council)	The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-013	450 square metres of arable land and hedgerow (north of Norwich Road) (Broadland District Council)	The Diocess Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	
18-014	547 square metres of arable land and public footpath (Cawton FP7 and FP8) (north of Norwich Road) (Broadland District Council)	Unknown	
18-015	9454 square metres of arable land (north east of Norwich Road) (Broadland District Council)	Unknown (in respect of a right to enter the land and maintain and repair the water pipes in respect of rights reserved)	
18-016	998 square metres of verge and public highway (Norwich Road) (Broadland District Council)	Unknown	
19-003	507 square metres of arable land and hedgerow (north of Reepham Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB (in respect of rights of access)	
19-004	1535 square metres of Section of verge, hedgerow and public highway (Reepham Road) (Broadland District Council)	Unknown	
19-005	209 square metres of hedgerow (Reepham Road) (Broadland District Council)	Unknown	
19-009	15 square metres of public highway (Unnamed Road, south of Reepham Road) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-010	24898 square metres of arable land and hedgerow (north of Church Lane) (Broadland District Council)	P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)	
20-001	465 square metres of Section of verge, hedgerow and public highway (Church Lane) (Broadland District Council)	Unknown	
20-002	266 square metres of arable land and hedgerow (south of Church Lane) (Broadland District Council)	P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)	
20-003	75441 square metres of arable land and hedgerow (south of Church Lane) (Broadland District Council)	P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)	
21-001	31318 square metres of arable land and hedgerow (west of Clay Lane) (Broadland District Council)	Unknown (in respect of right of way restrictive covenants)	
21-002	584 square metres of verge, hedgerow and public highway (Clay Lane) (Broadland District Council)	Unknown	
21-005	8955 square metres of wooded area (east of Clay Lane) (Broadland District Council)	Unknown (in respect of shooting rights)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-007	3032 square metres of arable land and tree line (north east of Church Lane) (Broadland District Council)	Unknown (in respect of right of way restrictive covenants)	
21-009	75 square metres of wooded land and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Unknown (in respect of right of way restrictive covenants)	
21-010	5780 square metres of wooded land, arable land and hedgerow and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Unknown (in respect of right of way stated in Conveyance dated 03 April 1965)	
21-012	1392 square metres of hedgerow, private road (Church Lane) and public footpath (Swannington FP6) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-012 cont'd		Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN (in respect of right of way over access track) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of way stated in Conveyance dated 03 April 1965)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-013	43782 square metres of arable land, private road (Church Lane), hedgerow and public footpath (Swannington FP6 and Swannington FP7) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-013 cont'd		Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track)	
21-015	197 square metres of private road and public footpath (Swannington FP5) (Woodlands Farm, north of Church Lane) (Broadland District Council)	Unknown (in respect of rights of access reserved)	
21-016	669 square metres of private road and public footpath (Swannington FP5) (north east of Church Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-017	183 square metres of private road, farm courtyard and public footpath (Swannington FP5) (Woodlands Farm, Church Lane) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passage of services granted) Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passage of services granted) Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passage of services granted) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of way over access track)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-018	2025 square metres of arable land, private road (Church Lane) and public footpath (Swannington FP5 and FP6) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-018 cont'd		Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track)	
21-019	142 square metres of hedgerow, private road (Church Lane) and public footpath (Swannington FP6) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
21-019 cont'd		George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of access and right of passage of services through conducting media)	
22-001	337 square metres of Section of watercourse (Trout Stream) (south west of Church Lane) (Broadland District Council)	Unknown	
22-003	982 square metres of hedgerow and public highway (Upgate Road) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
22-005	19 square metres of verge, public highway (Upgate Road) and public footpath (Swannington FP8) (Broadland District Council)	Unknown	
22-006	29 square metres of public highway (Upgate Road) (Broadland District Council)	Unknown	
22-007	316 square metres of verge, arable land (west of Upgate Road) and public footpath (Swannington FP13) (Broadland District Council)	Unknown (in respect of right of access and right of light and air and to lay water pipes)	
22-008	39526 square metres of arable land and hedgerow and public footpath (Swannington FP13) and restricted public byway (Swannington RB12) (south of Upgate Road) (Broadland District Council)	Unknown (in respect of right of access and right of light and air and to lay water pipes)	
22-009	28761 square metres of arable land and hedgerow (north of Reepham Road) (Broadland District Council)	Graham Martin Dacre Drayton Hall Hall Lane Drayton NORWICH Norfolk NR8 6DP (in respect of personal covenants contained in transfer dated 3 April 2018) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to lay and maintain service media)	
22-010	1259 square metres of verge and public highway (Reepham Road) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
22-011	38380 square metres of arable land and hedgerow (south of Reepham Road) and restricted byway (Attlebridge RB1) (Broadland District Council)	Graham Martin Dacre Drayton Hall Hall Lane Drayton NORWICH Norfolk NR8 6DP (in respect of personal covenants contained in transfer dated 3 April 2018) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to lay and maintain service media)	
23-001	1359 square metres of public highway (Marriott's Way) (Broadland District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of disused railway) Unknown	
23-003	31610 square metres of arable land and hedgerow (south of Marriott's Lane and north of Felthorpe Road) (Broadland District Council)	Unknown (in respect of right of way)	
23-004	445 square metres of verge and public highway (Felthorpe Road) (Broadland District Council)	Unknown	
23-005	585 square metres of arable land and hedgerow (north of Felthorpe Road) (Broadland District Council)	Unknown (in respect of right of way)	
23-006	2 square metres of verge (Felthorpe Road) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-007	22093 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right of way)	
23-008	30747 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right of way)	
23-009	258 square metres of verge and public highway (Old Fakenham Road) (Broadland District Council)	Unknown	
23-010	180 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right to take water from pump and well and restrictive covenants)	
23-011	108 square metres of hedgerow (east of Fakenham, A1067) (Broadland District Council)	Unknown	
23-012	1300 square metres of verge and public highway (Old Fakenham Road) (Broadland District Council)	Unknown	
23-013	6142 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right to take water from pump and well and restrictive covenants)	
23-014	3726 square metres of verge and public highway (Fakenham Road, A1067) (Broadland District Council)	Unknown	
23-015	113 square metres of verge and public highway (Ivy Cottages, south west of Fakenham Road, A1067) (Broadland District Council)	Unknown	
23-016	724 square metres of private highway and arable land (south west of Fakenham Road, A1067) (Broadland District Council)	Keith Robert Miller 1 Ivy Cottages Fakenham Road ATTLEBRIDGE Norfolk NR9 5TG (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-016 cont'd		Simon James Cram 2 Ivy Cottages Fakenham Road ATTLEBRIDGE Norfolk NR9 5TG (in respect of right of way) Unknown (in respect of right of way)	
23-017	20334 square metres of arable land and hedgerow (south west of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right of way)	
23-018	622 square metres of watercourse (River Wensum) (south west of Fakenham Road, A1067) (Broadland District Council)	Unknown	
23-022	989 square metres of arable land and private road (south of Fakenham Road, A1067) (Broadland District Council)	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access) Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-023	946 square metres of private road (south of Fakenham Road, A1067) (Broadland District Council)	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access) Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (in respect of right of access) Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5JT (in respect of right of access) Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access) William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ (in respect of right of access)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-028	165 square metres of public highway (Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (in respect of personal covenants relating to erection of boundary fence) Unknown (in respect of rights reserved)	
24-002	19 square metres of access track and verge (to the east of Ringland Lane) (Broadland District Council)	Unknown	
24-003	106 square metres of verge and public highway (Ringland Lane) (Broadland District Council)	Unknown	
24-005	821 square metres of public highway (Ringland Lane) (Broadland District Council)	Unknown	
24-006	608 square metres of arable land and private access road (west of Ringland Lane) (Broadland District Council)	Unknown (in respect of rights reserved by conveyance dated 6 January 1983)	
24-007	103393 square metres of hedgerow and arable land (north east of Church Hill Lane) (Broadland District Council)	Unknown (in respect of rights reserved by conveyance dated 6 January 1983)	
25-001	891 square metres of ditch and thicket (east of Church Hill Lane) (Broadland District Council)	Unknown	
25-002	4 square metres of private road (east of Church Hill Lane) (Broadland District Council)	Unknown	
25-003	85 square metres of verge and public highway (Church Hill Lane) (Broadland District Council)	Unknown	
25-004	16 square metres of verge (east of Church Hill Lane) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
25-005	1524 square metres of arable land (east of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
25-006	61433 square metres of hedgerow, arable land and verge (east of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)	
25-007	57 square metres of verge (south of Church Hill Lane) (Broadland District Council)	Unknown	
25-008	539 square metres of verge and public highway (Church Hill Lane) (Broadland District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
25-009	1111 square metres of private road and arable land (north of Weston Road) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)	
25-010	57 square metres of verge and public highway (Weston Road) (Broadland District Council)	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of right of access) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
25-011	16 square metres of public highway (Church Hill Lane), verge and public footpath (Weston Longville FP9) (Broadland District Council)	Unknown	
25-013	5 square metres of verge and public footpath (Weston Longville FP9) (South of Church Hill Lane) (Broadland District Council)	Unknown	
25-015	219 square metres of verge and public footpath (Weston Longville FP9) (south of Church Hill Lane) (Broadland District Council)	Unknown	
25-016	1101 square metres of hedgerow and arable land (south of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SCD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of right of access to lay and maintain gas main and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
25-017	47124 square metres of arable land (south of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of right of access to lay and maintain gas main and restrictive covenants)	
26-001	208 square metres of public highway (The Broadway) (Broadland District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
26-001 cont'd		National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)	
26-002	654 square metres of woodland, verge and public highway (The Broadway) (Broadland District Council)	Unknown	
26-003	83 square metres of verge and public highway (The Broadway) (Broadland District Council)	Unknown	
26-004	13 square metres of woodland and verge (The Broadway) (Broadland District Council)	Unknown	
26-005	66 square metres of woodland and verge (The Broadway) (Broadland District Council)	Unknown	
26-006	380 square metres of hedgerow and woodland (The Broadway) (Broadland District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
26-006 cont'd		Unknown (in respect of right of way and shooting rights)	
26-007	111007 square metres of hedgerow, arable land, woodland, private road, verge and public highway (The Broadway) (Broadland District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of right of access to lay and maintain gas main and restrictive covenants)	
26-008	937 square metres of arable land and verge (west of Taverham Road) (Broadland District Council)	Matthew John Rampton Kingfisher House 1 Gilders Way NORWICH Norfolk NR3 1UB (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right of way and shooting rights)	
27-001	70 square metres of public highway (Taverham Road) (Broadland District Council)	Unknown	
27-002	1835 square metres of arable land and private road (east of Taverham Road) (Broadland District Council)	Unknown (in respect of right of access and right of passage of services through conducting media)	

Page 958 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
27-003	806 square metres of hedgerow and public highway (Taverham Road) (Broadland District Council)	Unknown	
27-004	64154 square metres of arable land, hedgerow and private road (east of Taverham Road) (Broadland District Council)	Unknown (in respect of right of access and right of passage of services through conducting media)	
27-005	36546 square metres of watercourse (River Tud), arable land and hedgerow (north of the A47) (Broadland District Council)	Unknown	
27-006	2245 square metres of arable land and hedgerow (north of A47) (Broadland District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of right of access to lay and maintain apparatus)	
28-001	20155 square metres of arable land and hedgerow (north of the A47) (Broadland District Council)	Unknown (in respect of right of access and right of passage of services through conducting media)	
28-002	845 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-003	901 square metres of arable land and access track (north of the A47) (South Norfolk District Council)	Unknown (in respect of right of access and right of passage of services through conducting media)	
28-004	121 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-005	3 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown (in respect of restrictive covenants and rentcharges as may have been imposed before 12 February 2009)	
28-006	298 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown	
28-007	14 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-008	589 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown (in respect of restrictive covenants and rentcharges as may have been imposed before 12 February 2009)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-009	1231 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown	
28-010	649 square metres of public highway, lay by and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-011	49638 square metres of arable land, woodland, hedgerow and verges (south of the A47 and north of Church Lane) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain) Unknown (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-012	NUMBER NOT IN USE	NUMBER NOT IN USE	
28-013	NUMBER NOT IN USE	NUMBER NOT IN USE	
28-014	243 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain and in respect of right to maintain woodland path) Unknown (in respect of right of access and right of passage of services through conducting media)	
28-015	403 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-021	358 square metres of public highway and verges (Broom Lane) (South Norfolk District Council)	Unknown	
28-023	1 square metre of arable land and access track (north of Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road Norwich NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021) Norfolk County Council County Hall Martineau Lane Norwich NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)	
28-023 cont'd		Norwich City College of Further and Higher Education c/o Martin Colbourne lpswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016) Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX	

Page 963 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		(in respect of rights granted) Condimentum Limited Food Enterprise Park Colmans Close Easton Noriwch NR9 5FG (in respect of rights granted)	
28-023 cont'd		Eurotek Engineering Limited Centrum Noriwch Research Park Colney Norwich NR4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-024	2 square metres of arable land and access track (north of Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road Norwich NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021) Norfolk County Council County Hall Martineau Lane Norwich NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain) Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-024 cont'd		Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Condimentum Limited Food Enterprise Park Colmans Close Easton Norwich NR9 5FG (in respect of rights granted) Eurotek Engineering Limited Centrum Noriwch Research Park Colney Norwich NR4 7UG (in respect of rights granted)	
28-024 cont'd		Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited	

Page 966 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	
28-025	123 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Unknown	
28-026	6041 square metres of arable land and private road (Coleman Road) (north of Church Lane) (South Norfolk District Council) 483 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)	
28-026 cont'd		Unknown (in respect of right of access and right of passage of services through conducting media) Condimentum Limited Food Enterprise Park Colmans Close Easton NORWICH	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		NR9 5FG (in respect of rights granted) Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Eurotek Engineering Limited Centrum Norwich Research Park Colney Lane Colney Norwich NE4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NE4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX	
28-026 cont'd		(in respect of rights granted) Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	

Page 968 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-027	483 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021) Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)	
28-027 cont'd		Condimentum Limited Food Enterprise Park Colmans Close Easton NORWICH NR9 5FG (in respect of rights granted) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain) Norwich City College of Further and Higher Education	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016) Unknown (in respect of right of access and right of passage of services through conducting media) Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)	
28-027 cont'd		Eurotek Engineering Limited Centrum Norwich Research Park Colney Lane Colney Norwich NE4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	
29-001	2802 square metres of access track and verges (east of Colton Road) (South Norfolk District Council)	Annette Lucie Alston Honingham Thorpe Farm Norwich Road Colton NORWICH NR9 5BZ (in respect of right of access)	
29-001 cont'd		Ian James Alston Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)	
		William Young (Dereham) Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)	
29-002	95 square metres of public highway and verges (Colton Road) (South Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
29-003	60617 square metres of arable land, woodland, hedgerow and track (east of Colton Road) (South Norfolk District Council)	Annette Lucie Alston Honingham Thorpe Farm Norwich Road Colton NORWICH NR9 5BZ (in respect of right of access)	
29-003 cont'd		Ian James Alston Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access) William Young (Dereham) Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)	
29-004	8190 square metres of arable land and hedgerows (east of Colton Road) (South Norfolk District Council)	Unknown (in respect of rights reserved)	
29-005	685 square metres of public highway and verges (Colton Road) (South Norfolk District Council)	Unknown	
29-007	3726 square metres of arable land and hedgerows (west of Colton Road) (South Norfolk District Council)	Unknown (in respect of rights reserved)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
29-008	43135 square metres of arable land, grassland, woodland and hedgerows (west of Chapel Street) (South Norfolk District Council)	Colin Herbert Bothway Cherry Tree Farm Salle NORWICH Norfolk NR10 4SF (in respect of right of entry to extract mines and minerals, timber rights, positive covenants and restrictive covenants) Unknown (in respect of light and air)	
30-001	608 square metres of watercourse (River Yare) (South Norfolk District Council)	Unknown	
30-007	7 square metres of private access road and verge (north of Chapel Street) (South Norfolk District Council)	Unknown	
30-008	25 square metres of public highway (Chapel Street) (South Norfolk District Council)	Unknown	
30-009	18 square metres of verge (north of Chapel Street) (South Norfolk District Council)	Unknown	
30-010	959 square metres of public highway (Chapel Street) (South Norfolk District Council)	Unknown	
30-011	35447 square metres of arable land, hedgerows and tree line (south of Chapel Street) (South Norfolk District Council)	Unknown	
30-012	532 square metres of arable land and hedgerow (south of Chapel Street) (South Norfolk District Council)	Unknown	
30-013	1227 square metres of watercourse (River Tiffey) (north of Watton Road, B1108) (South Norfolk District Council)	Unknown	
30-014	9178 square metres of woodland (south of River Tiffey) (South Norfolk District Council)	Unknown (in respect of rights reserved)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
30-015	174 square metres of woodland (south of River Tiffey) (South Norfolk District Council)	Unknown	
30-016	693 square metres of woodland (south of the River Tiffey) (South Norfolk District Council)	Helga Erika Jalil 1A Gilbert Way Cringleford NORWICH Norfolk NR4 7RN (in respect of a right of access, right of passage and running of surface water and provision of light and air)	
31-001	1209 square metres of verge and public highway (Watton Road, B1108) (South Norfolk District Council)	Unknown	
31-003	702 square metres of private road (south of Watton Road, B1108) (South Norfolk District Council)	Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	
31-004	25870 square metres of private road, arable land and hedgerow (north of Burdock Lane) (South Norfolk District Council)	Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	
31-005	679 square metres of woodland (north of Burdock Lane) (South Norfolk District Council)	Unknown (in respect of rights reserved)	
31-006	2467 square metres of private road (north of Burdock Lane) (South Norfolk District Council)	Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	
31-007	600 square metres of verge and public highway (Burdock Lane) (South Norfolk District Council)	Unknown	
31-008	33 square metres of verge (south of Burdock Lane) (South Norfolk District Council)	Unknown	
31-010	14 square metres of public highway (Burdock Lane) (South Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
31-012	31953 square metres of arable land, drain, hedgerow and public footpath (Great Melton FP1) (west of Pockthorpe Road) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to use, lay and maintain electricity cables) Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	
32-002	13733 square metres of hedgerow and arable land (north of Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of right of way over access track)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
32-003	520 square metres of verge and public highway (Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Unknown	
32-004	110 square metres of verge and public highway (Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Unknown	
32-005	390 square metres of arable land and track (south of Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Susanne Eva Markham Old Boundary Barn Green Lane Wramplingham	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	
32-005 cont'd		Thomas Charles Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	
32-006	28012 square metres of arable land, ditch and hedgerows (south of Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Susanne Eva Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		Thomas Charles Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	
33-002	29 square metres of public highway (Pockthorpe Road) (South Norfolk District Council)	Unknown	
33-004	11 square metres of public highway (Melton Road) (South Norfolk District Council)	Unknown	
33-005	594 square metres of private road (west of Melton Road) (South Norfolk District Council)	Unknown	
33-006	1295 square metres of hedgerow (west of Melton Road) (South Norfolk District Council)	Unknown (in respect of restrictive covenants as may have been imposed before 16 September 2011)	
33-007	13970 square metres of arable land and hedgerow (west of Melton Road) (South Norfolk District Council)	Unknown (in respect of restrictive covenants as may have been imposed before 16 September 2011)	
33-010	826 square metres of public highway (Melton Road) (South Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-011	439 square metres of arable land and verge (east of Melton Road) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	
33-011 cont'd		Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-012	23630 square metres of arable land, hedgerow and verge (east of Melton Road) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	
33-012 cont'd		Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-013	820 square metres of arable land (east of Melton Road) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Persimmon Homes Limited Persimmon House Fulford YORK North Yorkshire YO19 4FE (in respect of rights granted)	
33-013 cont'd		Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-017	28370 square metres of arable land, drain and hedgerow (north of Ketts Oak, B1172) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of right of way over access track) Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU (in respect of right of passage of services through conducting media)	
34-001	19636 square metres of arable land and hedgerow (north of Ketts Oaks, B1172) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right of access to lay and maintain apparatus)	
34-003	2192 square metres of arable land (north of Ketts Oak, B1172) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right of access to lay and maintain apparatus)	

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-004	117 square metres of public highway (Ketts Oaks, B1172) (South Norfolk District Council)	Unknown	
34-005	139 square metres of public highway (Ketts Oaks, B1172) (South Norfolk District Council)	Unknown	
34-006	2059 square metres of public highway (Ketts Oaks, B1172) and verge (South Norfolk District Council)	Unknown	
34-010	50248 square metres of arable land and hedgerow (north of the A11) (South Norfolk District Council)	Unknown	
35-002	4801 square metres of public highway (A11), verge and hedgerows (west of Ketteringham Lane) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights of way over the land and maintenance of access ways and bridges)	
35-003	2626 square metres of rail line and verge (south of the A11) (South Norfolk District Council)	Unknown	
35-004	435 square metres of private road (west of Ketteringham Lane and south of A11) (South Norfolk District Council)	Unknown (in respect of rights granted as contained in a Deed dated 27th June 1983)	
35-005	34 square metres of private road (west of Ketteringham Lane) (South Norfolk District Council)	Chris Thrower Forgestone House Low Street Ketteringham WYMONDHAM NR18 9RY (in respect of rights of access and to draw water from an ancient well) Unknown	
35-006	40884 square metres of Land and buildings on the north side of arable land, private road and verge (west of High Street) (South Norfolk District Council)	Unknown (in respect of rights granted as contained in a Deed dated 27th June 1983)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
35-007	1076 square metres of public highway (High Street) (South Norfolk District Council)	Unknown	
35-008	87 square metres public highway (High Street) (South Norfolk District Council)	Unknown	
35-009	308 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948) Unknown (in respect of right to use drainage system, right to enter and right to lay and maintain water mains and pipes stated in conveyance dated 10th June 1959)	
35-010	24020 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948) Unknown (in respect of right to use drainage system, right to enter and right to lay and maintain water mains and pipes stated in conveyance dated 10th June 1959)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
35-011	107807 square metres of private road (Church Road), woodland, hedgerow, verge and public footpath (Ketteringham FP4) (excluding all interests of the Crown) (South Norfolk District Council)	Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Sharon Gowing Park Lodge Ketteringham Park Ketteringham WYMONDHAM NR18 9PF (in respect of right of way over access track and drainage rights as contained in a Conveyance dated 5th January 1959) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)
36-001	4970 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
		Sharon Gowing Park Lodge Ketteringham Park Ketteringham WYMONDHAM NR18 9PF (in respect of right of way over access track and drainage rights as contained in a Conveyance dated 5th January 1959) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)	
36-002	226 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Unknown	
36-003	266 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Unknown	
36-004	1002 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Unknown	
36-005	7623 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
36-006	56249 square metres of verge, arable land and private road (east of Hethersett Road) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of gas main and right of way) Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)
36-007	4812 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipes)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights of access to lay and maintain electricity cables and restrictive covenants) Unknown (in respect of right of access and timber rights) Unknown (in respect of rights reserved)
36-008	254 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Unknown
36-009	19618 square metres of hedgerow and arable land (east of Hethersett Road) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights of access to lay and maintain electricity cables and restrictive covenants) Unknown (in respect of right of access and timber rights)
36-009 cont'd		Unknown (in respect of rights reserved)

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
36-011	55959 square metres of arable land, hedgerows, verge and public footpath (East Carleton FP4) (west of Intwood Lane) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	
37-001	903 square metres of arable land and public footpath (East Carleton FP4) (west of Intwood Lane) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	
37-002	808 square metres of verge and public highway (Intwood Lane) (South Norfolk District Council)	Unknown	
37-003	553 square metres of arable land (east of Intwood Lane) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	
37-004	25 square metres verge (south of Intwood Lane) (South Norfolk District Council)	Unknown	

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
37-005	27480 square metres of hedgerow, arable land and verge (east of Intwood Lane) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	
37-006	82917 square metres of public footpath (Swardeston FP2, Swardeston FP3 and Swardeston FP4), hedgerow, watercourse (unnamed) and arable land (west of Norwich Road, B1113) (South Norfolk District Council)	Unknown (in respect of rights as contained in a Conveyance dated 21st October 1902)	
38-001	276 square metres of access track, verge and arable land (west of Main Road, B1113) (South Norfolk District Council)	Unknown (in respect of rights as contained in a Conveyance dated 21st October 1902)	
38-002	1135 square metres of verge and public highway (Main Road, B1113) (South Norfolk District Council)	Unknown	
38-009	83 square metres of public highway (Swardeston Lane and Gowthorpe Lane) and verge (South Norfolk District Council)	Unknown	
38-010	17 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Unknown	
38-011	2 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
38-011 cont'd		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
38-012	42 square metres of private road (east of Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
38-013	1149 square metres of public footpath (Swainsthorpe FP1), verge and arable land (east of Swardeston Lane) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
38-014	604 square metres of public highway (Swardeston Lane) and verge (South Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
38-015	6 square metres of verge (east of Swardeston Lane) (South Norfolk District Council)	Unknown	
38-016	61 square metres of verge (east of Swardeston Lane) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
38-017	22746 square metres of arable land (north of Church Road) and public footpath (Swainsthorpe FP1) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-001	19624 square metres of arable land (south of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-002	NUMBER NOT IN USE	NUMBER NOT IN USE	
39-003	NUMBER NOT IN USE	NUMBER NOT IN USE	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-004	NUMBER NOT IN USE	NUMBER NOT IN USE	
39-005	4101 square metres of arable land and verge (south of Hickling Lane and west of Ipswich, A140) (South Norfolk District Council)	Unknown National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights granted)	
39-006	8397 square metres of arable land and verge (south of Hickling Lane and west of Ipswich, A140) (South Norfolk District Council)	Unknown National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights granted)	
39-007	NUMBER NOT IN USE	NUMBER NOT IN USE	
39-008	308 square metres of verge, hedgerow and public byway open to all traffic (Swainsthorpe BOAT6) (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-009	NUMBER NOT IN USE	NUMBER NOT IN USE	
39-010	1 square metres of land and public byway open to all traffic (Swainsthorpe BOAT6) (Hickling Lane) (South Norfolk District Council)	Unknown	
39-011	455 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-012	464 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-013	67 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-014	112837 square metres of arable land and part of public byway open to all traffic (Swainsthorpe BOAT6) (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Unknown (in respect of right of way over Hickling Lane reserved)	
39-015	52057 square metres of arable land and hedgerow (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-016	936 square metres of hedgerow and arable land (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-016 cont'd		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-017	1005 square metres of hedgerow and arable land (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-018	5791 square metres of arable land (north of Hickling Lane) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-018 cont'd		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-019	83657 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-020	1227 square metres of arable land and hedgerow (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-020 cont'd		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-021	5569 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-022	5093 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-022 cont'd		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-023	3018 square metres of woodland (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-025	15 square metres of public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-025 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track)	
39-026	56 square metres of public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of a right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-026 cont'd		Unknown	
39-027	24 square metres of public bridleway (Stoke Holy Cross BR3) and hedgerow (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of a right of way) Unknown	
39-028	5153 square metres of woodland, hedgerow and public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
39-028 cont'd		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacook CHIPPENHAM SN15 2PP (in respect of right of way over access track)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-029	9833 square metres of woodland, hedgerow and public bridleways (Stoke Holy Cross BR3 and Swardeston BR12) (south of Norwich Main Substation and west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-029 cont'd		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track)	
39-030	203 square metres of public bridleway (Swardeston BR12) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-031	654 square metres of public bridleway (Swardeston BR12) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-032	1979 square metres of arable land and hedgerow (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-033	1925 square metres of arable land, hedgerow and public bridleway (Swardeston BR12) (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-034	40840 square metres of arable land (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millistream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-035	15126 square metres of arable land and hedgerow (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	
39-036	5908 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of right to maintain gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
39-036 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) Unknown (in respect of right granted, reserved and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-037	25343 square metres of Norwich Main Substation and private road (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	
39-038	4098 square metres of verge, hedgerow and public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-038 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-039	2117 square metres of hedgerow and verge (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	
39-040	634 square metres of hedgerow and verge (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-041	7492 square metres of private road and grassland (Norwich Main Substation and west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	
39-042	3223 square metres of private road and verge (west of Ipswich Road, A140 and south of Mangreen Road) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-043	14336 square metres of verge, grassland and hedgerows (Norwich Main Substation, west of Ipswich Road, A140 and south of Mangreen Road) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	
39-044	2901 square metres of private road (west of Ipswich Road and south of Mangreen Road) (South Norfolk District Council)	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN (in respect of right of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-044 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	
40-001	16719 square metres of grassland (west of Ipswich Road, A140) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of right to maintain gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
40-001 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)	
40-002	5925 square metres of public highway (Ipswich Road, A140 and Mangreen Road) (South Norfolk District Council)	Unknown	
40-003	126 square metres of verge (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
40-004	19 square metres of private road (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
40-005	12 square metres of verge (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
40-006	27 square metres of Mangreen Hall Farm, Swardeston, Norwich, NR14 8DD (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media)	

Page 1014 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
40-006 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)	
40-007	161 square metres of verge and public highway (Mangreen Road) (South Norfolk District Council)	Unknown	
40-008	5875 square metres of hedgerow and verge (Mangreen Hall Farm, west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
40-009	219 square metres of verge and private road (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Unknown (in respect of rights granted, reserved and restrictive covenants)	
40-010	353 square metres of public highway (Ipswich Road, A140) (South Norfolk District Council)	Unknown	
N/A	48 Chapel Street, Barford, Norwich, NR9 4AB	Lauren Kate Dingle 48 Chapel Street Barford NORWICH Norfolk NR9 4AB	
N/A	Gralie, Burdock Lane, Barford, Norwich, NR9 4BL	Russell Philip Marling Gralie Burdock Lane Barford NORWICH Norfolk NR9 4BL	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Sarah Melissa Marling Gralie Burdock Lane Barford NORWICH Norfolk NR9 4BL	
N/A	Turnpike Farm Cottage, Burdock Lane, Barford, NR9 4BL	Karen Clarke Turnpike Farm Cottage Burdock Lane Barford NORWICH Norfolk NR9 4BL	
		Paul Thomas Clarke Turnpike Farm Cottage Burdock Lane Barford NORWICH Norfolk NR9 4BL	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Apple Lodge, Chapel Street, Barford, Norwich, NR9 4BJ	Kenneth Edwin Palmer Apple Lodge Chapel Street Barford NORWICH NR9 4BJ Susan Margaret Palmer Apple Lodge Chapel Street Barford NORWICH NR9 4BJ	
N/A	Mill Cottage, Colton Road, Marlingford, Norwich, NR9 5HS	Edward Patrick Morris Mill Cottage Colton Road Marlingford NORWICH Norfolk NR9 5HS Mary-Anne Christine Morris Mill Cottage Colton Road Marlingford NORWICH Norfolk NR9 5HS	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	4 Colton Road, Marlingford, Norwich, NR9 5HS	Charles Alistair Brearley 4 Colton Road Cottages Marlingford NORWICH Norfolk NR9 5HS	
		Jacqueline James 4 Colton Road Cottages Marlingford NORWICH Norfolk NR9 5HS	
N/A	3 Colton Road, Marlingford, Norwich, NR9 5HS	Edward Dunstan Smith 3 Colton Road Marlingford NORWICH NR9 5HS	
		Sally Marie Toms 3 Colton Road Marlingford NORWICH NR9 5HS	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	2 Colton Road, Marlingford, Norwich, NR9 5HS	Julian Pierce Blackmore 2 Colton Road Marlingford NORWICH Norfolk NR9 5HS	
N/A	Farm View, Old Fakenham Road, Attlebridge, Norwich, NR9 5ST	Heather Jane Harris Farm View Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST	
		Peter David Harris Farm View Old Fakenham Road ATTLEBRIDGE Norfolk NR9 5ST	
N/A	Petersfield, Felthorpe Road, Attlebridge, NR9 5SU	Laurence Philip Tanner-Ashby Petersfield Felthorpe Road ATTLEBRIDGE Norfolk NR9 5SU	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Bridge Barn, Spinks Lane, Heydon, Norwich NR11 6RF	Andrew Richard Marriott Bridge Barn Spinks Lane Heydon NORWICH NR11 6RF	
N/A	The Oaks, Reepham Road, Attlebridge, NR9 5TA	Emma Louise Savage The Oaks Reepham Road ATTLEBRIDGE Norfolk NR9 5TA Timothy Michael Savage The Oaks Reepham Road ATTLEBRIDGE Norfolk NR9 5TA	
N/A	Bodham House, Rectory Road, Lower Bodham, Holt, Norwich, NR25 6PR	Benjamin Stuart Shrive 2 St. Martins Castle Bytham GRANTHAM Lincolnshire NG33 4RH	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Aspenvale, Aylsham Road, Cawston, Norwich, NR10 4HH	Richard Henry Bothway Howard Aspenvale Cawston NORWICH Norfolk NR10 4HH	
N/A	Weybourne House, The Street, Weybourne, Holt, NR25 7SY	Jonathan Richard Davey Hawkins Weybourne House The Street Weybourne HOLT NR25 7SY Philip David Candice Weybourne House The Street Weybourne HOLT NR25 7SY	
N/A	1 Red Tile Cottages, Holt Road, Cawston, Norwich, NR10 4HT	Jean Heather Hufton Red Tile Cottage Holt Road Cawston NORWICH NR10 4HT	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		William Hufton Red Tile Cottage Holt Road Cawston NORWICH NR10 4HT	
N/A	The Old Railway Gatehouse, The Street, Oulton, Norwich, NR11 6AF	Clive Charles Searson The Old Railway Gatehouse The Street Oulton NORWICH NR11 6AF Nicola June Tanner The Old Railway Gatehouse The Street Oulton NORWICH NR11 6AF	
N/A	Renfield, Weybourne, Holt, Norwich, NR25 7EH	Mary P Wells Renfield Weybourne HOLT NR25 7EH (as reputed owner)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Stanley T Wells Renfield Weybourne HOLT NR25 7EH (as reputed owner) Unknown	
N/A	1 and 1A Colton Road, Marlingford, Norwich, NR9 5HS	Adrian James Horner Holly Cottage 1 Colton Road Marlingford NORWICH Norfolk NR9 5HS (as reputed owner) Deborah S Horner Holly Cottage 1 Colton Road Marlingford NORWICH Norfolk NR9 5HS (as reputed owner) Katie J Horner 1A Colton Road Marlingford Norwich NR9 5HS (as reputed owner) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	High Fields, Weybourne, Holt, NR25 7EH	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH	
N/A	Field House, Norwich Road, Cawston, Norwich, NR10 4EU	Evelyn V Wright Field House Norwich Road Cawston NORWICH NR10 4EU (as reputed owner) Jonathan W Wright Field House Norwich Road Cawston NORWICH NR10 4EU (as reputed owner) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Turnpike Cottage North, Norwich Road, Barford, Norwich, NR9 4BQ	Kerry Allan 43 Queens Road BURY ST. EDMUNDS IP33 3EP (as reputed owner) Marie Christina Bridge Turnpike Cottage North Norwich Road Barford NORWICH Nofolk NR9 4BQ (as reputed owner) Michael Frank Bridge Turnpike Cottage North Norfolk NR9 4BQ (as reputed owner) Michael Frank Bridge Turnpike Cottage North Norwich Road Barford NORWICH Norfolk NR9 4BQ (as reputed owner) Tracey Bridge Via Pio La Torre 8B Pedara Catania Sicily 95030 (as reputed owner) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Hamer, The Street, Bodham, Holt, NR25 6NW	Moya Elisabeth Garrard Myerscough Hamer The Street Bodham HOLT Norfolk NR25 6NW Robin Nigel Myerscough Hamer The Street Bodham HOLT Norfolk NR25 6NW	
N/A	Walnut Barn, The Street, Bodham, Holt, NR25 6NW	Lemur UK LLC 10960 Wilshire Blvd 11th Floor LOS ANGELES CA 90024 UNITED STATES OF AMERICA	
N/A	Land and buildings at Bluestone Cottage, Heydon, Norwich, NR11 6RA	Norman Earnest Durrant Bluestone Cottage Heydon NORWICH NR11 6RA (as reputed owner) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Land on the west side of Colton Road, Marlingford, Norfolk, NR9	The Executor Of The Estate Of The Late Edward Christopher Evans-Lombe Cavell Barn The Common Swardeston NORWICH NR14 8DZ	
N/A	Greenacre, Holt Road, Cawston, Norwich, NR10 4HT	Gwendoline Iris Cockaday Greenacre Holt Road Cawston NORWICH NR10 4HT John Richard Cockaday Greenacre Holt Road Cawston NORWICH NR10 4HT	
N/A	Land at Pastures Grove, Barford, Norwich	Greengate Homes (Barford) Limited Nelson House The Street Hempnall NORWICH NR15 2AD	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 2 County of Norfolk		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Hilltop Barn, Burdock Lane, Barford, Norwich, NR9 4BL	Peter Richard Saul Hilltop Barn Burdock Lane Barford NORWICH NR9 4BL Sarah Anne Saul Hilltop Barn Burdock Lane Barford NORWICH NR9 4BL	
N/A	1 Pastures Grove, Barford, Norwich, NR9 4DZ	Ann Paterson Halls 1 Pastures Grove Barford NORWICH NR9 4DZ Stephen Robert Halls 1 Pastures Grove Barford NORWICH NR9 4DZ	
N/A	Old Orchard House, The Street, Weybourne, Holt, Norfolk, NR25 7SY	Yvonne Anne Marie Odrowaz-Pieniazek Old Orchard House The Street Weybourne HOLT Norfolk NR25 7SY	

Page 1029 of 1368

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-001	5165 square metres of foreshore land seaward of the Mean High Water (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted by lease dated 18 November 2009) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted by lease dated 18 November 2009)	
01-002	1523 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-002 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-003	32 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-003 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Minis Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-004	1195 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
01-005	366 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-005 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	
		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-006	814 square metres of beach and public footpath (Weybourne FP7) (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-006 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-007	4767 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-007 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-008	765 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-008 conťd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)	
		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	
		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018)	
		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-009	2135 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Unknown	

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-010	1680 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Unknown	
01-011	652 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018)	

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-011 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-012	36 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-012 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-013	57 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-013 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-014	4655 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-014 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-015	393 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-016	33 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Ministry of Defence Ministry of Defence Ministry of Defence Ministry of Defence Ministry of Defence Ministry alliding Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe over part)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-017	51 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-018	7854 square metres of access track (north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dereham and District Angling Club 13 New Road Gaywood KINC'S LYNN PE30 4BB (in respect of right of way over access roads) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-018 cont'd		Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads) Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-018 cont'd		The Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-019	10156 square metres of access track and public footpath (Weybourne FP7) (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-019 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-020	9694 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dereham and District Angling Club 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-020 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads) Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-020 cont'd		Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-020 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access)	
01-021	75 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-021 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-022	24 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-022 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-023	630 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-023 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-024	3359 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	
		Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	
		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-024 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-025	4474 square metres of grassland and access track (Weybourne Military Camp) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-025 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of rights granted) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	
01-026	265 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access and right to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-026 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights reserved by lease dated 30 May 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-027	304 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-027 cont'd		Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-028	5462 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edga Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-028 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-029	9686 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-029 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-030	8080 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electricity cables and junction point)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-030 cont'd		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of rights granted by lease dated 13 March 2018) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) Unknown (in respect of exceptions, reservations and declarations)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-031	4437 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-031 cont'd		TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of electricity apparatus) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-032	21 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-032 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) Unknown (in respect of right of access and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-033	9105 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-033 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SEI 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-033 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-034	22426 square metres of grassland (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	
		Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point)	
		Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-034 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-035	15 square metres of public highway (The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-035 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Openreach Limited Kelvin House 123 Judd Street LONDON WCH 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-035 cont'd		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-036	156 square metres of public highway (The Street, A149) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-037	29 square metres of access track (north of The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay, use and maintain of electricity cables and junction point) Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of rights granted by leases dated 2 December 2015 and 13 July 2016) Dereham and District Angling Club 13 New Road Gaywood KING'S LYNN PE30 4BB (in respect of right of way over access roads) Dudgeon Offshore Wind Limited 1 Kingdom Street LONDON W2 6BD (in respect of right of access to lay, install, use and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-037 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SET 6NP (in respect of right of way over access track, right to lay and maintain electricity apparatus and restrictive covenants) Holt Sea Angling Club 17 St. Josephs Road SHERINGHAM NR26 8JA (in respect of right of way over access roads) Kelling Model Flying Club 40 Spenser Avenue NORTH WALSHAM Norfolk NR28 9HZ (in respect of right of way over access roads) Norfolk Angling Club 32 Cromer Road WEST RUNTON Norfolk NR27 9AD (in respect of right of way over access roads) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-037 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) TC Dudgeon OFTO PLC 3 More London Riverside LONDON SE1 2AQ (in respect of rights granted by lease dated 13 March 2018) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe) The University Of East Anglia Norwich Research Park Earlham Road NORWICH NR4 7TJ (in respect of rights of access) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-038	707 square metres of public highway and verges (The Street, A149) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
01-039	1 square metres of access track (south of The Street, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-040	3 square metres of access track (south of The Street, A149) (North Norfolk District Council)	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-040 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables) Unknown	
01-041	285 square metres of access track and land (south of The Street, A149) (North Norfolk District Council)	Benjamin George Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-041 cont'd		Joanna Elizabeth Soper Highfields Weybourne HOLT Norfolk NR25 7EH (in respect of right of way) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
01-042	7698 square metres of hedgerow and arable land (south of The Street, A149) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)	
01-043	486 square metres of access track, woodland and arable land (south of The Street, A149) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-043 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
01-044	36555 square metres of arable land and hedgerow (north of Holgate Hill) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-001	2057 square metres of arable land (north of Holgate Hill) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
02-002	442 square metres of public highway and verges (Holt Road) (North Norfolk District Council)	Unknown	
02-003	2844 square metres of arable land, hedgerow and verges (north of Holgate Hill) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-003 cont'd		Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	



Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
583 square metres of public highway and verge (Holgate Hill) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus)	
816 square metres of arable land and hedgerow (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus)	
	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and electricity cables)	
	583 square metres of public highway and verge (Holgate Hill) (North Norfolk District Council) 816 square metres of arable land and hedgerow (south of Holgate Hill and	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-005 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and electricity cables)	
02-006	323 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Unknown	
02-007	305 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) Unknown	
02-008	45 square metres of track and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-008 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) Unknown	
02-009	730 square metres of track, arable land and verges (south of Holgate Hill and Holt Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) Unknown (in respect of restrictive covenants stated in conveyance dated 23rd December 1954)	
02-010	45859 square metres of arable land, grassland, hedgerow, watercourse (Spring Beck) and verges (east of Holt Road and west of Station Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted in deed dated 12th August 2011)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

on Land Plans Description of Land over water) which it is proposed shall be extinguished, suspended or interferent Infrastructure Planning (Applications: Prescribed Forms and Pro- Scira Offshore Energy Limited Wind Farm Place 02-010 cont'd Scira Offshore Energy Limited Wind Farm Place Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted in deed dated 12th August 2011) Unknown (in respect of restrictive covenants stated in conveyance dated 23rd December 02-012	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
cont'd Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted in deed dated 12th August 2011) Unknown (in respect of restrictive covenants stated in conveyance dated 23rd December 02-012 871 square metres of public highway and verges (Station Road) (North	Description of Land over water) which it is proposed s	asements or other private rights over land (including private rights of navigation Il be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Applications: Prescribed Forms and Procedures) Regulations 2009	
	Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted in deed Unknown		
Norfolk District Council) Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON UONDON WC1H 9NP (in respect of telecommunications) Unknown	Exercise Council) Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)		



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-013	952 square metres of public highway and verges (Station Road) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
02-014	2367 square metres of arable land, access track and verges (east of Station Road) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-014 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)	
02-015	60663 square metres of arable land, access track, woodland and hedgerows (east of Station Road) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Busineess Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)	
03-001	214 square metres of public highway and verges (Sheringham Road, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-002	8425 square metres of access track and verges (south of Sheringham Road, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted in lease dated 05th March 2009)	
03-003	653 square metres of railway track, compound and verges (north east of Weybourne Train Station) (North Norfolk District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants) Unknown	
03-004	1829 square metres of railway track, compound and verges (north east of Weybourne Train Station) (North Norfolk District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants) Unknown	
03-005	7893 square metres of arable land (east of Sandy Hill Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-005 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
03-006	77 square metres of public highway and verges (Sandy Hill Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-006 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
03-007	22 square metres of access track (east of Sandy Hill Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-010	15195 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-010 cont'd		Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Artington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamilingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-010 cont'd		Susan Rodda Burton 60 Fitzwatter Road Fitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamilingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-010 cont'd		William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-002	1869 square metres of woodland and track (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Fitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-002 cont'd		Green Farm Property Limited 1 Green Farm Industrial Units Wendower Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media) Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting Mordial	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-002 cont'd		Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turmbull 9 Cowslip Lane Gamilingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamilingay SANDY SG19 3LZ (in respect of right of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-002 cont'd		The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wittshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way) Theresa Patricia Mason 41 Artington Way THETFORD Notrolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-003	38365 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-003 cont'd		Ivy Elizabeth Black 58A Grove Lane HOLT NR25 GED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Artington Way THETFORD Norfolk IP24 2D2 (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Tumbull 9 Cowelip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-003 cont'd		Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SWIP 4DF (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-003 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-004	9 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004 cont'd		Catherine Elizabeth Smith 60 Fitzwalter Road Filtch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004 cont'd		Ivy Elizabeth Black SA Grove Lane HOLT NR25 6ED (in respect of right of access) Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamilngay SANDY SAIDY SAIDY	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004 cont'd		Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SWIP 4DF (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-004 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-005	13 square metres of public highway and verge (Sandy Hill Lane) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-006	43 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
04-007	10 square metres of public highway (Sandy Hill Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect to a right of way/passage over the track)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-008	13 square metres of public highway (Sandy Hill Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect to a right of way/passage over the track)	
04-009	2108 square metres of access track and verges (east of Sandy Hill Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Unknown (in respect to a right of way/passage over the track)	
04-010	35 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-011	1355 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Filtch Green DUMMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-011 cont'd		Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-011 cont'd		Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SWIP 4DF (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-011 conťď		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-012	67 square metres of hedgerow and verge (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	
04-013	465 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Catherine Elizabeth Smith 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-013 cont'd		Clare Frances Todd The Old Stables Sandy Hill Lane Weybourne HOLT Norfolk NR25 7HW (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Clive Malcolm Hay-Smith Abbey Farm The Street Weybourne HOLT Norfolk NR25 7SZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Green Farm Property Limited 1 Green Farm Property Limited 1 Green Farm Industrial Units Wendover Road Rackheath Norwich NR13 6LQ (in respect of right of access and right of passage of services through conducting media) Ivy Elizabeth Black 58A Grove Lane HOLT NR25 6ED (in respect of right of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-013 cont'd		Nicola Louise Lambert-John Redland Cottage 5 Weston Wood Close NORWICH Norfolk NR7 0JX (in respect of right of access and right of passage of services through conducting media) Philip Raymond Froud 41 Artington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Roberta Boyce 7 Ormesby Chine South Woodham Ferrers CHELMSFORD CM3 7AR (in respect of right of access and right of passage of services through conducting media) Stephen Elliott Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-013 cont'd		Susan Rodda Burton 60 Fitzwalter Road Flitch Green DUNMOW Essex CM6 3FH (in respect of right of access and right of passage of services through conducting media) Susan Turnbull 9 Cowslip Lane Gamlingay SANDY SG19 3LZ (in respect of rights of access as contained in a Transfer dated 29th March 2016) The National Trust For Places Of Historic Interest Or Natural Beauty Heelis Kemble Drive Swindon Wiltshire SN2 2NA (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SWIP 4DF (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-013 cont'd		Theresa Patricia Mason 41 Arlington Way THETFORD Norfolk IP24 2DZ (in respect of right of access, right to fell and transport trees and right of passage of services through conducting media) Unknown (in respect of right of way and service medias) Unknown (in respect of rights and easements) William Martin Redman 7 Arwen Grove South Woodham Ferrers CHELMSFORD CM3 5ZJ (in respect of right of access and right of passage of services through conducting media)	
04-014	277 square metres of woodland and track (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown	
04-015	46631 square metres of arable land, hedgerow and verges (east of Sandy Hill Lane) (North Norfolk District Council)	Unknown (in respect to a right of way/passage over the track)	
04-017	38597 square metres of arable land, grassland and hedgerow (east of Gipsies' Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-017 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
05-001	182 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
05-002	642 square metres of arable land, hedgerow and verge (east of Sandy Hill Lane) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-004	2000 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
05-005	360 square metres of public highway and verges (Holt Road, A148) (North Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-005 cont'd		Unknown	
05-006	4033 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Unknown	
05-007	1530 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Lemur UK LLC 10960 Wilshire Blvd 11th Floor LOS ANGELES CA 90024 UNITED STATES OF AMERICA (in respect of right of access for maintenance)	
05-008	6023 square metres of arable land and hedgerow (south of Holt Road, A148) (North Norfolk District Council)	Lemur UK LLC 10960 Wilshire Blvd 11th Floor LOS ANGELES CA 90024 UNITED STATES OF AMERICA (in respect of right of access for maintenance)	
05-009	34484 square metres of arable land and hedgerow (south of Holt Road, A148 and north of The Street) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-010	655 square metres of access track and hedgerow (north of The Street) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
05-011	27 square metres of public highway (The Street) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
05-012	811 square metres of public highway and verges (The Street) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-012 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
05-013	306 square metres of verges and hedgerow (south of The Street) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
05-014	355 square metres of arable land, verges and hedgerow (south of The Street) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-015	39060 square metres of arable land and hedgerow (south of The Street) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
05-016	23118 square metres of arable land and hedgerow (north of Norfolk Coast, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted) Unknown (in respect of right of access to lay and maintain electricity cables contained in transfer dated 28 July 1995)	
06-001	242 square metres of arable land and hedgerow (north of Norfolk Coast, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted) Unknown (in respect of right of access to lay and maintain electricity cables contained in transfer dated 28 July 1995)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-002	466 square metres of verges and public highway (Norfolk Coast, A149) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
06-003	65 square metres of verge (Norfolk Coast, A149) (North Norfolk District Council)	Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-004	7320 square metres of private road and public footpath (Bodham FP8) (west of New Road, A149) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
06-005	61348 square metres of arable land and public footpath (Bodham FP8) (Pond Farm, west of New Road, A149) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-005 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
06-007	549 square metres of public highway (New Road, Holt, A149) (North Norfolk District Council)	Unknown	
07-001	16979 square metres of arable land and hedgerow (south east side of New Road, A149) (North Norfolk District Council)	Arthur Richard Gee The Bungalow Chestnut Farm Castle Road Baconsthorpe HOLT NR25 6L (in respect of right of way) BW Sipp Trustees Limited St. James House St. James Square CHELTENHAM Gloucestershire GL50 3PR (in respect of right of way) Margaret Thelma Gee The Bungalow Chestnut Farm Castle Road Baconsthorpe HOLT NR25 6L (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-004	778 square metres of public highway being Gresham Road (north of Gresham Road) (North Norfolk District Council)	Unknown	
07-006	16784 square metres of arable land and hedgerow (north of Gresham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter) Unknown (in respect of drainage rights reserved by transfer dated 25 March 2008)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-007	1133 square metres of arable land and private track situated (south east Side Of New Road, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
07-008	101 square metres of private track (north of Gresham Road and on the south side of New Road, A149) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
07-009	392 square metres of verges and public highway (Gresham Road) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
07-010	606 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-010 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-011	1788 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-011 cont'd		Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay, use and maintain of electricity cables and junction point) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-011 conťď	4 square metres of arable land (north of Gresham Road) (North Norfolk District Council)	Unknown	
07-013	2 square metres of arable land (south of Gresham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electricity apparatus) Unknown	
07-014	2345 square metres of arable land (south of Gresham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of electric) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electric) Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-014 cont'd		William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)	
07-015	24769 square metres of arable land and hedgerow (south of Gresham Road) (North Norfolk District Council)	Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of drainage rights reserved by transfer dated 25 March 2008) William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of drainage rights reserved by transfer dated 25 March 2008)	
07-016	8641 square metres of arable land, hedgerow and private track (north of Church Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-016 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-017	194 square metres of arable land (north of Church Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunication cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-017 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-018	29 square metres of agricultural land (north of Church Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
07-019	452 square metres of hedgerow and public highway (Church Lane) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-020	435 square metres of arable land (south of Church Lane) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
07-021	41848 square metres of arable land, hedgerow and public footpath (Matlask FP1) (south of Church Lane) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
08-001	52920 square metres of arable land and hedgerow (north of Northfield Lane) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
08-002	1129 square metres of arable land and hedgerow (Barningham Park, north of Northfield Lane) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-001	892 square metres of public highway and verges (Northfield Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
09-002	519 square metres of arable land and public footpath (Plumstead FP11) (south east of Northfield Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-002 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
09-003	35596 square metres of arable land and public footpath (Plumstead FP11) (south east of Northfield Lane) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-003 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
09-004	15597 square metres of arable land and hedgerow (west of The Street, A149) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-005	124 square metres of arable land (north of Matlaske Road (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
09-006	777 square metres of public highway and verges (Matlaske Road) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
09-007	196 square metres of arable land (south of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-007 conť d		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
09-008	2035 square metres of arable land (south of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
09-009	41722 square metres of arable land and hedgerow (south of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to lay and maintain electric lines and right to enter)	

equinor 🐉

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-009 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to lay and maintain electric lines and right to enter)	
10-001	31524 square metres of arable land and hedgerow (Ferndale Farm, south west of The Street, A149) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-002	88 square metres of hedgerow (Ferndale Farm, south west of The Street, A149) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-002 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-003	550 square metres of arable land (Ferndale Farm, north of Little Barningham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-004	347 square metres of public highway and verges (Little Barningham Road) (North Norfolk District Council)	Unknown	
10-005	71 square metres of verge (south west of Little Barningham Road) (North Norfolk District Council)	Unknown	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-006	685 square metres of arable land (Ferndale Farm, south of Little Barningham Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-007	19814 square metres of arable land and hedgerow (Ferndale Farm, south of Little Barningham Road) (North Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-007 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	
10-008	400 square metres of public highway and verge (Sweetbriar Lane) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-009	1104 square metres of arable land and access track (south of Sweetbriar Lane) (excluding all interests of the Crown) (North Norfolk District Council)	Arthur Thomas Steel Flat 29 Charleville Mansions Charleville Road LONDON W14 9JA (in respect of rights granted) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Christopher Edward Steel 56 Nasmyth Street LONDON W6 0HB (in respect of rights granted) Emma Louise Suffield Dairy Farm Gunton Park Hanworth NORWICH Norfolk NR11 7HL (in respect of legal easements reserved)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-009 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant) Unknown (in respect of rights of way, wayleaves, drainage and watercourse easements)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-009 cont'd		William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of legal easements reserved)	
10-010	26850 square metres of arable land and hedgerow (south of Sweetbriar Lane and west of Matlaske Road) (excluding all interests of the Crown) (North Norfolk District Council)	Arthur Thomas Steel Flat 29 Charleville Mansions Charleville Road LONDON W14 9JA (in respect of rights granted) Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Christopher Edward Steel 56 Nasmyth Street LONDON W6 0HB (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-010 cont'd		Emma Louise Suffield Dairy Farm Gunton Park Harworth NORWICH Norfolk NR11 7HL (in respect of legal easements reserved) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-010 conťď		Unknown (in respect of rights of way, wayleaves, drainage and watercourse easements)	
		William Robert Bartle Edwards Hardingham Hall High Common Hardingham NORWICH Norfolk NR9 4AE (in respect of legal easements reserved)	
10-012	462 square metres of verge and public highway (Matlaske Road) (North Norfolk District Council)	Unknown	
10-013	10291 square metres of arable land and hedgerow (Ferndale Farm, Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right of access to lay and maintain electricity cables) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-014	53753 square metres of arable land, hedgerow and public footpath (Itteringham FP3) (south east of Matlaske Road) (North Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Mere Farm (Mannington) Limited Sankence Farm Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N SEH (in respect of right of access to lay and maintain gas pipeline)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-014 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved and granted)	
11-001	12 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline) Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-001 cont'd		Michael John Chantler Compass House Lypiatt Road CHELTENHAM GL50 2QJ (in respect of rights reserved) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Outon NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved and granted)	
11-002	1298 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipeline)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-002 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Mere Farm (Mannington) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of right of access to maintain water pipe and right of passage of water through water pipe) Michael John Chantler Compass House Lypiatt Road CHELTENHAM GL50 2CJ (in respect of rights reserved) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline)	

Page 1156 of 1368



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-002 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Unknown (in respect of rights reserved and granted)	
11-004	672 square metres of public highway and verge (Matlaske Road) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
11-005	620 square metres of verge and hedgerow (Matlaske Road) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-006	28631 square metres of arable land and hedgerow (Mere north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of easement contained in deed dated 12 September 2011) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus) Nicholas Benjamin Harris Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of rights stated in conveyance dated 5 November 1982) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-006 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of easement contained in deed dated 12 September 2011) Seth Elton Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of right of access to conduct drainage maintenance stated in conveyance dated 5 November 1982)	
11-007	40407 square metres of arable land and hedgerow Mere Farm and Wood Farm, north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of easement contained in deed dated 12 September 2011)	

equinor 😽

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-007 cont'd		Nicholas Benjamin Harris Mere Farmhouse Matiaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of rights stated in conveyance dated 5 November 1982) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Outon NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of easement contained in deed dated 12 September 2011)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-007 cont'd		Seth Elton Harris Mere Farmhouse Matlaske Road Mannington NORWICH Norfolk NR11 7BE (in respect of right of access to conduct drainage maintenance stated in conveyance dated 5 November 1982)	
11-008	5466 square metres of arable land and hedgerow (north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-008 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-001	865 square metres of arable land (north west of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-001 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-002	1592 square metres of public highway, verges and hedgerow (Matlaske Road) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-002 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right of access to lay and maintain electricity cables) Unknown	
12-003	173 square metres of wooded land (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-003 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-004	8386 square metres of arable land and hedgerow (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

equinor 😽

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-004 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-005	792 square metres of arable land (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of right to access to lay and maintain electricity cables) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-005 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of right to access to lay and maintain electricity cables)	
12-006	40552 square metres of arable land, hedgerow and public footpath (Corpusty FP17) (Wood Farm, south of Matlaske Road) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Oak Park Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting and sporting rights)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-006 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) Sharon Brooks Wood Farm Saxthorpe Road Itteringham NORWICH NR11 7BG (in respect of private water supply)	
12-007	18286 square metres of arable land and public footpath (Corpusty FP16) (south east of Matlaske Road) (North Norfolk District Council)	Blue Transmission Sheringham Shoal Limited Cannon Place 78 Cannon Street London EC4N 6AF (in respect of rights granted) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-007 cont'd		Scira Offshore Energy Limited Wind Farm Place Edgar Road WALSINGHAM Norfolk NR22 6EJ (in respect of rights granted) Unknown (in respect of restrictive covenants)	
13-001	4758 square metres of arable land and hedgerow (Hall Farm, Aylsham Road, B1354) (North Norfolk District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of shooting rights) (in respect of a right of pre-emption contained in an Agreement dated 15 October 2010)	
13-004	1112 square metres of verge and public highway (Blickling Road, B1354) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of electricity apparatus) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-005	76 square metres of verge, hedgerow and arable land (Lake Farm, north of Aylsham Road, B1354) (North Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
13-006	6505 square metres of hedgerow and arable land (Saxthorpe Hall, south of Aylsham Road, B1354) (North Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
13-007	27 square metres of public highway and verge (Aylsham Road, B1354) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights reserved and granted)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-008	30 square metres of public highway and verge (Aylsham Road, B1354) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights reserved and granted)	
13-009	2481 square metres of arable land and hedgerow (south of Aylsham Road, B1354) (Broadland District Council)	National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) Unknown (in respect of rights reserved and granted)	
13-010	56523 square metres of arable land and hedgerow (north of Spa Lane and south of Aylsham Road, B1354) (Broadland District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-010 cont'd		National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strandv LONDON WC2N 5EH (in respect of gas transmission pipeline) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights reserved and granted)	
13-011	194 square metres of arable land (north of Spa Lane and south of Aylsham Road, B1354) (Broadland District Council)	National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain pipeline and restrictive covenants) Unknown (in respect of rights reserved and granted)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-012	6 square metres of arable land (north of Spa Lane) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
13-013	694 square metres of verges and public highway (Spa Lane) (Broadland District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas transmission pipeline) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-014	879 square metres of arable land (Irmingland Hall Estate, south of Spa Lane) (Broadland District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline and restrictive covenants)	
13-015	58 square metres of arable land (Church Farm, south of Spa Lane) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
13-016	32735 square metres of arable land and hedgerow (Irmingland Hall Estate, south of Spa Lane) (Broadland District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-016 conťd		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipeline and restrictive covenants)	
14-001	35234 square metres of arable land, hedgerow and public footpath (Oulton FP4) (Church Farm, north of Spink's Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
14-002	269 square metres of arable land (Church Farm, north of Spink's Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

Page 1175 of 1368



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-003	797 square metres of verge, hedgerow and public highway (Spink's Lane) (Broadland District Council)	Unknown	
14-004	19 square metres of verge and public highway (Spink's Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
14-005	29 square metres of public highway (Spink's Lane) (Broadland District Council)	Unknown	
14-006	895 square metres of arable land and hedgerow (Church Farm, south of Spink's Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
14-007	56335 square metres of arable land, hedgerow and public footpath (Heydon FP2 and Oulton FP9) (Church Farm, south east of Spink's Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
15-002	577 square metres of verge mad public highway (B1149) (Broadland District Council)	Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-003	2295 square metres of arable land (The Grange, north of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right to maintain and use service media and right of entry)	
15-004	87228 square metres of arable land and hedgerow (The Grange, north of B1149) (Broadland District Council)	E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right to maintain and use service media and right of entry) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
16-001	957 square metres of verge and public highway (Heydon Road) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-001 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of electricity apparatus) Unknown	
16-002	1 square metres of verge (Heydon Road) (Broadland District Council)	Unknown	
16-003	23468 square metres of arable land and hedgerow (north west of Beerhouse Farm, north east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-003 cont'd		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of rights granted by Transfer dated 7th April 2009)	
16-004	912 square metres of arable land and hedgerow (north east of Beerhouse Farm, east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-004 cont'd		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of rights granted by Transfer dated 7th April 2009)	
16-005	838 square metres of wooded area and public highway (B1149 and Heydon Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
16-006	1390 square metres of verge and private highway (east of B1149) (Broadland District Council)	Unknown	

Page 1180 of 1368

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-007	158 square metres of arable land and hedgerow (north east of Beerhouse Farm, B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access and right of passage of services through conducting media)	
		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access and right of passage of services through conducting media)	
16-007 cont'd		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access and right of passage of services through conducting media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-008	311 square metres of arable land (east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
16-008 cont'd		Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right reserved by deed dated 30 December 1969)	
16-009	55342 square metres of arable land and hedgerow (east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
16-009 cont'd		Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants) Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right reserved by deed dated 30 December 1969)	
16-010	1847 square metres of arable land and hedgerow (north east of B1149) (Broadland District Council)	Alexander Gavin Angell Lane Manor Farm TITTLESHALL Norfolk PE32 2RH (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
		E.F. Harrold Limited Street Farm The Street Oulton NORWICH NR11 6AF (in respect of right of access to lay and maintain water pipe) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Jennifer Ponsonby 6 Heathview Gardens LONDON SW15 3SZ (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)
16-010 cont'd		Rhona Jane Kirwan Bulwer-Long Heydon Hall Heydon NORWICH Norfolk NR11 6RE (in respect of right of access to lay and maintain apparatus, right of passage of services through conducting media, drainage rights and restrictive covenants)

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-011	890 square metres of verge and public highway (The Street) (Broadland District Council)	Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right reserved by deed dated 30 December 1969) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
16-012	18282 square metres of arable land and hedgerow (Docking Farm, east of B1149) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
16-013	232 square metres of verge, hedgerow and public highway (The Street) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

Page 1187 of 1368



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
16-014	259 square metres of verge, hedge row and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	
16-014 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987)	
16-014 conťd		Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)	
16-015	502 square metres of hedgerow and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	
16-015 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987) Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)	
16-016	2337 square metres of arable land (Docking Farm, east of B1149) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
16-017	258 square metres of verge, arable land and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	Christopher Herbert Harrold Docking Farm Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Rita Durham 1 Docking Farm Cottages Oulton NORWICH NR11 6QZ (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media)	
16-017 cont'd		Saltcarr Farms Limited Saltcarr House Bure Valley Lakes Oulton NORWICH Norfolk NR11 6NW (in respect of right of way over access tracks, right of access to lay and maintain apparatus and right of passage of services through conducting media) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Unknown (in respect of right of way over access tracks contained in assent dated 2 October 1987)	
16-020	78814 square metres of arable land and hedgerow (Docking Farm, east of B1149) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
17-001	2108 square metres of tree and hedge line (east of Marriott Way, B1149) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of restrictive covenants)	
17-003	87 square metres of hedgerow (east of B1149) (Broadland District Council)	WM.Youngs & Son (Farms) Limited Sankence Farm Sankence Aylsham NORWICH Norfolk NR11 6UW (in respect of rights granted)	
17-005	1726 square metres of public highway and verge (B1149) (Broadland District Council)	Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-006	2065 square metres of arable land (east of B1149) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
17-007	19627 square metres of arable land and hedgerow (Dixs House, north of Bird's Lane and south west of B1149) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
17-009	560 square metres of verge, hedgerow and public highway (Birds Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
17-012	51650 square metres of arable land and hedgerow (west of Holt Road, B1149) (Broadland District Council)	Unknown (in respect of a right to enter the land and maintain and repair the water pipes in respect of rights reserved)	
18-001	27689 square metres of arable land and hedgerow (north of Aylsham Road, B1145) (Broadland District Council)	Matthew Paul Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of rights granted)	
18-001 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Pauline Susan Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)	
18-002	252 square metres of arable land and verge (north of Aylsham Road, B1145) (Broadland District Council)	Matthew Paul Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Pauline Susan Brockis Whitehouse Farm High Street Cawston NORWICH NR10 4AA (in respect of sewerage rights)	
18-003	813 square metres of verges and public highway (B1145) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
18-007	515 square metres of hedgerow and public highway (Old Friendship Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-008	217 square metres of arable land and hedgerow (south west of Old Friendship Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	
18-009	20048 square metres of arable land and hedgerow(south of Old Friendship Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-009 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	
18-011	1 square metres of arable land (north of Norwich Road) (Broadland District Council)	The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	
18-013	450 square metres of arable land and hedgerow (north of Norwich Road) (Broadland District Council)	The Diocese Of Norwich Education And Academies Trust Diocesan House 109 Dereham Road Easton NORWICH Norfolk NR9 5ES (in respect of right of access to maintain pavilion)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-014	547 square metres of arable land and public footpath (Cawton FP7 and FP8) (north of Norwich Road) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Unknown	
18-015	9454 square metres of arable land (north east of Norwich Road) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of a right to enter the land and maintain and repair the water pipes in respect of rights reserved)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-016	998 square metres of verge and public highway (Norwich Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
19-001	42844 square metres of arable land and hedgerow (Church Farm, north of Reepham Road) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
19-003	507 square metres of arable land and hedgerow (north of Reepham Road) (Broadland District Council)	Pagepost Limited Ollands Farm Heydon NORWICH Norfolk NR11 6RB (in respect of rights of access)	
19-004	1535 square metres of Section of verge, hedgerow and public highway (Reepham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-005	209 square metres of hedgerow (Reepham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
19-006	30564 square metres of arable land and hedgerow (Church Farm, south of Reepham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
19-008	458 square metres of arable land (south of Reepham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
19-009	15 square metres of public highway (Unnamed Road, south of Reepham Road) (Broadland District Council)	Unknown	
19-010	24898 square metres of arable land and hedgerow (north of Church Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-010 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)	
20-001	465 square metres of Section of verge, hedgerow and public highway (Church Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
20-002	266 square metres of arable land and hedgerow (south of Church Lane) (Broadland District Council)	P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-003	75441 square metres of arable land and hedgerow (south of Church Lane) (Broadland District Council)	P.D. Chapman Limited Guton Hall Guton Hall Lane Brandiston NORWICH NR10 4PH (in respect of rights granted)	
21-001	31318 square metres of arable land and hedgerow (west of Clay Lane) (Broadland District Council)	Unknown (in respect of right of way restrictive covenants)	
21-002	584 square metres of verge, hedgerow and public highway (Clay Lane) (Broadland District Council)	Unknown	
21-005	8955 square metres of wooded area (east of Clay Lane) (Broadland District Council)	Unknown (in respect of shooting rights)	
21-006	33434 square metres of wooded land, arable land, hedgerow and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
21-007	3032 square metres of arable land and tree line (north east of Church Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of way restrictive covenants)	
21-009	75 square metres of wooded land and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Unknown (in respect of right of way restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-010	5780 square metres of wooded land, arable land and hedgerow and public footpath (Swannington FP4) (north east of Church Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of way stated in Conveyance dated 03 April 1965)	
21-012	1392 square metres of hedgerow, private road (Church Lane) and public footpath (Swannington FP6) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-012 cont'd		Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (In respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN (in respect of right of way over access track) The Mutimer Partnership Limited Woodlands Farm Church Lane Swannington NORWICH Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of way over access track)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-013	43782 square metres of arable land, private road (Church Lane), hedgerow and public footpath (Swannington FP6 and Swannington FP7) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Five Ways Farm Limited Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-013 conťd		Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track)	
21-015	197 square metres of private road and public footpath (Swannington FP5) (Woodlands Farm, north of Church Lane) (Broadland District Council)	Unknown (in respect of rights of access reserved)	
21-016	669 square metres of private road and public footpath (Swannington FP5) (north east of Church Lane) (Broadland District Council)	Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-017	183 square metres of private road, farm courtyard and public footpath (Swannington FP5) (Woodlands Farm, Church Lane) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passage of services granted) Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passge of services granted) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of rights of access and passge of services granted) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of way over access track)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-018	2025 square metres of arable land, private road (Church Lane) and public footpath (Swannington FP5 and FP6) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track) George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-018 cont'd		Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track)	
21-019	142 square metres of hedgerow, private road (Church Lane) and public footpath (Swannington FP6) (Broadland District Council)	Alex Peter Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Five Ways Farm Limited Five Ways Farm The Street Swannington NORWICH Norfolk NR9 5NW (in respect of right of way over access track)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-019 conťď		George Thomas Robert Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Helen Frances Mutimer Woodlands Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track, right of access to lay and maintain conducting media and right of passage of services through conducting media) Stephanie Rowley Nut Pit Farm Church Lane SWANNINGTON Norfolk NR9 5NN (in respect of right of way over access track) Unknown (in respect of right of access and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-001	337 square metres of Section of watercourse (Trout Stream) (south west of Church Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
22-002	32927 square metres of wooded land, arable land, hedgerow and public footpath (Swannington FP8) (north of School Road) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
22-003	982 square metres of of hedgerow and public highway (Upgate Road) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
22-004	394 square metres of arable land and public footpath (Swannington FP8) (north of Upgate Road) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-005	19 square metres of verge, public highway (Upgate Road) and public footpath (Swannington FP8) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
22-006	29 square metres of public highway (Upgate Road) (Broadland District Council)	Unknown	
22-007	316 square metres of verge, arable land (west of Upgate Road) and public footpath (Swannington FP13) (Broadland District Council)	Unknown (in respect of right of access and right of light and air and to lay water pipes)	
22-008	39526 square metres of arable land and hedgerow and public footpath (Swannington FP13) and restricted public byway (Swannington RB12) (south of Upgate Road) (Broadland District Council)	Unknown (in respect of right of access and right of light and air and to lay water pipes)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-009	28761 square metres of arable land and hedgerow (north of Reepham Road) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Graham Martin Dacre Drayton Hall Hall Lane Drayton NORWICH Norfolk NR6 6DP (in respect of personal covenants contained in transfer dated 3 April 2018) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Vodafone Limited Vodafone Limited Vodafone Limited Vodafone Limited NEWBURY Berkshire RG14 2FN (in respect of right of access to lay and maintain service media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-010	1259 square metres of verge and public highway (Reepham Road) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
22-011	38380 square metres of arable land and hedgerow (south of Reepham Road) and restricted byway (Attlebridge RB1) (Broadland District Council)	Graham Martin Dacre Drayton Hall Hall Lane Drayton NORWICH Norfolk NR8 6DP (in respect of personal covenants contained in transfer dated 3 April 2018) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to lay and maintain service media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-001	1359 square metres of public highway (Marriott's Way) (Broadland District Council)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of disused railway) Unknown	
23-003	31610 square metres of arable land and hedgerow (south of Marriott's Lane and north of Felthorpe Road) (Broadland District Council)	Unknown (in respect of right of way)	
23-004	445 square metres of verge and public highway (Felthorpe Road) (Broadland District Council)	Unknown	
23-005	585 square metres of arable land and hedgerow (north of Felthorpe Road) (Broadland District Council)	Unknown (in respect of right of way)	
23-006	2 square metres of verge (Felthorpe Road) (Broadland District Council)	Unknown	
23-007	22093 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of right of way)	
23-008	30747 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-008 conť d		Unknown (in respect of right of way)	
23-009	258 square metres of verge and public highway (Old Fakenham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
23-010	180 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right to take water from pump and well and restrictive covenants)	
23-011	108 square metres of hedgerow (east of Fakenham, A1067) (Broadland District Council)	Unknown	
23-012	1300 square metres of verge and public highway (Old Fakenham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
23-013	6142 square metres of arable land and hedgerow (east of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right to take water from pump and well and restrictive covenants)	
23-014	3726 square metres of verge and public highway (Fakenham Road, A1067) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-014 cont'd		Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications)	
23-015	113 square metres of verge and public highway (Ivy Cottages, south west of Fakenham Road, A1067) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications)	
23-016	724 square metres of private highway and arable land (south west of Fakenham Road, A1067) (Broadland District Council)	Keith Robert Miller 1 Ivy Cottages Fakenham Road ATTLEBRIDGE Norfolk NR9 5TG (in respect of right of way)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-016 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Simon James Cram 2 Ivy Cottages Fakenham Road ATTLEBRIDGE Norfolk NR9 5TG (in respect of right of way) Unknown (in respect of right of way)	
23-017	20334 square metres of arable land and hedgerow (south west of Fakenham Road, A1067) (Broadland District Council)	Unknown (in respect of right of way)	
23-018	622 square metres of watercourse (River Wensum) (south west of Fakenham Road, A1067) (Broadland District Council)	Unknown	
23-022	989 square metres of arable land and private road (south of Fakenham Road, A1067) (Broadland District Council)	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-022 cont'd		Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access)	
23-023	946 square metres of private road (south of Fakenham Road, A1067) (Broadland District Council)	Billy Joe Baldry The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access) Oliver Wales Field Farm Ringland Lane Weston Longville NORWICH NR9 5JT (in respect of right of access) Sadie Michelle Clarke The Lodge Fakenham Road Morton on the Hill NORWICH NR9 5SP (in respect of right of access)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-023 cont'd		William Anthony Wales Abbey Farm Warham Road Binham FAKENHAM Norfolk NR21 0DQ (in respect of right of access)	
23-028	165 square metres of public highway (Fackenham Road, A1067) (Broadland District Council)	Margaret Ann Prince-Smith Morton Hall Morton Hall Estate Morton On The Hill NORWICH Norfolk NR9 5JS (in respect of personal covenants relating to erection of boundary fence) Unknown (in respect of rights reserved) Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications)	
23-031	85136 square metres of arable land, woodland and hedgerow (west of Fackenham Road, A1067) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-031 conťď		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
24-002	19 square metres of access track and verge (to the east of Ringland Lane) (Broadland District Council)	Unknown	
24-003	106 square metres of verge and public highway (Ringland Lane) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
24-005	821 square metres of public highway (Ringland Lane) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
24-006	608 square metres of arable land and private access road (west of Ringland Lane) (Broadland District Council)	Unknown (in respect of rights reserved by conveyance dated 6 January 1983)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
24-007	103393 square metres of hedgerow and arable land (north east of Church Hill Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights reserved by conveyance dated 6 January 1983)	
25-001	891 square metres of ditch and thicket (east of Church Hill Lane) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
25-002	4 square metres of private road (east of Church Hill Lane) (Broadland District Council)	Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-003	85 square metres of verge and public highway (Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
25-004	16 square metres of verge (east of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
25-005	1524 square metres of arable land (east of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-005 cont'd		Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millistream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of right of access to lay and maintain gas main and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-006	61433 square metres of hedgerow, arable land and verge (east of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Anglian Water Services Limited Lancaster House Lancaster House Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-006 conť d		Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)	
25-007	57 square metres of verge (south of Church Hill Lane) (Broadland District Council)	Unknown	
25-008	539 square metres of verge and public highway (Church Hill Lane) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
25-009	1111 square metres of private road and arable land (north of Weston Road) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-009 cont'd		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas transmission pipeline) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)	
25-010	57 square metres of verge and public highway (Weston Road) (Broadland District Council)	Orsted Hornsea Project Three (UK) Limited 5 Howick Place LONDON SW1P 1WG (in respect of right of access) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-011	16 square metres of public highway (Church Hill Lane), verge and public footpath (Weston Longville FP9) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
25-012	54 square metres of public highway (Church Hill Lane), verge and public footpath (Weston Longville FP9) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-012 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
25-013	5 square metres of verge and public footpath (Weston Longville FP9) (South of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
25-014	261 square metres of public footpath (Weston Longville FP9), land and verge (south of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-014 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
25-015	219 square metres of verge and public footpath (Weston Longville FP9) (south of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
25-016	1101 square metres of hedgerow and arable land (south of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-016 cont'd		Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millistream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-016 cont'd		National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus) Unknown (in respect of rights of service media and restrictive covenants)	
25-017	47124 square metres of arable land (south of Church Hill Lane) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of right of access to lay and maintain water main and restrictive covenants) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-017 cont'd		Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of gas apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-017 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights of service media and restrictive covenants)	
26-001	208 square metres of public highway (The Broadway) (Broadland District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights granted, rights to lay and maintain service media and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-002	654 square metres of woodland, verge and public highway (The Broadway) (Broadland District Council)	Unknown	
26-003	83 square metres of verge and public highway (The Broadway) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
26-004	13 square metres of woodland and verge (The Broadway) (Broadland District Council)	Unknown	
26-005	66 square metres of woodland and verge (The Broadway) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-006	380 square metres of hedgerow and woodland (The Broadway) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster House Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of right of access to lay and maintain gas main and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-007	111007 square metres of hedgerow, arable land, woodland, private road, verge and public highway (The Broadway) (Broadland District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas main and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas main and restrictive covenants) Unknown (in respect of right of access to lay and maintain gas main and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-008	937 square metres of arable land and verge (west of Taverham Road) (Broadland District Council)	Matthew John Rampton Kingfisher House 1 Gilders Way NORWICH Norfolk NR3 1UB (in respect of right of access to lay and maintain apparatus and right of passage of services through conducting media) Unknown (in respect of right of way and shooting rights)	
27-001	70 square metres of public highway (Taverham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
27-002	1835 square metres of arable land and private road (east of Taverham Road) (Broadland District Council)	Unknown (in respect of right of access and right of passage of services through conducting media)	
27-003	806 square metres of hedgerow and public highway (Taverham Road) (Broadland District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
27-004	64154 square metres of arable land, hedgerow and private road (east of Taverham Road) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of right of access and right of passage of services through conducting media)	
27-005	36546 square metres of watercourse (River Tud), arable land and hedgerow (north of the A47) (Broadland District Council)	Unknown	
27-006	2245 square metres of arable land and hedgerow (north of A47) (Broadland District Council)	National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of right of access to lay and maintain apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-001	20155 square metres of arable land and hedgerow (north of the A47) (Broadland District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of access and right of passage of services through conducting media)	
28-002	845 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-003	901 square metres of arable land and access track (north of the A47) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of access and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-004	121 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-005	3 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown (in respect of restrictive covenants and rentcharges as may have been imposed before 12 February 2009)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-006	298 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown	
28-007	14 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-008	589 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown (in respect of restrictive covenants and rentcharges as may have been imposed before 12 February 2009)	
28-009	1231 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-009 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted) Unknown	
28-010	649 square metres of public highway, lay by and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-010 cont'd		The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-011	49638 square metres of arable land, woodland, hedgerow and verges (south of the A47 and north of Church Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain) Unknown (in respect of right of access and right of passage of services through conducting media)	
28-012	NUMBER NOT IN USE	NUMBER NOT IN USE	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-013	NUMBER NOT IN USE	NUMBER NOT IN USE	
28-014	243 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain and in respect of right to maintain woodland path) Unknown (in respect of right of access and right of passage of services through conducting media)	
28-015	403 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Unknown	
28-017	1458 square metres of arable land, track and hedgerow (south of Church Road) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
28-021	358 square metres of public highway and verges (Broom Lane) (South Norfolk District Council)	Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-023	1 square metre of arable land and access track (north of Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road Norwich NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021) Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Norfolk County Council County Hall Martineau Lane Norwich NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)	
28-023 cont'd		Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Condimentum Limited Food Enterprise Park Colmans Close Easton Noriwch NR9 5FG (in respect of rights granted) Eurotek Engineering Limited Centrum Noriwch Research Park Colney Lane Colney Norwich NR4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)	
28-023 cont'd		Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-024	2 square metres of arable land and access track (north of Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road Norwich NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021) Norfolk County Council County Hall Martineau Lane Norwich NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)	
28-024 cont''d		Norwich City College of Further and Higher Education c/o Martin Colbourne lpswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016) Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Condimentum Limited	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Colmans Close Easton Noriwch NR9 5FG (in respect of rights granted)	
28-024 cont'd		Eurotek Engineering Limited Centrum Noriwch Research Park Colney Norwich NR4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Property Vision (Norwich) Limited c/o Flower Vision (Last Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-025	123 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Unknown	
28-026	6041 square metres of arable land and private road (Coleman Road) (north of Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021) Condimentum Limited Food Enterprise Park Colmans Close Easton NORWICH NR9 5FG (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-026 cont'd		Norfolk County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain) Unknown (in respect of right of access and right of passage of services through conducting media) Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted) Eurotek Engineering Limited Centrum Norwich Research Park Colney Lane Colney Norwich NE4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NE4 7UG (in respect of rights granted) Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	
28-027	483 square metres of public highway and verge (Church Lane) (South Norfolk District Council)	Broadland District Council Thorpe Lodge 1 Yarmouth Road NORWICH Norfolk NR7 0DU (in respect of rights of services as contained in a Transfer dated 20th May 2021)	
		Broadland Food Innovation Centre Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)	
		Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right to use drain and right of access to lay and maintain drain)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-027 cont'd		Norwich City College of Further and Higher Education c/o Martin Colbourne Ipswich Road Norwich NR2 2LJ (in respect of rights to display advertisements and restriction contained in overage agreement dated 8 August 2016) Unknown (in respect of right of access and right of passage of services through conducting media) Condimentum Limited Food Enterprise Park Colmans Close Easton NORWICH NR9 5FG (in respect of rights granted) Eurotek Engineering Limited Centrum Norwich Research Park Colney Lane Colney Norwich NE4 7UG (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Fischer Farms (2) Limited Enterprise Way Food Enterprise Park Easton Norwich NR9 5FX (in respect of rights granted)	
28-027 cont'd		Property Vision (Norwich) Limited c/o Flower Vision (East Anglia) Limited 29/32 Francis Way Bowthorpe Norwich NR5 9JA (in respect of rights granted)	
29-001	2802 square metres of access track and verges (east of Colton Road) (South Norfolk District Council)	Annette Lucie Alston Honingham Thorpe Farm Norwich Road Colton NORWICH NR9 5BZ (in respect of right of access)	
		Ian James Alston Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		William Young (Dereham) Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)	
29-002	95 square metres of public highway and verges (Colton Road) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
29-003	60617 square metres of arable land, woodland, hedgerow and track (east of Colton Road) (South Norfolk District Council)	Annette Lucie Alston Honingham Thorpe Farm Norwich Road Colton NORWICH NR9 5BZ (in respect of right of access) lan James Alston Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		William Young (Dereham) Limited Honingham Thorpe Farm Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of right of access)	
29-004	8190 square metres of arable land and hedgerows (east of Colton Road) (South Norfolk District Council)	Unknown (in respect of rights reserved)	
29-005	685 square metres of public highway and verges (Colton Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
29-006	32592 square metres of arable land, grassland and hedgerows (west of Colton Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
29-006 conťd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
29-007	3726 square metres of arable land and hedgerows (west of Colton Road) (South Norfolk District Council)	Unknown (in respect of rights reserved)	
29-008	43135 square metres of arable land, grassland, woodland and hedgerows (west of Chapel Street) (South Norfolk District Council)	Colin Herbert Bothway Cherry Tree Farm Salle NORWICH Norfolk NR10 4SF (in respect of right of entry to extract mines and minerals, timber rights, positive covenants and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Unknown (in respect of light and air)	
30-001	608 square metres of watercourse (River Yare) (South Norfolk District Council)	Unknown	
30-002	23868 square metres of arable land, hedgerow, drain and private road (south of the River Yare) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	
30-002 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
30-003	20021 square metres of arable land, hedgerows and verge (north of Chapel Street) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
30-004	120 square metres of private road (north of Chapel Street) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	
30-006	484 square metres of land and buildings (north of Chapel Street) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus and sewer)	
30-007	7 square metres of private access road and verge (north of Chapel Street) (South Norfolk District Council)	Unknown	
30-008	25 square metres of public highway (Chapel Street) (South Norfolk District Council)	Unknown	
30-009	18 square metres of verge (north of Chapel Street) (South Norfolk District Council)	Unknown	
30-010	959 square metres of public highway (Chapel Street) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
30-011	35447 square metres of arable land, hedgerows and tree line (south of Chapel Street) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer)	
30-011 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
30-012	532 square metres of arable land and hedgerow (south of Chapel Street) (South Norfolk District Council)	Unknown	
30-013	1227 square metres of watercourse (River Tiffey) (north of Watton Road, B1108) (South Norfolk District Council)	Unknown	
30-014	9178 square metres of woodland (south of River Tiffey) (South Norfolk District Council)	Unknown (in respect of rights reserved)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
30-015	174 square metres of woodland (south of River Tiffey) (South Norfolk District Council)	Unknown	
30-016	693 square metres of woodland (south of the River Tiffey) (South Norfolk District Council)	Helga Erika Jalil 1A Gilbert Way Cringleford NORWICH Norfolk NR4 7RN (in respect of a right of access, right of passage and running of surface water and provision of light and air)	
31-001	1209 square metres of verge and public highway (Watton Road, B1108) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
31-001 conťd		Unknown	
31-003	702 square metres of private road (south of Watton Road, B1108) (South Norfolk District Council)	Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	
31-004	25870 square metres of private road, arable land and hedgerow (north of Burdock Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
31-005	679 square metres of woodland (north of Burdock Lane) (South Norfolk District Council)	Unknown (in respect of rights reserved)	
31-006	2467 square metres of private road (north of Burdock Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	
31-007	600 square metres of verge and public highway (Burdock Lane) (South Norfolk District Council)	Unknown	
31-008	33 square metres of verge (south of Burdock Lane) (South Norfolk District Council)	Unknown	
31-010	14 square metres of public highway (Burdock Lane) (South Norfolk District Council)	Unknown	
31-012	31953 square metres of arable land, drain, hedgerow and public footpath (Great Melton FP1) (west of Pockthorpe Road) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of right of access to use, lay and maintain electricity cables) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Unknown (in respect of restrictive covenants still subsisting and capable of being enforced)	
32-001	26466 square metres of hedgerow, drain and arable land (north of Skoyles Lane) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
32-002	13733 square metres of hedgerow and arable land (north of Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of right of way over access track) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009

32-003	520 square metres of verge and public highway (Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted)
		Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (In respect to water apparatus)
		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
32-004	110 square metres of verge and public highway (Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted)	
32-004 conť d		Unknown	
32-005	390 square metres of arable land and track (south of Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Susanne Eva Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Thomas Charles Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	
32-006	28012 square metres of arable land, ditch and hedgerows (south of Skoyles Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights granted) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (In respect to water apparatus)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Susanne Eva Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	
32-006 cont'd		Thomas Charles Markham Old Boundary Barn Green Lane Wramplingham WYMONDHAM Norfolk NR18 0SA (in respect of rights granted)	
33-002	29 square metres of public highway (Pockthorpe Road) (South Norfolk District Council)	Unknown	
33-004	11 square metres of public highway (Melton Road) (South Norfolk District Council)	Unknown	
33-005	594 square metres of private road (west of Melton Road) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
33-006	1295 square metres of hedgerow (west of Melton Road) (South Norfolk District Council)	Unknown (in respect of restrictive covenants as may have been imposed before 16 September 2011)	
33-007	13970 square metres of arable land and hedgerow (west of Melton Road) (South Norfolk District Council)	Unknown (in respect of restrictive covenants as may have been imposed before 16 September 2011)	
33-008	17097 square metres of arable land, ditch and verge (west of Melton Road) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	
33-010	826 square metres of public highway (Melton Road) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-011	439 square metres of arable land and verge (east of Melton Road) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	
33-011 cont'd		Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-012	23630 square metres of arable land, hedgerow and verge (east of Melton Road) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Susan Jane Spooner 9 The Meadows Hethersett NORWICH Norfolk NR9 3ND (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-013	820 square metres of arable land (east of Melton Road) (South Norfolk District Council)	John Roger Barnard Woodcocks Woodcocks Lane Great Melton NORWICH Norfolk NR9 3BY (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Mary Tuddenham 45A Norwich Common WYMONDHAM NR18 0SW (in respect of rights of drainage, rights of access to lay and maintain apparatus and right of passage of services through conducting media) Persimmon Homes Limited Persimmon House Fulford YORK North Yorkshire YO19 4FE (in respect of rights granted) Susan Jane Spooner 9 The Meadows Hethersett NORWICH NOR	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-017	28370 square metres of arable land, drain and hedgerow (north of Ketts Oak, B1172) (South Norfolk District Council)	Nicholas Edward Evans-Lombe Marlingford Hall Church Road Marlingford NORWICH Norfolk NR9 5HP (in respect of right of way over access track) Robert Alan Barnard Rowan Lodge Marlingford Road Bawburgh NORWICH Norfolk NR9 3LU (in respect of right of passage of services through conducting media)	
34-001	19636 square metres of arable land and hedgerow (north of Ketts Oaks, B1172) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right of access to lay and maintain apparatus)	
34-002	1243 square metres of verge (Ketts Oak, B1172) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-003	2192 square metres of arable land (north of Ketts Oak, B1172) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of right of access to lay and maintain apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H SNP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-004	117 square metres of public highway (Ketts Oaks, B1172) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-005	139 square metres of public highway (Ketts Oaks, B1172) (South Norfolk District Council)	Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of street apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
34-006	2059 square metres of public highway (Ketts Oaks, B1172) and verge (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-006 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Norfolk County Council County Hall Martineau Lane NORWICH Norfolk NR1 2DH (in respect of street apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
34-008	2987 square metres of arable land and hedgerow (south of Ketts Oaks, B1172) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-009	34350 square metres of arable land and hedgerow (south of Ketts Oaks, B1172) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
34-010	50248 square metres of arable land and hedgerow (north of the A11) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
35-002	4801 square metres of public highway (A11), verge and hedgerows (west of Ketteringham Lane) (excluding all interests of the Crown) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-002 conťd		The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights of way over the land and maintenance of access ways and bridges)	
35-003	2626 square metres of rail line and verge (south of the A11) (South Norfolk District Council)	Unknown	
35-004	435 square metres of private road (west of Ketteringham Lane and south of A11) (South Norfolk District Council)	Unknown (in respect of rights granted as contained in a Deed dated 27th June 1983)	
35-005	34 square metres of private road (west of Ketteringham Lane) (South Norfolk District Council)	Chris Thrower Forgestone House Low Street Ketteringham WYMONDHAM NR18 9RY (in respect of rights of access and to draw water from an ancient well) Unknown	
35-006	40884 square metres of Land and buildings on the north side of arable land, private road and verge (west of High Street) (South Norfolk District Council)	Unknown (in respect of rights granted as contained in a Deed dated 27th June 1983)	
35-007	1076 square metres of public highway (High Street) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-008	87 square metres public highway (High Street) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
35-009	308 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948) Unknown (in respect of right to use drainage system, right to enter and right to lay and maintain water mains and pipes stated in conveyance dated 10th June 1959)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-010	24020 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to use drainage system, right to enter and right to lay and maintain water mains and pipes stated in conveyance dated 10th June 1959)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-011	107807 square metres of private road (Church Road), woodland, hedgerow, verge and public footpath (Ketteringham FP4) (excluding all interests of the Crown) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-011 cont'd		Sharon Gowing Park Lodge Ketteringham Park Ketteringham WYMONDHAM NR18 9PF (in respect of right of way over access track and drainage rights as contained in a Conveyance dated 5th January 1959) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)	
36-001	4970 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-001 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Sharon Gowing Park Lodge Ketteringham Park Ketteringham WYMONDHAM NR18 9PF (in respect of right of way over access track and drainage rights as contained in a Conveyance dated 5th January 1959) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)	
36-002	226 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-003	266 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
36-004	1002 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-005	7623 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty CV7 9JU (in respect of gas apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NorkVICH Norfolk NR9 5BZ (in respect of shooting rights)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-005 conť d		Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)	
36-006	56249 square metres of verge, arable land and private road (east of Hethersett Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of gas main and right of way)	

equinor 💱

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-006 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Elliot Heinrich Torston Macrow 3 Grove Cottages Norwich Road Colton NORWICH Norfolk NR9 5BZ (in respect of shooting rights) Unknown (in respect of drainage, sewerage and water discharge and rights of way for Ketteringham Hall Park Lodge, Ketteringham Estate House Garage, Wood Cottage, Lodge Cottage and other property)	
36-007	4812 square metres of verge and arable land (east of Hethersett Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

equinor

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-007 cont'd		Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas pipes) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights of access to lay and maintain electricity cables and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-007 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of right of access and timber rights) Unknown (in respect of rights reserved)	
36-008	254 square metres of public highway (Hethersett Road) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Unknown	
36-009	19618 square metres of hedgerow and arable land (east of Hethersett Road) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights of access to lay and maintain electricity cables and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-009 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of right of access and timber rights) Unknown (in respect of rights reserved)	
36-011	55959 square metres of arable land, hedgerows, verge and public footpath (East Carleton FP4) (west of Intwood Lane) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-011 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
37-001	903 square metres of arable land and public footpath (East Carleton FP4) (west of Intwood Lane) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
37-002	808 square metres of verge and public highway (Intwood Lane) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
37-002 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
37-003	553 square metres of arable land (east of Intwood Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	
37-004	25 square metres verge (south of Intwood Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
37-004 conť d		Unknown	
37-005	27480 square metres of hedgerow, arable land and verge (east of Intwood Lane) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of right of access to lay and maintain gas mains)	



Doc. No. C282-DM-Z-GA-00003

The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
37-006	82917 square metres of public footpath (Swardeston FP2, Swardeston FP3 and Swardeston FP4), hedgerow, watercourse (unnamed) and arable land (west of Norwich Road, B1113) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown (in respect of rights as contained in a Conveyance dated 21st October 1902)
38-001	276 square metres of access track, verge and arable land (west of Main Road, B1113) (South Norfolk District Council)	Unknown (in respect of rights as contained in a Conveyance dated 21st October 1902)
38-002	1135 square metres of verge and public highway (Main Road, B1113) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
38-002 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
38-003	160 square metres of verge and arable land (east of Main Road, B1113) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	
38-004	16683 square metres of arable land and hedgerow (east of Main Road, B1113) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
38-006	63248 square metres of public footpaths (Mulbarton FP6 and Mulbarton FP7), arable land and hedgerows (south east of Main Road, B1113) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
38-007	3193 square metres of arable land and hedgerow (west of Swardeston Lane) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
38-008	1398 square metres of arable land (west of Swardeston Lane) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
38-009	83 square metres of public highway (Swardeston Lane and Gowthorpe Lane) and verge (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
38-010	17 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Unknown	
38-011	2 square metres of public highway (Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
38-011 cont'd		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
38-012	42 square metres of private road (east of Swardeston Lane) and public byway open to all traffic (Swainsthorpe BOAT6) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
38-013	1149 square metres of public footpath (Swainsthorpe FP1), verge and arable land (east of Swardeston Lane) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
38-014	604 square metres of public highway (Swardeston Lane) and verge (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown	
38-015	6 square metres of verge (east of Swardeston Lane) (South Norfolk District Council)	Unknown	
38-016	61 square metres of verge (east of Swardeston Lane) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
38-017	22746 square metres of arable land (north of Church Road) and public footpath (Swainsthorpe FP1) (South Norfolk District Council)	Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-001	19624 square metres of arable land (south of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Unknown	
39-002	7643 square metres of arable land and public bridleway (Swainsthorpe BR2) (north of Church Road and west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-003	1670 square metres of arable land and public bridleway (Swainsthorpe BR2) (north of Church Road and west of Ipswich Road, A140) (South Norfolk District Council)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights of access) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-004	289 square metres of sewage pumping station (west of Ipswich Road, A140) (South Norfolk District Council)	South Norfolk District Council South Norfolk Council South Norfolk House Cygnet Court Long Stratton NORWICH NR15 2XE (in respect of rights granted)	
39-005	4101 square metres of arable land and verge (south of Hickling Lane and west of Ipswich, A140) (South Norfolk District Council)	Unknown National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights granted)	
39-006	8397 square metres of arable land and verge (south of Hickling Lane and west of Ipswich, A140) (South Norfolk District Council)	Unknown National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of rights granted)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-006 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)	
39-007	468 square metres of arable land and verge (south of Hickling Lane and west of Ipswich, A140) (South Norfolk District Council)	Unknown	
39-008	308 square metres of verge, hedgerow and public byway open to all traffic (Swainsthorpe BOAT6) (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Unknown	
39-009	95 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Unknown	
39-010	1 square metres of land and public byway open to all traffic (Swainsthorpe BOAT6) (Hickling Lane) (South Norfolk District Council)	Unknown	
39-011	455 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-012	464 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-013	67 square metres of road (Hickling Lane), hedgerow, verge and public byway open to all traffic (Swainsthorpe BOAT6) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-014	112837 square metres of arable land and part of public byway open to all traffic (Swainsthorpe BOAT6) (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications) Unknown (in respect of right of way over Hickling Lane reserved)	
39-015	52057 square metres of arable land and hedgerow (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-015 conťd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)
39-016	936 square metres of hedgerow and arable land (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)
39-017	1005 square metres of hedgerow and arable land (north of Hickling Lane and west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-017 conťď		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-018	5791 square metres of arable land (north of Hickling Lane) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-019	83657 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-019 cont'd		Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)	
39-020	1227 square metres of arable land and hedgerow (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-021	5569 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-022	5093 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-022 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus)	
39-023	3018 square metres of woodland (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to lay and maintain electricity cables and restrictive covenants) Energis Communications Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of rights of access to lay and maintain telecommunication apparatus)	
39-025	15 square metres of public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-025 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-026	56 square metres of public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of a right of way) Unknown Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	
39-027	24 square metres of public bridleway (Stoke Holy Cross BR3) and hedgerow (west of Ipswich Road, A140) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-027 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of a right of way) Unknown Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	
39-028	5153 square metres of woodland, hedgerow and public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-028 cont'd		Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-029	9833 square metres of woodland, hedgerow and public bridleways (Stoke Holy Cross BR3 and Swardeston BR12) (south of Norwich Main Substation and west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitey Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-029 cont'd		Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track)	
39-030	203 square metres of public bridleway (Swardeston BR12) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-031	654 square metres of public bridleway (Swardeston BR12) (west of Ipswich Road, A140) (South Norfolk District Council)	Unknown	
39-032	1979 square metres of arable land and hedgerow (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-033	1925 square metres of arable land, hedgerow and public bridleway (Swardeston BR12) (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-033 conťd		Unknown (in respect of rights granted, reserved and restrictive covenants)	
39-034	40840 square metres of arable land (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-034 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)	
39-035	15126 square metres of arable land and hedgerow (west of Norwich Main Substation and Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-035 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	
39-036	5908 square metres of arable land (west of Ipswich Road, A140) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of right to maintain gas apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-036 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) National Grid function Unknown (in respect of rights granted, reserved and restrictive covenants)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-037	25343 square metres of Norwich Main Substation and private road (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	
39-038	4098 square metres of verge, hedgerow and public bridleway (Stoke Holy Cross BR3) (west of Ipswich Road) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-038 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Felicity Mary Whitley Cavell Barn The Common Swardeston NORWICH NR14 8DZ (in respect of right of way over access track) Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-038 cont'd		Rosemary Margaret Watkinson Hall Farm House Gowthorpe Lane Swardeston NORWICH NR14 8DS (in respect of right of way over access track) Switlana Steward 43 Bowden Hill Lacock CHIPPENHAM SN15 2PP (in respect of right of way over access track) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	
39-039	2117 square metres of hedgerow and verge (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-039 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-040	634 square metres of hedgerow and verge (west of Ipswich Road, A140) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millistream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-040 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	
39-041	7492 square metres of private road and grassland (Norwich Main Substation and west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-041 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	
39-042	3223 square metres of private road and verge (west of Ipswich Road, A140 and south of Mangreen Road) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-042 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SEI 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SEI 6NP (in respect of rights granted by lease dated 30 March 1990) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications) Vodafone Limited Vodafone Limited Vodafone Limited Sod Brook Drive READING RG2 6UU (in respect of telecommunications)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-043	14336 square metres of verge, grassland and hedgerows (Norwich Main Substation, west of Ipswich Road, A140 and south of Mangreen Road) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990)	



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-043 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	
39-044	2901 square metres of private road (west of Ipswich Road and south of Mangreen Road) (South Norfolk District Council)	Barry Ernest Brooks Woodlands Long Lane Bracon Ash NORWICH Norfolk NR14 8AN (in respect of right of access) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of easement)	

equinor 🗜

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-044 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of rights granted by lease dated 30 March 1990) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)	

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-001	16719 square metres of grassland (west of Ipswich Road, A140) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of right to maintain gas apparatus) Centrica PLC Millistream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-001 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) National Grid Gas PLC Grand Buildings 1-3 Strand LONDON WC2N 5EH (in respect of right of access to lay and maintain gas pipelines and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)		
40-002	5925 square metres of public highway (Ipswich Road, A140 and Mangreen Road) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus) Lumen Technologies UK Limited 260-266 Goswell Road LONDON EC1V 7EB (in respect of telecommunications)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-002 cont'd		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications) Vodafone Limited Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)		
40-003	126 square metres of verge (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications)		

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-003 cont'd		Unknown Virgin Media Limited 500 Brook Drive READING RG2 6UU (in respect of telecommunications) Vodafone Limited Vodafone House The Connection NEWBURY Berkshire RG14 2FN (in respect of telecommunications)		
40-004	19 square metres of private road (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown		
40-005	12 square metres of verge (south of Mangreen Road and west of Ipswich Road, A140) (South Norfolk District Council)	Cadent Gas Limited Unit 3 Ansty Park Pilot Way Ansty COVENTRY CV7 9JU (in respect of gas apparatus)		

Book of Reference (Revision H)



Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-005 cont'd		Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown		
40-006	27 square metres of Mangreen Hall Farm, Swardeston, Norwich, NR14 8DD (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants) Unknown (in respect of rights granted, reserved and restrictive covenants)		
40-007	161 square metres of verge and public highway (Mangreen Road) (South Norfolk District Council)	Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-007 conťď		National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown		
40-008	5875 square metres of hedgerow and verge (Mangreen Hall Farm, west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of right of access to maintain electricity cables, right to fell or lop trees and restrictive covenants)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-008 cont'd		Eastern Power Networks PLC Newington House 237 Southwark Bridge Road LONDON SE1 6NP (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1-3 Strand LONDON WC2N 5EH (in respect of electricity apparatus) Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown (in respect of rights granted, reserved and restrictive covenants)		
40-009	219 square metres of verge and private road (west of Ipswich Road, A140) (South Norfolk District Council)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of service media) Unknown (in respect of rights granted, reserved and restrictive covenants)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 3 County of Norfolk			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
40-010	353 square metres of public highway (Ipswich Road, A140) (South Norfolk District Council)	Openreach Limited Kelvin House 123 Judd Street LONDON WC1H 9NP (in respect of telecommunications) Unknown		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-001	Rights A	5165 square metres of foreshore land seaward of the Mean High Water (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
01-002	Rights A	1523 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-003	Rights A	32 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB	
01-004	Rights A	1195 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	(in respect of right of access) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as reputed owner)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-005	Rights A	366 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-006	Rights A	814 square metres of beach and public footpath (Weybourne FP7) (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-007	Rights A	4767 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-008	Rights A	765 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-011	Temporary Possession	652 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	
01-012	Rights B	36 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right of access)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4			
	Land which is	(Applications: Prescribed F	ich application is being made under Regulation 7(1)(d) of the Infrastructure Planning forms and Procedure) Regulations 2009 ounty of Norfolk	
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-013	Temporary Possession	57 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-014	Rights B	4655 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-015	Rights B	393 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-016	Rights B	33 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe over part)	
01-017	Rights C	51 square metres of beach and access track (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-018	Rights C	7854 square metres of access track (north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Obering New Oberland Dudge on Estensions Offshare Wind Form Development Operant Operant			
	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-019	Temporary Possession	10156 square metres of access track and public footpath (Weybourne FP7) (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-020	Rights C	9694 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-021	Rights B	75 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-022	Rights B	24 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-023	Rights B	630 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-024	Rights B	3359 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-026	Rights B	265 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-027	Rights B	304 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-028	Rights B	5462 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-029	Rights D	9686 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-030	Rights D	8080 square metres of arable land (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-031	Temporary Possession	4437 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Obstingthene Obstington Dudge on Extensions Offshare Wind Form Development Opposition			
	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009			
		C	ounty of Norfolk	
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-032	Temporary Possession	21 square metres of access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-033	Temporary Possession	9105 square metres of grassland and access track (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-034	Rights D	22426 square metres of grassland (Weybourne Military Camp) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-035	Rights C	15 square metres of public highway (The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
01-037	Rights C	29 square metres of access track (north of The Street, A149) (excluding all interests of the Crown) (North Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to enter, right to service media and right to lay, use and maintain a signals cable and a water pipe)	
03-009	Rights D	13406 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
03-010	Rights D	15195 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	
03-011	Rights D	9070 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	
04-001	Temporary Possession	1675 square metres of woodland and track (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
04-001 cont'd			The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	
04-002	Temporary Possession	1869 square metres of woodland and track (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
04-003	Rights D	38365 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF	
04-004	Temporary Possession	9 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
04-004 cont'd			The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)	
04-011	Temporary Possession	1355 square metres of access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way)	
04-013	Temporary Possession	465 square metres of woodland, access track and public footpath (Weybourne FP6) (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk				
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made		
04-013 cont'd			The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF (in respect of right of way) The Secretary Of State For Environment Food And Rural Affairs Seacole Building 2 Marsham Street London SW1P 4DF		
10-009	Temporary Possession	1104 square metres of arable land and access track (south of Sweetbriar Lane) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant)		
10-010	Rights D	26850 square metres of arable land and hedgerow (south of Sweetbriar Lane and west of Matlaske Road) (excluding all interests of the Crown) (North Norfolk District Council)	The Forestry Commission 620 Bristol Business Park Coldharbour Lane BRISTOL Avon BS16 1EJ (in respect of forestry dedication covenant)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
16-014	Temporary Possession	259 square metres of verge, hedge row and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987)	
16-015	Rights D	502 square metres of hedgerow and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987)	
16-017	Temporary Possession	258 square metres of verge, arable land and private highway (Docking Farm, east of B1149) (excluding all interests of the Crown) (Broadland District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of rights and covenants contained in assent dated 2 October 1987)	
28-002	Rights D	845 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
28-004	Temporary Possession	121 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-005	Temporary Possession	3 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-006	Temporary Possession	298 square metres of public highway and access gate (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-007	Temporary Possession	14 square metres of public highway (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-008	Rights D	589 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
28-009	Rights D	1231 square metres of public highway and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
28-010	Rights D	649 square metres of public highway, lay by and verges (A47) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights granted)	
35-002	Rights D	4801 square metres of public highway (A11), verge and hedgerows (west of Ketteringham Lane) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary of State for Transport Great Minster House 33 Horseferry Road LONDON SW1P 4DR (in respect of rights of way over the land and maintenance of access ways and bridges)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk				
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made		
35-009	Temporary Possession	308 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines) The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948)		
35-010	Rights D	24020 square metres of arable land, woodland and verge (east of High Street) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines)		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
35-010 cont'd			The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of right to maintain and use government pipeline with the provision of section 12 of the Requisitioned Land and Way Works Act 1948)	
35-011	Rights D	107807 square metres of private road (Church Road), woodland, hedgerow, verge and public footpath (Ketteringham FP4) (excluding all interests of the Crown) (South Norfolk District Council)	The Secretary Of State For Defence Ministry of Defence Main Building Horse Guards Avenue Whitehall LONDON SW1A 2HB (in respect of abandoned pipelines)	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

r	The Chemischers Cheel and Dudreen Estensions Offeners Wind Ferry Development Concept Order				
	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 County of Norfolk				
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land		
01-001	Rights A	5165 square metres of foreshore land seaward of the Mean High Water (Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space		
01-002	Rights A	1523 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space		
01-003	Rights A	32 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land	
01-004	Rights A	1195 square metres of foreshore land seaward of the Mean High Water (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space	
01-005	Rights A	366 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space	
01-006	Rights A	814 square metres of beach and public footpath (Weybourne FP7) (Weybourne Military Camp and Weybourne Beach, north of Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 County of Norfolk			
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land	
01-007	Rights A	4767 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space	
01-008	Rights A	765 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space	
01-009	Rights A	2135 square metres of beach and public footpath (Weybourne FP7)(Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space	

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 County of Norfolk				
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land		
01-010	Temporary Possession	1680 square metres of beach and public footpath (Weybourne FP7) (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space		
01-011	Temporary Possession	652 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space		
01-012	Rights B	36 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space		

Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5				
	County of Norfolk				
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land		
01-013	Temporary Possession	57 square metres of beach (Weybourne Beach, north of Weybourne Military Camp and Beach Lane, Weybourne) (excluding all interests of the Crown) (North Norfolk District Council)	Open Space		
03-009	Rights D	13406 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	National Trust Land		
03-011	Rights D	9070 square metres of woodland (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	National Trust Land		
04-001	Temporary Possession	1675 square metres of woodland and track (east of Sandy Hill Lane) (excluding all interests of the Crown) (North Norfolk District Council)	National Trust Land		
17-001	Rights D	2108 square metres of tree and hedge line (east of Marriott Way, B1149) (Broadland District Council)	Open Space		



Book of Reference (Revision H)

Doc. No. C282-DM-Z-GA-00003

Rev. H

	The Sheringham Shoal and Dudgeon Extensions Offshore Wind Farm Development Consent Order BOOK OF REFERENCE - PART 5 County of Norfolk			
Number on Land Plans	on Land Extent of acquisition or use Description of land Category of Land			
23-001	Rights D	1359 square metres of public highway (Marriott's Way) (Broadland District Council)	Open Space	

Page 1368 of 1368